## Laboratoire d'Analyse et de Mathématiques appliquées Université Paris-Est Marne-la-Vallée

Dipartimento di Matematica - Università Roma Tre

## Minimal surfaces in Riemannian manifolds

Filippo Morabito

A Thesis submitted for the degree of Doctor of Philosophy

Thesis Advisor Laurent Hauswirth Univ. Paris Est Supervisor Massimiliano Pontecorvo Univ. Roma Tre

11 October 2007

# Contents

1	Inde face 1.1	ex and nullity of the Gauss map of the Costa-Hoffman-Meeks sur- s Preliminaries	<b>7</b> 9
	1.2	The case of the Costa-Hoffman-Meeks surface of genus smaller than 38	11
	1.3	The case $g \ge 38$	17
	1.4	The index and the nullity of the Costa-Hoffman-Meeks surfaces	28
	1.5	Appendix	29
		<ul><li>1.5.1 Divisors and Riemann-Roch theorem</li></ul>	29
		at the ramification points	30
		minimal surface.	34
<b>2</b>	Abo	out a family of deformations of the Costa-Hoffman-Meeks surfaces	36
	2.1	Introduction	36
	2.2	Preliminaries and notation	38
	2.3	The deformation of the surface and its Jacobi operator	39
		2.3.1 The Jacobi operator	41
		2.3.2 The properties of the kernel and of the range of the Jacobi operator	48
	2.4	The proof of the main result	51
3	AC	$\operatorname{Costa-Hoffman-Meeks}$ type surface in $\mathbb{H}^2  imes \mathbb{R}$	<b>54</b>
	3.1	Introduction	54
	3.2	Minimal graphs in $\mathbb{H}^2 \times \mathbb{R}$	55
	3.3	The linearized operator about the hyperbolic plane	56
		3.3.1 The mapping properties of the Laplace operator	58
	3.4	A family of minimal surfaces close to the hyperbolic plane	60
	3.5	The catenoid in $\mathbb{H}^2 \times \mathbb{R}$	62
	3.6	A family of minimal surfaces close to a catenoid on $S^1 \times [r_{\epsilon}, 1)$	65
	3.7	The mean curvature in a neighbourhood of $\{0,0\} \times \mathbb{R}$ in $\mathbb{H}^2 \times \mathbb{R}$	68
	3.8	A rescaled Costa-Hoffman-Meeks type surface	71

		3.8.1 The Costa-Hoffman-Meeks surface.	71
		3.8.2 The Jacobi operator	75
	3.9	An infinite dimensional family of minimal surfaces which are close to a	
		compact part of a rescaled Costa-Hoffman-Meeks type surface in $\mathbb{H}^2\times\mathbb{R}.$ .	78
	3.10	The matching of Cauchy data	84
	3.11	Appendix	87
4	Sing	ly periodic minimal surfaces with arbitrary nonzero genus and in-	
	finit	ely many ends	90
	4.1	Introduction	90
	4.2	A Costa-Hoffman-Meeks type surface with bent	
		catenoidal ends	91
	4.3	An infinite dimensional family of minimal surfaces which are close to $M_k(\xi)$	97
	4.4	KMR examples $M_{\sigma,\alpha,\beta}$	102
		4.4.1 $M_{\sigma,\alpha,\beta}$ as a graph over $\{x_3=0\}/T$	105
		4.4.2 Parametrization of the KMR example on the cylinder $\ldots$	109
	4.5	The Jacobi operator about $M_{\sigma,\alpha,\beta}$	110
		4.5.1 The mapping properties of the Jacobi operator $\ldots$	111
	4.6	A family of minimal surfaces close to $M_{\sigma,0,\beta}$ and $M_{\sigma,0,\beta}$	120
	4.7	Periodic flat cylinder	125
	4.8	The existence of minimal graph close to a Scherk type surface	129
	4.9	The matching of Cauchy data	132
	4.10	Appendix A	136
	4.11	Appendix B	138
	4.12	Appendix C	139

#### Acknowledgements

I would like to thank my thesis advisor, Laurent Hauswirth, who introduced me to the minimal surfaces theory and brought interesting problems to my attention, for his invaluable advice and suggestions.

I am also deeply indebted with Massimiliano Pontecorvo for his helpfullness and encouragement and for having accepted to be my supervisor.

Of course I cannot forget to thank the members of the faculty of the Graduate school in Mathematics of University Roma Tre for having trusted me.

#### Summary of the thesis

This thesis is devoted to the solution of some problems of various nature about the minimal surfaces.

The study of minimal surfaces in  $\mathbb{R}^3$  started with Lagrange in 1762. He studied the problem of determining a graph over an open set W in  $\mathbb{R}^2$ , with the least possible area among all surfaces that assume given values on the boundary of W.

In 1776, Meusnier supplied a geometric interpretation of the minimal graph equation: the mean curvature H vanishes. Nowadays it has become customary to use the term minimal surface for any surface satisfying H = 0, notwithstanding the fact that such surfaces often do not provide a minimum for the area.

In all of the questions I dealed with in this work, one minimal surface plays the key role. It is the Costa-Hoffmann-Meeks surface, the most famous minimal surface. The discovery of the Costa surface was responsible for the rekindling of interest in minimal surfaces in 1982. In that year C. Costa showed the existence of a complete (i.e., it has no boundary) minimal surface of finite topology. It has genus 1 and three ends. D. Hoffman and W. H. Meeks III showed the embeddedness of this surface (i.e. it does not intersect itself). Until that moment the only other known embeddable complete minimal surfaces in  $\mathbb{R}^3$  were the plane, the catenoid and the helicoid. They were discovered over two hundred years ago, and it was conjectured that these were the only embedded complete minimal surfaces. Later D. Hoffman and W. H. Meeks III generalized the work of C. Costa showing the existence of a family of complete embedded minimal surfaces with three ends and genus  $k \ge 1$ . We denote by  $M_k$  the surface of genus k. It is known as Costa-Hoffman-Meeks of genus k.

An important property of the minimal surfaces is the non degeneracy. The non degeneracy is defined in terms of the space of the Jacobi functions on the surface, that is the functions which belong to the kernel of the Jacobi operator. This operator is defined as the linearized of the mean curvature operator.

J. Pérez and A. Ros showed that the set of the non degenerate properly embedded minimal surfaces with finite total curvature and fixed topology in  $\mathbb{R}^3$ , has a structure of finite dimensional real-analytic manifold. As application they showed that for  $M_k$  with  $2 \leq k \leq 37$ , there exists a family of minimal surfaces with three horizontal ends that are obtained by infinitesimal deformations by  $M_k$ . This result is based on a work of S. Nayatani which assures the non degeneracy of the Costa-Hoffman-Meeks surface only if its genus assumes the values described above. In his work S. Nayatani computed the dimension of the kernel and the index (i. e. the number of the negative eigenvalues) of the Jacobi operator about  $M_k$  but only if  $1 \le k \le 37$ . He showed that the dimension of the kernel equals 4. From that it follows the non degeneracy of  $M_k$ . In chapter 1 I show that it is possible to extend the result of S. Nayatani for bigger values of k. To be more precise I show that for  $k \ge 38$  the dimension of the kernel and the index of the Jacobi operator about  $M_k$ , are respectively equal to 4 and 2k + 3. That allows us to state that the surface  $M_k$  is non degenerate also for  $k \ge 38$ .

The non degeneracy of the surface  $M_k$  is one of the essential ingredients of the proof due to L. Hauswirth and F. Pacard of the existence of a new family of examples of minimal surfaces. Thanks to result described in 1, their construction extends automatically to higher values of k. The same result is used in the other sections of the thesis. Without it the constructions that I will describe briefly in the following, would hold only for  $k \leq 37$ .

In chapter 2 following J. Pérez and A. Ros, I show the existence of a family of immersed minimal deformations of  $M_k$  for  $k \ge 1$  having three embedded ends. In difference with  $M_k$ , the generic element of this family does not enjoy any property of symmetry. In fact, the admitted deformations are: the rotation about a vertical axis, the translation and dilation of any of the three ends of the surface. In addition, it is possible to bend the two catenoidal type ends and change the type of the middle end obtaining a catenoidal type end.

Here it is the statement of the main result.

**Theorem 1.** For each possible choice of the limit values of the normal vectors of the three ends, there is, up to isometries, a 1-dimensional real analytic family of smooth minimal deformations of  $M_k$ , for  $k \ge 1$ , letting the middle planar end horizontal.

The last two chapters of the thesis are devoted to the construction of new families of examples of minimal surfaces. Their construction is based on a gluing procedure which involves the surface  $M_k$ .

In chapter 3 I show the existence in the space  $\mathbb{H}^2 \times \mathbb{R}$ , where  $\mathbb{H}^2$  denotes the hyperbolic plane, of a family of minimal examples inspired to  $M_k$ . The statement of the main theorem is

**Theorem 2.** For all  $k \ge 1$  there exists in  $\mathbb{H}^2 \times \mathbb{R}$  a minimal surface of genus k with three horizontal ends: two catenoidal type ends and a middle planar end.

I glue the image by a homothety of parameter  $2\epsilon$ , with  $\epsilon$  sufficiently small, of a compact part of  $M_k$  along its three boundary curves to two minimal graphs that are respectively asymptotic to an upper half catenoid and a lower half catenoid defined in  $\mathbb{H}^2 \times \mathbb{R}$  and to a minimal graph asymptotic to  $\mathbb{H}^2 \times \{0\}$ . The chapter 4 is devoted to the construction of two new families of examples of periodic minimal surfaces with genus bigger than 1. This result has been obtained in collaboration with Laurent Hauswirth and M. Magdalena Rodríguez Pérez.

We denote by  $\mathcal{K} = \{M_{\sigma,\alpha,\beta}\}_{\sigma,\alpha,\beta}$  the family of minimal surfaces called Karcher-Meeks-Rosenberg examples or toroidal halfplane layers. They have been classified as the only doubly periodic minimal surfaces in  $\mathbb{R}^3$  with genus one and finitely many parallel Scherktype ends in the quotient. We denote by  $\widetilde{M}_{\sigma,\alpha,\beta}$  the lifting of  $\{M_{\sigma,\alpha,\beta}\}_{\sigma,\alpha,\beta}$  to  $\mathbb{S}^1 \times \mathbb{R}^2$  by forgetting its non horizontal period.

The construction is based on a gluing procedure which involves a compact part of the surface  $M_k$  with the catenoidal type ends slightly bent, which is glued with a minimal graph about a strip of finite breadth along its middle boundary curve, to one half of  $\widetilde{M}_{\sigma,\alpha,\beta}$ , (one time with  $\alpha = 0$  and a second time with  $\beta = 0$ ) along the upper boundary curve and a Scherk type surface along the lower boundary curve.

We obtain two families of properly embedded minimal surfaces in  $\mathbb{S}^1 \times \mathbb{R}^2$  with genus  $k \ge 1$ , infinitely many parallel Scherk-type and two limit ends.

## Chapter 1

## Index and nullity of the Gauss map of the Costa-Hoffman-Meeks surfaces

#### Introduction

In the years 80's and 90's the study of the index of minimal surfaces in Euclidean space has been quite active. D. Fisher-Colbrie in [8], R. Gulliver and H. B. Lawson in [10] proved independently that a complete minimal surface M in  $\mathbb{R}^3$  with Gauss map G has finite index if and only if it has finite total curvature. D. Fisher-Colbrie also observed that if M has finite total curvature its index coincides with the index of an operator  $L_{\bar{G}}$ (that is the number of its negative eigenvalues) associated to the extended Gauss map  $\bar{G}$ of  $\bar{M}$ , the compactification of M. Moreover  $N(\bar{G})$ , the null space of  $L_{\bar{G}}$ , if restricted to M consists of the bounded solutions of the Jacobi equation. The nullity, Nul( $\bar{G}$ ), that is the dimension of  $N(\bar{G})$ , and the index are invariants of  $\bar{G}$  because they are independent of the choice of the conformal metric on  $\bar{M}$ .

The computation of the index and of the nullity of the Gauss map of the Costa surface and of the Costa-Hoffman-Meeks surface of genus g = 2, ..., 37 appeared respectively in the works [30] and [29] of S. Nayatani. The aim of this work is to extend his results to the case where  $g \ge 38$ .

In [30] he studied the index and the nullity of the operator  $L_G$  associated to an arbitrary holomorphic map  $G: \Sigma \to S^2$ , where  $\Sigma$  is a compact Riemann surface. He considered a deformation  $G_t: \Sigma \to S^2$ ,  $t \in (0, +\infty)$ , with  $G_1 = G$  (see equation (1.2)) and gave lower and upper bounds for the index of  $G_t$ ,  $\operatorname{Ind}(G_t)$ , and its nullity,  $\operatorname{Nul}(G_t)$ , for t near to 0 and  $+\infty$  and t = 1. The computation of the index and the nullity in the case of the Costa surface is based on the fact that the Gauss map of this surface is a deformation for a particular value of t of the map G defined by  $\pi \circ G = 1/\wp'$ , that is its stereographic projection is equal to the inverse of the derivative of the Weierstrass  $\wp$ -function for an unit square lattice. S. Nayatani computed  $\operatorname{Ind}(G_t)$  and  $\operatorname{Nul}(G_t)$  for  $t \in (0, +\infty)$ , where G is the map defined above. So the result concerning the Costa surface follows as a simple consequence from that. He obtained that for this surface the index and the nullity are equal respectively to 5 and 4.

In [29] S. Nayatani extended the last result treating the case of the Costa-Hoffman-Meeks surface of genus g but only for  $2 \leq g \leq 37$ . He obtained that the index is equal to 2g + 3 and the nullity is equal to 4. Here we will show that these results continue to hold also for  $g \geq 38$ .

J. Pérez and A.Ros in [34] call a minimal surface non degenerate if the bounded Jacobi functions about the surface are induced by the isometries of the ambient space. As consequence of the works [29] and [30], the Costa-Hoffman-Meeks surface was known to be non degenerate with respect to this definition, but only for  $1 \leq g \leq 37$ .

The result of S. Nayatani about the nullity of the Gauss map of the Costa-Hoffman-Meeks surface is essential for the construction due to L. Hauswirth and F. Pacard [11] of a family of minimal surfaces with two limit ends asymptotic to half Riemann minimal surfaces and of genus g with  $1 \leq q \leq 37$ . Their construction is based on a gluing procedure which involves the Costa-Hoffman-Meeks surface of genus q and two half Riemann minimal surfaces. In particular the authors needed show the existence of a family of minimal surfaces close to the Costa-Hoffman-Meeks surface, invariant under the action of the symmetry with respect to the vertical plane  $x_2 = 0$ , having one horizontal end asymptotic to the plane  $x_3 = 0$  and having the upper and the lower end asymptotic (up to translation) respectively to the upper and the lower end of the standard catenoid whose axis of revolution is directed by the vector  $\sin \theta e_1 + \cos \theta e_3$ ,  $\theta \leq \theta_0$  with  $\theta_0$  sufficiently small. That was obtained by Schauder fixed point theorem and using the fact that the nullity of the Gauss map of the surface is equal to 4. In [11] the authors refer to this last result as a non degeneracy property of the Costa-Hoffman-Meeks surface. It is necessary to remark that here the choice of working with symmetric deformations of the surface with respect to the plane  $x_2 = 0$ , has a key role. Because of the restriction on the value of the genus which affects the result of S. Nayatani, it was not possible to prove the existence of this family of minimal surfaces for higher values of the genus.

So one of the consequences of our work is the proof of the non degeneracy of the Costa-Hoffman-Meeks surface for  $g \ge 1$  in the sense of the definition given in [34] and also, only in a symmetric setting, in [11]. So we can state that the family of examples constructed by L. Hauswirth and F. Pacard exists for all the values of the genus. Moreover our result allows us to show in chapter 2 the existence of a family of minimal deformations of the Costa-Hoffman-Meeks surface for each value of the genus.

The author wishes to thank S. Nayatani for having provided the background computations on which are based the results about the Costa-Hoffman-Meeks surfaces contained in [29].

The author is grateful to his thesis director, L. Hauswirth, for his support and for having brought this problem to his attention.

#### **1.1** Preliminaries

Let M be a complete oriented minimal surface in  $\mathbb{R}^3$ . The Jacobi operator of M is

$$L = -\triangle + 2K$$

where  $\Delta$  is the Laplace-Beltrami operator and K is the Gauss curvature. Moreover we suppose that M has finite total curvature. Then M is conformally equivalent to a compact Riemann surface with finitely many punctures and the Gauss map  $G: M \to S^2$  extends to the compactified surface holomorphically. So in the following we will pay attention to a generic compact Riemann surface, denoted by  $\Sigma$  and  $G: \Sigma \to S^2$  a not constant holomorphic map, where  $S^2$  is the unit sphere in  $\mathbb{R}^3$  endowed with the complex structure induced by the stereographic projection from the north pole (denoted by  $\pi$ ). We fix a conformal metric  $ds^2$  on  $\Sigma$  and consider the operator  $L_G = -\Delta + |dG|^2$ , acting on functions on  $\Sigma$ .

We denote by N(G) the kernel of  $L_G$ . We define Nu(G), the nullity of G, as the dimension of N(G). Since  $L(G) = \{a \cdot G \mid a \in \mathbb{R}^3\}$  is a three dimensional subspace of N(G), then  $Nu(G) \ge 3$ . We denote the index of G, that is the number of negative eigenvalues of  $L_G$ , by Ind(G). The index and the nullity are invariants of the map G: they are independent of the metric on the surface  $\Sigma$ . So we can consider on  $\Sigma$  the metric induced by G from  $S^2$ .

N. Ejiri and M. Kotani in [6] and S. Montiel and A. Ros in [25] proved that a non linear element of N(G) is expressed as the support function of a complete branched minimal surface with planar ends whose extended Gauss map is G. In the following we will review briefly some results contained in [25] used by S. Nayatani in [30].

We will use some definitions and concepts of the algebraic geometric. They are recalled in subsection 1.5.1.

Let  $\gamma$  be the meromorphic function defined by  $\pi \circ G$ . Let  $p_j$  and  $r_i$  be respectively the poles and the branch points of  $\gamma$ . We denote by  $P(G) = \sum_{i=1}^{\nu} n_j p_j$ ,  $S(G) = \sum_{i=1}^{\mu} m_i r_i$  respectively the polar and ramification divisor of  $\gamma$ . Here  $n_j, m_i$  denote, respectively, the

multiplicity of the pole  $p_j$  and the multiplicity with which  $\gamma$  takes its value at  $r_i$ . We define on the surface  $\Sigma$  the divisor

$$D(G) = S(G) - 2P(G)$$

and introduce the vector space  $\overline{H}(G)$  (see [25], theorem 4)

$$\bar{H}(G) = \left\{ \omega \in H^{0,1}(k_{\Sigma} + D(G)) \mid \operatorname{Res}_{r_i} \omega = 0, 1 \leq i \leq \mu, \\ \operatorname{Re} \int_{\alpha} (1 - \gamma^2, i(1 + \gamma^2), 2\gamma) \omega = 0, \, \forall \alpha \in H_1(\Sigma, \mathbb{Z}) \right\},$$

where  $k_{\Sigma}$  is a canonical divisor of  $\Sigma$  and  $H_1(\Sigma, \mathbb{Z})$  is the first group of homology of  $\Sigma$ . Suppose that the divisor D has an expression of the form  $\sum n_j v_j - \sum m_i u_i$ , with  $n_j, m_i \in \mathbb{N}$ . An element of  $H^{0,1}(D)$  can be expressed as fdz, where f is a meromorphic function on  $\Sigma$  with poles of order not bigger than  $n_j$  at  $v_i$  and zeroes of order not smaller than  $m_i$  at  $u_i$ . Equivalently, if gdz, where g is a meromorphic function, is the differential form associated with the divisor D, the product fg must be holomorphic.

For  $\omega \in \overline{H}(G)$ , let  $X(\omega) : \Sigma \setminus \{r_1, \ldots, r_\mu\} \to \mathbb{R}^3$  be the conformal immersion defined by

$$X(\omega)(p) = Re \int^{p} (1 - \gamma^2, i(1 + \gamma^2), 2\gamma)\omega.$$

Then  $X(\omega) \cdot G$ , the support function of  $X(\omega)$ , extends over the ramification points  $r_1, \ldots, r_{\mu}$  smoothly and thus gives an element of N(G). Conversely, every element of N(G) is obtained in this way. In fact the map

$$\begin{array}{rccc} i: \bar{H}(G) & \to & N(G)/L(G) \\ \omega & \to & [X(\omega) \cdot G] \end{array}$$
 (1.1)

is an isomorphism. This result, used in association with the Weierstrass representation formula, gives a description of the space N(G). To obtain the dimension of N(G) it is sufficient to compute the dimension of  $\bar{H}(G)$ . Since the dimension of L(G) is equal to 3, then  $Nul(G) = 3 + \dim \bar{H}(G)$ .

We denote by  $A_t$  a one parameter family  $(0 < t < +\infty)$  of conformal diffeomorphisms of the sphere  $S^2$  defined by

$$\pi \circ A_t \circ \pi^{-1} w = tw, \quad w \in \mathbb{C} \cup \{\infty\}.$$

We define for  $0 < t < \infty$ 

$$G_t = A_t \circ G. \tag{1.2}$$

S. Nayatani in [30] gave lower and upper bounds for the index and, applying the method recalled above, for the nullity of  $G_t, t \in (0, \infty)$ , a deformation of an arbitrary holomorphic map  $G : \Sigma \to S^2$ , where  $\Sigma$  is a compact Riemann surface. In the same work, choosing appropriately the map G and the surface  $\Sigma$ , he computed the index and the nullity for the Gauss map of the Costa surface. In fact the extended Gauss map of this surface is a deformation of G for a particular value of t. We describe briefly the principal steps to get this result.

Firstly it is necessary to study the vector space  $\overline{H}(G_t)$ . A differential  $\omega \in H^{0,1}(k_{\Sigma} + D(G))$ with null residue at the ramification points, is an element of  $\overline{H}(G_t)$  if and only if the pair  $(t\gamma, \omega)$  defines a branched minimal surface by the Weierstrass representation. If one sets  $\gamma = 1/\wp'$  then there exist only two values of t, denoted by t' < t'', for which the condition above is verified and moreover dim  $H(G_t) = 1$ . In other words, thanks to the characterization of the non linear elements of  $N(G_t)$  by the isomorphism described by (1.7), if t = t', t'',  $\operatorname{Nul}(G_t) = 4$ . As for the index, if t = t', t'' then  $\operatorname{Ind}(G_t) = 5$ . Since  $G_{t''}$ is the extended Gauss map of the Costa surface, one can state:

**Theorem 3.** Let  $\overline{G}$  be the extended Gauss map of the Costa surface. Then

$$\operatorname{Nul}(\overline{G}) = 4$$
,  $\operatorname{Ind}(\overline{G}) = 5$ .

The same author in [29] treated the more difficult case of the Costa-Hoffman-Meeks surfaces of genus  $2 \leq g \leq 37$ . That is the subject of next section.

#### 1.2 The case of the Costa-Hoffman-Meeks surface of genus smaller than 38

In this section we expose some of the background details at the base of section 3 of the work [29]. S. Nayatani provided them to us in [31].

We denote by  $M_g$  the Costa-Hoffman-Meeks surface of genus g. Let  $\Sigma_g$  be the compact Riemann surface

$$\Sigma_g = \{ (z, w) \in (\mathbb{C} \cup \{\infty\})^2 \, | \, w^{g+1} = z^g (z^2 - 1) \}$$
(1.3)

and let  $Q_0 = (0,0), P_+ = (1,0), P_- = (-1,0), P_{\infty} = (\infty,\infty)$ . It is known that  $M_g = \Sigma_g \setminus \{P_+, P_-, P_{\infty}\}.$ 

The following result describes the properties of symmetry of  $M_g$  and  $\Sigma_g$ .

Lemma 4. ([14]) Consider the conformal mappings

$$\kappa(z,w) = (\bar{z},\bar{w}) \quad \lambda(z,w) = (-z,\rho w), \tag{1.4}$$

where  $\rho = e^{\frac{i\pi g}{g+1}}$  of  $(\mathbb{C} \cup \{\infty\})^2$ . The map  $\kappa$  is of order 2 and  $\lambda$  is of order 2g+2. The group generated by  $\kappa$  and  $\lambda$  is the dihedral group  $D_{2g+2}$ . This group of conformal diffeomorphisms leaves  $M_g$  invariant, fixes both  $Q_0$  and  $P_{\infty}$  and extend to  $\Sigma_g$ . Also  $\kappa$  fixes the points  $P_{\pm}$  while  $\lambda$  interchanges them.

We set  $\gamma(w) = w$ . Let  $G: \Sigma_q \to S^2$  be the holomorphic map defined by

$$\pi \circ G(z, w) = \gamma(w). \tag{1.5}$$

We denote by  $r_i$ ,  $i = 1, ..., \mu$ , the ramification points of  $\gamma$  and by R(G) the divisor  $\sum_{i=1}^{\mu} r_i$ . Theorem 5 of [25] shows that the space N(G)/L(G), that we have introduced in previous section, is also isomorphic to a space of meromorphic quadratic differentials. This alternative description of N(G)/L(G) that we present in the following, was adopted by S. Nayatani in [29]. We start defining the vector spaces  $\hat{H}(G)$  and H(G).

$$\hat{H}(G) = \left\{ \sigma \in H^{0,2}(2k_{\Sigma} + R(G)) \mid \operatorname{Res}_{r_i} \frac{\sigma}{d\gamma} = 0, i = 1, \dots, \mu \right\},$$

$$H(G) = \left\{ \sigma \in \hat{H}(G) \mid \operatorname{Re} \int_{\alpha} (1 - \gamma^2, i(1 + \gamma^2), 2\gamma) \frac{\sigma}{d\gamma} = 0, \, \forall \alpha \in H_1(\Sigma, \mathbb{Z}) \right\},$$
(1.6)

where  $k_{\Sigma}$  is a canonical divisor of  $\Sigma$ . We remark that the elements of  $H^{0,2}(2k_{\Sigma} + R(G))$ are quadratic differentials (see subsection 1.5.1). Since hereafter we will work only with quadratic differentials, we can set  $H^{0}(\cdot) = H^{0,2}(\cdot)$  to simplify the notation. If we suppose that the divisor  $2k_{\Sigma} + R(G)$  has an expression of the form  $\sum n_{j}v_{j} - \sum m_{i}u_{i}$ , with  $n_{j}, m_{i} \in \mathbb{N}$ , an element of  $H^{0}(2k_{\Sigma} + R(G))$  can be expressed as  $f(dz)^{2}$ , where f is a meromorphic function on  $\Sigma$  with poles of order not bigger than  $n_{j}$  at  $v_{i}$  and zeroes of order not smaller than  $m_{i}$  at  $u_{i}$ . Equivalently, if  $g(dz)^{2}$ , where g is a meromorphic function, is the differential form associated with the divisor  $2k_{\Sigma} + R(G)$ , the product fg must be holomorphic.

For 
$$\sigma \in H(G)$$
, let  $X(\sigma) : \Sigma \setminus \{r_1, \dots, r_\mu\} \to \mathbb{R}^3$  be the conformal immersion defined by  
 $X(\sigma)(p) = Re \int^p (1 - \gamma^2, i(1 + \gamma^2), 2\gamma) \frac{\sigma}{d\gamma}.$ 

Then  $X(\sigma) \cdot G$ , the support function of  $X(\sigma)$ , extends over the ramification points  $r_1, \ldots, r_{\mu}$ smoothly and thus gives an element of N(G). Conversely, every element of N(G) is obtained in this way. In fact the map

$$\begin{array}{rccc} i: H(G) & \to & N(G)/L(G) \\ \sigma & \to & [X(\sigma) \cdot G] \end{array}$$
 (1.7)

is an isomorphism. So to obtain the dimension of N(G) it is sufficient to compute the dimension of H(G). We recall that the dimension of L(G) is equal to 3, so Nul(G) =  $3 + \dim H(G).$ 

Since the extended Gauss map of the Costa-Hoffman-Meeks surfaces is a deformation in the sense of the definition (1.2) of the map G, we need to study the space  $H(G_t)$ . From (1.6) and (1.2) it is clear that  $\hat{H}(G) = \hat{H}(G_t)$  and

$$H(G_t) = \left\{ \sigma \in \hat{H}(G_t) \mid \operatorname{Re} \int_{\alpha} (1 - t^2 \gamma^2, i(1 + t^2 \gamma^2), 2t\gamma) \frac{\sigma}{d\gamma} = 0, \, \forall \alpha \in H_1(\Sigma_g, \mathbb{Z}) \right\}.$$

Long computations ([31], see subsection 1.5.2 for some details) show that a basis of the differentials of the form  $\sigma/d\gamma$ , where  $\sigma \in \hat{H}(G) = \hat{H}(G_t)$ , and whose residue at the ramification points of  $\gamma(w) = w$  is zero, is formed by

$$\omega_k^{(1)} = \frac{z^{k-1}}{w^k} \frac{dz}{w}, \quad \text{with} \quad k = 0, \dots, g-1,$$
$$\omega_k^{(2)} = \frac{((k-2)z^2 - kA^2)}{(z^2 - A^2)^2} \left(\frac{z}{w}\right)^{k-1} \frac{dz}{w}, \quad \text{with} \quad k = 0, \dots, g,$$
$$\omega_k^{(3)} = \frac{((k-2)z^2 - kA^2)}{w(z^2 - A^2)^2} \left(\frac{z}{w}\right)^{k-1} \frac{dz}{w}, \quad \text{with} \quad k = 0, \dots, g-1,$$
re  $A = \sqrt{\frac{g}{g+2}}.$ 

when

Now we put attention to the space  $H(G_t)$ . We recall that we are interested in the computation of its dimension. By the definition of  $H(G_t)$ , a differential  $\sigma \in H(G_t)$  belongs to  $H(G_t)$  if and only if  $\forall \alpha \in H_1(\Sigma_g, \mathbb{Z})$  the differential form  $\omega = \frac{\sigma}{d\gamma} = \frac{\sigma}{dw}$  satisfies

$$\int_{\alpha} \omega = t^2 \overline{\int_{\alpha} \gamma^2(w)\omega},\tag{1.8}$$

$$Re \int_{\alpha} \gamma(w)\omega = 0.$$
 (1.9)

If these two conditions are satisfied then  $(\gamma, w)$  are the Weierstrass data of a branched minimal surface. Of course, it is sufficient to impose that these equations are satisfied when  $\alpha$ varies between the elements of a basis of  $H_1(\Sigma_g, \mathbb{Z})$ . The convenient basis of  $H_1(\Sigma_g, \mathbb{Z})$  is constructed as follows. Let  $\beta(t) = \frac{1}{2} + e^{i2\pi t}, 0 \leq t \leq 1$ . Let  $\tilde{\beta}(t) = (\beta(t), w(\beta(t)))$  be a lift of  $\beta$  to  $\Sigma_q$  such that, for example,  $\tilde{\beta}(0) = (\frac{3}{2}, w(0))$ , with  $w(0) \in \mathbb{R}$ . As stated in lemma 4 the group of conformal diffeomorphisms of  $\Sigma_q$  is isomorphic to the dihedral group  $D_{2q+2}$ . The collection  $\{\lambda^l \circ \beta, l = 0, \dots, 2g-1\}$ , where  $\lambda$  is the generator of  $D_{2g+2}$  of order 2g+2,

is a basis of  $H_1(\Sigma_g, \mathbb{Z})$ .

Now we must impose (1.8) and (1.9) for  $\alpha = \lambda^l \circ \tilde{\beta}$ , with  $l = 0, \ldots, 2g - 1$ . To do that we collapse  $\beta$  to the unit interval. In other terms we deform continuously  $\beta$  in such a way the limit curve is the union of two line segments lying on the real line. We set

$$\omega = \sum_{0}^{g-1} c_k^{(1)} \omega_k^{(1)} + \sum_{0}^{g} c_k^{(2)} \omega_k^{(2)} + \sum_{0}^{g-1} c_k^{(3)} \omega_k^{(3)},$$

where  $c_k^{(i)} \in \mathbb{C}$ .

Taking into account these assumptions, it is possible to show that the equation (1.8), if the genus g is 2, is equivalent to the following system of four equations (see subsection 1.5.3)

$$\begin{cases}
f_0 = -t^2 \bar{h}_0 \\
f_1 = 0 \\
p_1 = -t^2 \bar{q}_1 \\
p_2 = -t^2 \bar{q}_0.
\end{cases}$$
(1.10)

If  $g \ge 3$  there are the following additional 2g - 4 equations to consider

$$\begin{cases} f_k = -t^2 \bar{q}_{g-k+2} \\ p_{g-k+2} = -t^2 \bar{h}_k \end{cases}$$
(1.11)

where  $k = 2, \ldots, g - 1$  and

$$f_0 = \frac{(g+2)^2}{2(g+1)} c_0^{(3)} \sin\left(\frac{\pi}{g+1}\right) K_0,$$

$$f_k = \left(-c_k^{(1)} + \frac{(g+2)(g+2+k)}{2(g+1)} c_k^{(3)}\right) \sin\left(\frac{(k+1)\pi}{g+1}\right) K_k, \quad k = 1, \dots, g-1,$$

$$h_0 = \frac{(g+2)^2}{2(g+1)} c_0^{(3)} \sin\left(\frac{-\pi}{g+1}\right) J_0,$$

$$h_k = \left(c_k^{(1)} + \frac{(g+2)(g+2-k)}{2(g+1)} c_k^{(3)}\right) \sin\left(\frac{(k-1)\pi}{g+1}\right) J_k, \quad k = 2, \dots, g-1,$$

$$p_k = -\frac{(g+2)k}{2(g+1)} c_k^{(2)} \sin\left(\frac{k\pi}{g+1}\right) I_k, \quad k = 1, \dots, g,$$

$$q_k = \frac{(g+2)(2g+4-k)}{2(g+1)} c_k^{(2)} \sin\left(\frac{(k-2)\pi}{g+1}\right) L_k, \quad k = 0, 1, 3, \dots, g,$$

and

$$I_m = \frac{g+1}{m} \frac{\Gamma\left(1 + \frac{m}{2(g+1)}\right) \Gamma\left(1 - \frac{m}{g+1}\right)}{\Gamma\left(1 - \frac{m}{2(g+1)}\right)},$$
$$J_m = \frac{g+1}{g-m+2} \frac{\Gamma\left(\frac{1}{2} + \frac{m-1}{2(g+1)}\right) \Gamma\left(1 - \frac{m-1}{g+1}\right)}{\Gamma\left(\frac{1}{2} - \frac{m-1}{2(g+1)}\right)},$$
$$K_m = J_{m+2},$$

$$L_m = \frac{m-2}{2g - m + 4} I_{m-2}.$$

The equation (1.9) if the genus g is 2, is equivalent to the following system of two equations (see subsection 1.5.3)

$$\begin{cases}
 d_1 = 0 \\
 e_2 = \bar{e}_0.
\end{cases}$$
(1.12)

If  $g \ge 3$  there are the following additional g - 2 equations to consider

$$d_k = \bar{e}_{g-k+2} \tag{1.13}$$

where  $k = 2, \ldots, g - 1$ , and

$$d_k = \left(c_k^{(1)} - \frac{k(g+2)}{2(g+1)}c_k^{(3)}\right)\sin\left(\frac{k\pi}{g+1}\right)I_k, \quad k = 1, \dots, g-1,$$
$$e_k = \frac{(g+2)(g+2-k)}{2(g+1)}c_k^{(2)}\sin\left(\frac{(k-1)\pi}{g+1}\right)J_k, \quad k = 0, 2, \dots, g.$$

We are looking for the values of t such that the previous systems have non trivial solutions in terms of  $c_i^{(j)}$ . Only for these special values of t it holds dim  $H(G_t) > 0$  or equivalently  $\operatorname{Nul}(G_t) > 3$ .

We start with the analysis of the system (1.10). This system admits non trivial solutions if and only if t assumes three values denoted by  $t_1, t_2, t_3$ . Obviously they are functions of g.

If we set  $s = \frac{1}{g+1}$  then we can write

$$t_{1} = \sqrt{\frac{K_{0}}{J_{0}}} = \frac{\sqrt{1-s^{2}}}{2} \sqrt{\frac{\Gamma(1-s)}{\Gamma(1+s)}} \frac{\Gamma\left(1-\frac{s}{2}\right)}{\Gamma\left(1+\frac{s}{2}\right)},$$

$$t_{2} = \sqrt{\frac{I_{1}}{(2g+3)L_{1}}} = \sqrt{\frac{\Gamma(1-s)}{\Gamma(1+s)}} \frac{\Gamma\left(1+\frac{s}{2}\right)}{\Gamma\left(1-\frac{s}{2}\right)},$$

$$t_{3} = \sqrt{\frac{I_{2}J_{0}}{gL_{0}K_{0}}} = \frac{2}{1-s} \sqrt{\left(\frac{\Gamma(1+s)}{\Gamma(1-s)}\right)^{3}} \sqrt{\frac{\Gamma(1-2s)}{\Gamma(1+2s)}} \frac{\Gamma(3/2-s/2)}{\Gamma(1/2+s/2)}$$

We recall that if  $g \ge 3$  there are other equations to consider. They are

$$\begin{cases} f_k = -t^2 \bar{q}_{g-k+2} \\ p_{g-k+2} = -t^2 \bar{h}_k \\ d_k = \bar{e}_{g-k+2} \end{cases}$$

where k = 2, ..., g - 1. Thanks to the particular structure of the equations, it is possible to study separately for each set of three equations the existence of solutions. Each set of three equations admits non trivial solutions if and only if the following matrix has determinant equal to zero

$$\begin{pmatrix} -K_k & (g+2+k)K_k & (g+2+k)t^2L_{g-k+2} \\ t^2J_k & (g+2-k)t^2J_k & (g+2-k)I_{g-k+2} \\ I_k & -kI_k & -kJ_{g-k+2} \end{pmatrix}.$$

After the change of variable l = g - k + 1 so that  $2 \leq l \leq g - 1$ , it is possible to show that the determinant is

$$-(g+2)(at^4+bt^2+c), (1.14)$$

with

$$a = (2g - l + 3)I_{g-l+1}J_{g-l+1}L_{l+1}$$
  

$$b = -2(g - l + 1)J_{l+1}J_{g-l+1}K_{g-l+1}$$
  

$$c = (l + 1)I_{g-l+1}I_{l+1}K_{g-l+1}.$$

We are interested in finding the positive values of t such that

$$at^4 + bt^2 + c = 0. (1.15)$$

To simplify the notation we introduce the following three functions

$$F(v) = \left(\frac{\Gamma(\frac{1}{2} + \frac{v}{2})}{\Gamma(\frac{1}{2} - \frac{v}{2})}\right)^2 \frac{\Gamma(1-v)}{\Gamma(1+v)},$$

$$I(v) = \left(\frac{\Gamma(1-\frac{v}{2})}{\Gamma(1+\frac{v}{2})}\right)^2 \frac{\Gamma(1+v)}{\Gamma(1-v)},$$
$$L(v) = \left(\frac{\Gamma(1+\frac{v}{2})}{\Gamma(1-\frac{v}{2})}\right)^2 \frac{\Gamma(1-v)}{\Gamma(1+v)} = \frac{1}{I(v)}.$$

The discriminant  $b^2 - 4ac$  of the equation (1.15), seen like an equation of degree two in the variable  $t^2$ , is negative if and only if  $X = b^2/4ac < 1$ . It is possible to show that

$$X = \frac{l^2}{l^2 - 1} F^2\left(\frac{l}{g+1}\right) I\left(\frac{l-1}{g+1}\right) I\left(\frac{l+1}{g+1}\right).$$
 (1.16)

S. Nayatani in [31] showed that if  $2 \leq g \leq 37$ , then X < 1 and as consequence the equation (1.15) has not any solution since its discriminant is negative. Then dim  $H(G_t) > 0$  only for  $t = t_1, t_2, t_3$ . Summarizing we can state (see [29] for other details):

**Theorem 5.** If  $2 \leq g \leq 37$  and  $t \in (0, +\infty)$ , then

$$\operatorname{Nul}(G_t) = \begin{cases} 4 & if \quad t = t_1, t_2 \\ 5 & if \quad t = t_3 \\ 3 & elsewhere. \end{cases}$$

Since the extended Gauss map of the Costa-Hoffman-Meeks surfaces is exactly  $G_{t_2}$ , it is possible to state that the null space of the Jacobi operator of  $M_g$  has dimension equal to 4 for  $2 \leq g \leq 37$ .

Other values of t for which  $Nul(G_t) > 3$  are admitted only if  $g \ge 38$ . In [29] S. Nayatani conjectured these values were bigger than  $t_3$ . The proof of the conjecture and its consequences will be showed in sections 1.3 and 1.4.

#### **1.3** The case $g \ge 38$

Thanks the previous observations it is clear that if we assume  $g \ge 38$  It is possible to prove that X is a decreasing function in the variables

$$x = \frac{l}{g+1}, y = \frac{l+1}{g+1}, z = \frac{l-1}{g+1}$$

with  $2 \leq l \leq g-1$ . We recall that we have set  $s = \frac{1}{g+1}$ . We know that for l = 2 and g = 37 the discriminant of the equation (1.15) is negative. For these values of l and g the variables x, y, z, s are respectively equal to  $x_{max} = 2s_{max}, y_{max} = 3s_{max}, z_{max} = s_{max} = 1/38$ .

Then we will study the solutions of (1.15) for  $i \in [0, i_{max}]$  (we call these admissible values) where *i* denotes x, y, z, s because for bigger values of the three variables the discriminant continues to be negative and so the equation (1.15) does not admit solutions. Numerical tests show that the value  $i_{max}$ , become smaller as *g* is bigger. Since it is not possible to explicit the dependence of  $i_{max}$  on *g* we shall work with constant quantities.

All the solutions of (1.15), that we denote by  $t_{\pm}(l,g)$ , satisfy  $t_{\pm}^2(l,g) = T_1 \pm T_2$ , with

$$T_{1} = \frac{l}{l-1} F(x) I(z)$$
(1.17)

and

$$T_{2} = \sqrt{\left(\frac{l}{l-1}\right)^{2} F^{2}(x) I^{2}(z) - \frac{l+1}{l-1}L(y) I(z)}.$$
(1.18)

We will prove that, for  $0 \leq \frac{l}{g+1} \leq x_{max} = \frac{2}{38}$ , with  $2 \leq l \leq g-1$  and  $g \geq 38$ , it holds

$$t_3^2(s) < t_-^2(l,g). \tag{1.19}$$

We need study the behaviour of the functions  $F, I, L, F^2, I^2$  that appear in (1.17) and (1.18). This aim is pursued by the use of first order series of these functions. The Mac-Laurin series of the functions  $F(x), G(z), L(y), F^2(x), I^2(z)$  for admissible values of x, y, z are

$$F(x) = 1 + R_F(d_1x)x, \quad I(z) = 1 + R_I(d_2z)z, \quad L(y) = 1 + R_L(d_3y)y,$$
(1.20)  
$$F^2(x) = 1 + R_{F^2}(c_1x)x, \quad I^2(x) = 1 + R_{I^2}(c_2x)x,$$

where  $c_i, d_i \in (0, 1)$ . So we can write

$$F(x)I(z) = 1 + R_{FI}(x,z), \ F^2(x)I^2(z) = 1 + R_{F^2I^2}(x,z), \ L(y)I(z) = 1 + R_{LI}(y,z),$$

with

$$R_{FI}(x,z) = R_F(d_1x)x + R_I(d_2z)z + R_F(d_1x)R_I(d_2z)xz,$$
  

$$R_{F^2I^2}(x,z) = R_{F^2}(c_1x)x + R_{I^2}(c_2z)z + R_{F^2}(c_1x)R_{I^2}(c_2z)xz,$$
  

$$R_{LI}(y,z) = R_L(d_3y)y + R_I(d_2z)z + R_I(d_2z)R_L(d_3y)zy.$$

In the following  $\psi(x)$  the digamma function. It is related to  $\Gamma(x)$ , the gamma function, by

$$\psi(x) = \frac{d}{dx} \left( \ln \Gamma(x) \right).$$

For the properties of this special function we refer to [1].

The following proposition gives useful properties of the functions just introduced.

**Proposition 6.** If  $x \in [0, x_{max}]$ ,  $z \in [0, z_{max}]$ , and  $y \in [0, y_{max}]$ , the following assertions hold:

- 1.  $R_F(x) < 0$
- 2.  $R_I(z) \leq 0$
- 3.  $R_L(y) \ge 0$
- 4.  $(R_F)'_x(x) > 0$
- 5.  $min(R_I)'_z(z) = -0.095\cdots$
- 6.  $R_{FI}(x,z) \ge Cx$  with  $C = -4 \ln 2$
- 7.  $R_{LI}(y, z) \ge 0$
- 8.  $R_{I^2}(z) \leq 0$
- 9.  $W(x) = R_{F^2}(x) < 0$
- 10.  $W'_x(x) > 0$ , so  $R_{F^2}(x)$  is an increasing function
- 11.  $W''_{xx}(x) < 0$
- 12.  $W_{xxx}^{\prime\prime\prime}(x) > 0$
- 13. If we set Y(x) = x W(x), then  $Y'_x(x) \leq 0$
- 14.  $Y''_{xx}(x) \ge 0$
- 15.  $Y_{xxx}''(x) \leq 0.$

#### Proof.

1.  $R_F(x) = F'_x(x) = F(x)\Psi_F(x)$ , where

$$\Psi_F(x) = -\psi(1-x) - \psi(1+x) + \psi\left(\frac{1}{2} - \frac{x}{2}\right) + \psi\left(\frac{1}{2} + \frac{x}{2}\right).$$

We observe that

$$\Psi_F(x) = 2\sum_{k=0}^{\infty} \frac{1}{(2k)!} \left(\frac{1}{2^{2k}} \psi^{(2k)}\left(\frac{1}{2}\right) - \psi^{(2k)}(1)\right) x^{2k}.$$

Since  $\Psi_F(0) = 2\psi\left(\frac{1}{2}\right) - 2\psi(1) = -4\ln 2$ ,  $\psi^{(2k)}(1) < 0$  and  $\psi^{(2k)}\left(\frac{1}{2}\right) = (2^{2k+1} - 1)\psi^{(2k)}(1) < 0$ , if  $k \ge 1$  (see formulas 6.4.2 and 6.4.4 of [1]), we can conclude that  $\Psi_F(x) < 0$  and it is a decreasing function. Since F(x) > 0 then  $R_F(x) < 0$  and F(x) is a decreasing function.

2.  $R_I(z) = I'_z(z) = I(z)\Psi_I(z)$ , where

$$\Psi_I(z) = \psi(1-z) + \psi(1+z) - \psi\left(1-\frac{z}{2}\right) - \psi\left(1+\frac{z}{2}\right).$$

We observe that

$$\Psi_I(z) = 2\sum_{k=1}^{\infty} \frac{1}{(2k)!} \psi^{(2k)}(1) \left(1 - \frac{1}{2^{2k}}\right) z^{2k}$$

Since  $\psi^{(2k)}(1) < 0$  for  $k \ge 1$  then  $\Psi_I(z) \le 0$  and it is a decreasing function. Since I(z) > 0 then  $R_I(z) \le 0$ .

- 3.  $R_L(y) = L'_y(y) = L(y)\Psi_L(y)$ , where  $\Psi_L(y) = -\Psi_I(y)$ . Then  $\Psi_L(y) \ge 0$  and it is an increasing function. Since L(y) = 1/I(y) > 0, then  $R_L(y) \ge 0$ .
- 4. The derivative of  $R_F$  is  $F''_{xx}(x) = F(x)(\Psi_F^2(x) + (\Psi_F)'_x(x))$ . Since  $\Psi_F(x) < 0$  and it is a decreasing function,  $\Psi_F^2(x) > 0$  and increasing. It holds  $\Psi_F^2(x) \ge \Psi_F^2(0) = 16 \ln^2 2$ .

$$(\Psi_F)'_x(x) = 2\sum_{k=1}^{\infty} \frac{1}{(2k-1)!} \left(\frac{1}{2^{2k}} \psi^{(2k)}\left(\frac{1}{2}\right) - \psi^{(2k)}(1)\right) x^{2k-1}$$

All the coefficients of the series are negative (see the point 1) so  $(\Psi_F)'_x(x) \leq 0$  and it is a decreasing function. In particular  $(\Psi_F)'_x(x) \geq (\Psi_F)'_x(x_{max}) = -0.19 \cdots$ . Since F(x) > 0 and it is a decreasing function we can conclude that

$$F''_{xx}(x) \ge F(x_{max})(\Psi_F^2(0) + (\Psi_F)'_x(x_{max})) = 6.4 \cdots$$

5. The derivative of  $R_I$  is  $I''_{zz}(z) = I(z)(\Psi_I^2(z) + (\Psi_I)'_z(z))$ . Since  $\Psi_I(z) \leq 0$  and it is a decreasing function (see the point 2),  $\Psi_I^2(z) \geq 0$  and increasing. It holds  $\Psi_I^2(z) \leq \Psi_I^2(z_{max}) = 1.5 \cdots 10^{-6}$ .

$$(\Psi_I)'_z(z) = 2\sum_{k=1}^{\infty} \frac{1}{(2k-1)!} \psi^{(2k)}(1) \left(1 - \frac{1}{2^{2k}}\right) z^{2k-1}.$$

All the coefficients of the series are negative so  $(\Psi_I)'_z(z) \leq 0$  and it is a decreasing function. In particular  $(\Psi_I)'_z(z) \geq (\Psi_I)'_z(z_{max}) = -0.095 \cdots$ . Since I(z) > 0 and it is a decreasing function we can conclude that

$$I_{zz}'' \ge I(z_{max})(\Psi_I^2(0) + (\Psi_I)_z'(z_{max})) = -0.095 \cdots$$

6. Since  $R_F < 0$  and  $R_I \leq 0$ , it holds that

$$R_{FI}(x,z) \ge R_F(d_1x)x + R_I(d_2z)z,$$

where  $d_i \in (0, 1)$ . In the point 4 we have proved that  $R_F$  is an increasing function and we have computed the minimum value of its derivative. Moreover it is also clear thanks to the point 5 that this value is always bigger than the maximum of the absolute value of the derivative of  $R_I$ . Now it is sufficient to remember that the variables t and z are not independent. We can conclude that  $R_{FI}$  is an increasing function. Then  $R_{FI} \ge R_F(0)x + R_I(0)z = Cx$ .

7. We recall that  $R_{LI}(y, z) = L(y)I(z) - 1$ , L(t) = 1/I(t) and

$$y = \frac{l+1}{g+1} > \frac{l-1}{g+1} = z.$$

We want to prove that  $L(y)I(z) - 1 \ge 0$  or equivalently  $L(y) \ge 1/I(z)$ . But thanks to the point 3, L is an increasing function, so

$$L(y) > L(z) = \frac{1}{I(z)}$$

- 8.  $R_{I^2}(z) = (I^2)'_z(z) = 2I^2(z)\Psi_I(z)$ . From the proof of the point 2,  $\Psi_I(z) \leq 0$  and it is a decreasing function. Since  $2I^2(z) > 0$ , then also  $R_{I^2}(z) \leq 0$ .
- 9.  $W(x) = (F^2)'_x(x) = 2F^2(x)\Psi_F(x)$ . In the point 1 we have observed that  $\Psi_F(x)$  is a negative and decreasing function. Since  $2F^2(x) > 0$ , then also W(x) is a negative function.
- 10.  $W'_x(x) = F^2 \left(4\Psi_F^2(x) + 2(\Psi_F)'_x(x)\right)$ . Since  $\Psi_F(x) < 0$  and it is a decreasing function,  $\Psi_F^2(x)$  is a positive and increasing function. In the proof of the point 4 we observed that  $(\Psi_F)'_x(x) \leq 0$  and it is a decreasing function. Since  $2(\Psi_F)'_x(x_{max}) = -0.38 \cdots$ and  $4\Psi_F^2(x) \geq 4\Psi_F^2(0) = 64 \ln^2 2 = 30.74 \cdots$ , we can conclude that  $W'_x(x) > 0$ .
- 11. The explicit expression of  $W''_{xx}$  is

$$W_{xx}'' = \frac{1}{2}F^2(x)\left(16\Psi_F^3(x) + 24\Psi_F(x)(\Psi_F)_x'(x) + 4(\Psi_F)_{xx}''(x)\right)$$

In the proof of the point 1 we observed that  $\Psi_F(x)$  is a negative and decreasing function. So  $16\Psi_F^3(x) \leq 16\Psi_F^3(0) = -1024 \ln^3 2 = -341.\cdots$ . Thanks to the proof of the point 10 we know that  $(\Psi_F)'_x(x) \leq 0$  and it is a decreasing function. In particular  $0 \geq (\Psi_F)'_x(x) \geq (\Psi_F)'_x(x_{max}) = -0.19\cdots$ . We can conclude that

$$24\Psi_F(x)(\Psi_F)'_x(x) \leq 24(\Psi_F)'_x(x_{max})\Psi_F(x_{max}) = 12.\cdots$$

As for the last summand, it is negative. In fact

$$(\Psi_F)''_{xx}(x) = 2\sum_{k=1}^{\infty} \frac{1}{(2k-2)!} \left(\frac{1}{2^{2k}} \psi^{(2k)}\left(\frac{1}{2}\right) - \psi^{(2k)}(1)\right) x^{2k-2}.$$

Since all the coefficients of the series are negative, we get

$$4(\Psi_F)''_{xx}(x) \leqslant 4(\Psi_F)''_{xx}(0) = -12\zeta(3) = -14.4\cdots$$

where  $\zeta(\cdot)$  denotes the Riemann zeta function.

Summarizing we can conclude that

$$16\Psi_F^3(x) + 24\Psi_F(x)(\Psi_F)'_x(x) + 4(\Psi_F)''_{xx}(x) \leqslant$$

 $\leq 16\Psi_F^3(0) + 24\Psi_F(x_{max})(\Psi_F)'_x(x_{max}) + 4(\Psi_F)''_{xx}(0) = -342.7\cdots.$ 

That assures  $W''_{xx} < 0$ .

12. The explicit expression of  $W''_{xxx}$  is

$$W_{xxx}^{\prime\prime\prime} = \frac{1}{4} F^2(x) \left( 64\Psi_F^4 + 192\Psi_F^2(\Psi_F)_x^{\prime} + 48((\Psi_F)_x^{\prime})^2 + 64\Psi_F(\Psi_F)_{xx}^{\prime\prime} + 8(\Psi_F)_{xxx}^{\prime\prime\prime} \right).$$

We start observing that, since  $\Psi_F$  is a negative decreasing function,

$$64\Psi_F^4(x) \ge 64\Psi_F^4(0) = 64(4\ln 2)^4 = 3782.\cdots$$

Since  $(\Psi_F)'_x(x)$  is a not positive and decreasing function (point 10), then  $192\Psi_F^2(\Psi_F)'_x$  enjoys the same property. In particular

$$192\Psi_F^2(\Psi_F)'_x \ge 192\Psi_F^2(x_{max})(\Psi_F)'_x(x_{max}) = -282.\cdots$$

From the previous observations it follows that  $64\Psi_F(\Psi_F)''_{xx} \ge 0$ ,  $48((\Psi_F)'_x)^2 \ge 0$ and they are increasing functions.

As for the last summand which appears in the expression of  $W''_{xxx}$ , we observe that

$$(\Psi_F)_{xxx}^{\prime\prime\prime} = 2\sum_{k=1}^{\infty} \frac{1}{(2k-1)!} \left( \frac{1}{2^{(2k+2)}} \psi^{(2k+2)}\left(\frac{1}{2}\right) - \psi^{(2k+2)}(1) \right) x^{2k-1}.$$

It is a not positive and decreasing function. We can conclude that

$$8(\Psi_F)'''_{xxx}(x) \ge 8(\Psi_F)'''_{xxx}(x_{max}) = -19.9\cdots$$

Summarizing we can state  $W_{xxx}^{\prime\prime\prime}(x) > 0$ . Furthermore from our observations it follows that

$$W_{xxx}''(x) \leqslant \left(16\Psi_F^4(x_{max}) + 24((\Psi_F)_x')^2(x_{max}) + 16\Psi_F(x_{max})(\Psi_F)_{xx}''(x_{max})\right) < C_W$$

with  $C_W = 1125$ .

- 13. It holds that  $Y'_x(x) = W(x) + x W'_x(x)$ . From the points 9, 10 and 11 we know that W(x) is a negative increasing function and  $W'_x(x)$  is positive and decreasing for  $x \in [0, x_{max}]$ . So we can write  $W(x) \leq W(x_{max}) = -4.1 \cdots$  and  $W'_x(x) \leq W'_x(0) = 64 \ln^2 2 = 30.7 \cdots$ . Then  $Y'_x(x) \leq W(x_{max}) + x_{max}W'_x(0) < 0$ .
- 14. It holds that  $Y''_{xx}(x) = 2W'_x(x) + x W''_{xx}(x)$ . From the points 10, 11 and 12 we know that  $W'_x(x)$  is a positive decreasing function and  $W''_{xx}(x)$  is negative and increasing. So we can write  $W'_x(x) \ge W'_x(x_{max}) = 22.\cdots$  and  $W''_{xx}(x) \ge W''_{xx}(0) = -64 \ln^3 4 6\zeta(3) = -177.\cdots$ . Then  $Y'_x(x) \ge 2W'_x(x_{max}) + x_{max}W''_{xx}(0) > 0$ .
- 15. It holds that  $Y''_{xxx}(x) = 3W''_{xx}(x) + x W''_{xxx}(x)$ . From the points 11 and 12 we know that  $W''_{xx}(x)$  is a negative increasing function and  $W''_{xxx}(x) < C_W$  is positive. Then  $Y''_{xxx}(x) \leq 3W''_{xx}(x_{max}) + x_{max}C_W < 0$ .

**Proposition 7.** For all the admissible values of x, y, z it holds that

$$T_2 \leqslant \frac{1 + Cl^2 x}{l - 1},$$

where  $C = -4 \ln 2$ .

**Proof.** The epression of  $T_2$  is given by (1.18). We rewrite it in the following way

$$T_2 = \frac{1}{l-1} \sqrt{l^2 F^2(x) I^2(z) - (l^2 - 1)L(y)I(z)}.$$

If  $1 + \overline{R}(x, y, z, l)$  is the Mac-Laurin series of the function under the square root then we can write

$$T_2 = \frac{1}{l-1}\sqrt{1+\bar{R}(x,y,z,l)},$$

where  $\overline{R}(x, y, z, l) = l^2(R_{F^2}(c_1x)x(1 + R_{I^2}(c_2z)z) + R_{I^2}(c_2z)z) - (l^2 - 1)R_{LI}(y, z)$ , and  $c_1, c_2 \in (0, 1)$ . Thanks to the points 7,8,9 and 10 of proposition 6, we know that  $R_{LI}(y, z) \ge 0$ ,  $R_{I^2}(x) \le 0$  and that  $R_{F^2}(x)$  is a negative increasing function, so  $R_{F^2}(c_1x) \le R_{F^2}(x)$ . We can conclude that, if we set

$$R(x, z, l) = l^2 R_{F^2}(x) x (1 + R_{I^2}(c_2 z) z),$$

$$\bar{R}(x, y, z, l) \leq l^2 R_{F^2}(c_1 x) x (1 + R_{I^2}(c_2 z) z) \leq R(x, z, l),$$

then

$$T_2 = \frac{1}{l-1}\sqrt{1+\bar{R}(x,z,l)} \leqslant \frac{1}{l-1}\sqrt{1+R(x,z,l)}.$$

We know that

$$\sqrt{1+f(x)} = \sqrt{1+f(0)} + \frac{f'_t(t)}{2\sqrt{1+f(t)}}|_{t=cx}x,$$

where  $c \in (0, 1)$ . If we apply this result to the function f(x) = R(x, z, l), we get

$$T_2 \leqslant \frac{\sqrt{1 + R(x, z, l)}}{l - 1} = \frac{1}{l - 1} \left( \sqrt{1 + R(0, z, l)} + \frac{R'_t(t, z, l)}{2\sqrt{1 + R(t, z, l)}} \Big|_{t = cx} x \right),$$

where  $c \in (0, 1)$ . We observe that R(0, z, l) = 0. Then

$$T_2 \leq \frac{1}{l-1} \left( 1 + \frac{R'_t(t,z,l)}{2\sqrt{1+R(t,z,l)}} |_{t=cx} x \right).$$

The proof will be completed after having proved the following result.

**Proposition 8.** For all the admissible values of t, y, z,

$$\frac{R'_t(t,z,l)}{2\sqrt{1+R(t,z,l)}} \leqslant Cl^2,$$

where  $C = -4 \ln 2$ .

**Proof.** We set  $H(z,l) = l^2(1 + R_{I^2}(c_2z)z) \leq l^2$  and  $Y(t) = R_{F^2}(t)t$ . From the expression of R(t,z,l) = H(z,l)Y(t), it follows that  $R'_t(t,z,l) = H(z,l)Y'_t(t)$ . Furthermore we can write

$$\frac{R'_t(t,z,l)}{2\sqrt{1+R(t,z,l)}} = \frac{H(z,l)Y'_t(t)}{2\sqrt{1+H(z,l)Y(t)}}.$$

We know from proposition 6 that  $Y(t) \leq 0$  and  $Y'_t(t) < 0$ , then  $R'_t(t, z, l) = H(z, l)Y'_t(t) \ge l^2 Y'_t(t)$ , and

$$-\frac{1}{2\sqrt{1+R(t,z,l)}} \ge -\frac{1}{2\sqrt{1+l^2Y(t)}}.$$

We can conclude that

$$-\frac{R'_t(t,z,l)}{2\sqrt{1+R(t,z,l)}} \ge -\frac{l^2 Y'_t(t)}{2\sqrt{1+l^2 Y(t)}}.$$

We shall show that this last function is increasing with respect to the variable t. The derivative with respect to the variable t of this function is

$$D(t,l) = -\frac{l^2}{2} \frac{Y_{tt}''\sqrt{1+l^2Y} - l^2 \frac{(Y_t')^2}{2\sqrt{1+l^2Y}}}{1+l^2Y}$$

We want to show that  $D(t, l) \ge 0$ . We start observing that  $1 + l^2 Y > 0$ . So it is sufficient to prove that the quantity

$$E(t,l) = 2Y_{tt}''(1+l^2Y) - l^2(Y_t')^2$$

is always not positive. It holds that

$$Y'_t(t) = R_{F^2}(t) + t(R_{F^2})'_t(t)$$

and

$$Y_{tt}''(t) = 2(R_{F^2})_t'(t) + t(R_{F^2})_{tt}''(t).$$

Then Y(0) = 0,  $Y'_t(0) = R_{F^2}(0) = 2C$  and  $Y''_{tt}(0) = 2(R_{F^2})'_t(0) = 8\Psi_F(0)^2 = 8C^2$ . Furthermore we observe that  $l \ge 2$ . So

$$E(0,l) = 16C^2 - 4l^2C^2 \le 0$$

and the equality holds if l = 2. The next step is to show that  $E'_t(t, l) \leq 0$ . It is possible to find the following relation

$$E'_t(t,l) = Y'''_{ttt}(1+l^2Y)$$

Observing that  $1 + l^2 Y > 0$  and  $Y_{ttt}^{\prime\prime\prime} \leq 0$  (see the point 15 of proposition 6), we can conclude that D(t, l) > 0. We have showed that

$$-\frac{l^2 Y_t'(t)}{2\sqrt{1+l^2 Y(t)}}$$

is an increasing function. It gets the minimum for t = 0 and its value is  $-Cl^2$ . Then

$$-\frac{R_t'(t,z,l)}{2\sqrt{1+R(t,z,l)}} \geqslant -Cl^2$$

and the proof is completed.

As for the first summand which appears in the expression of  $t_{-}^2$ , that is  $T_1$ , the following result holds.

**Proposition 9.** For all the admissible values of x, z, it holds that

$$T_1 \ge \frac{l}{l-1}(1+Cx)$$

where  $C = -4 \ln 2$ .

**Proof.** We recall that

$$T_1 = \frac{l}{l-1}F(x)I(z) = \frac{l}{l-1}(1+R_{FI}(x,z)).$$

Thanks to the point 6 of proposition 6 we have  $R_{FI}(x,z) \ge Cx$ . Then the result is immediate.

The following result gives the estimate of  $t_{-}^2$ .

**Proposition 10.** For all the admissible values of x, y, z

$$t_{-}^2 \ge 1 - Clx,$$

where  $C = -4 \ln 2$ .

We recall that  $t_{-}^2 = T_1 - T_2$ . Thanks to propositions 7 and 9 we get

$$t_{-}^{2} \ge \frac{l}{l-1}(1+Cx) + \frac{1}{l-1}(-1-Cl^{2}x) = 1 + \left(\frac{Cl}{l-1} - \frac{Cl^{2}}{l-1}\right)x = 1 + \left(\frac{-Cl}{l-1}(l-1)\right)x = 1 - Clx.$$

Now we turn our attention to the function  $t_3$ . We recall that  $s_{max} = \frac{1}{38}$ . **Proposition 11.** For  $s \in [0, s_{max}]$ 

$$t_3^2(s) \leqslant 1 + \frac{7}{2}s.$$

**Proof.** We recall that

$$t_3^2(s) = T(s) = \frac{4}{(1-s)^2} \left(\frac{\Gamma(1+s)}{\Gamma(1-s)}\right)^3 \frac{\Gamma(1-2s)}{\Gamma(1+2s)} \left(\frac{\Gamma(3/2-s/2)}{\Gamma(1/2+s/2)}\right)^2.$$

It holds that

$$T'_{s}(s) = \frac{1}{(1-s)}T(s)B(s),$$

where

$$B(s) = 2 + (1-s)\left(-2\psi(1-2s) - 2\psi(1+2s) + 3\psi(1-s) + 3\psi(1+s) - \psi\left(\frac{3}{2} - \frac{s}{2}\right) - \psi\left(\frac{1}{2} + \frac{s}{2}\right)\right).$$

To complete the proof we need the following result.

**Proposition 12.** *If*  $s \in [0, s_{max}]$  *then* 1 < B(s) < 3*.* 

**Proof.** We observe that for  $s \in [0, s_{max}]$ 

$$0 < \psi\left(\frac{3}{2} - \frac{s}{2}\right) < \psi\left(\frac{3}{2}\right) = 0.036\cdots, \quad \frac{3}{2} < -\psi\left(\frac{1}{2} + \frac{s}{2}\right) < -\psi\left(\frac{1}{2}\right) < 2.$$

We can conclude that

$$1 < -\psi\left(\frac{1}{2} + \frac{s}{2}\right) - \psi\left(\frac{3}{2} - \frac{s}{2}\right) < 2.$$

Furthermore

$$\psi(1-s) + \psi(1+s) = 2\sum_{k\ge 0} \frac{\psi^{(2k)}(1)}{(2k)!} s^{2k},$$

from which it follows that

$$D(s) = -2\psi(1-2s) - 2\psi(1+2s) + 3\psi(1-s) + 3\psi(1+s) = 2\sum_{k\ge 0}\frac{\psi^{(2k)}(1)}{(2k)!}s^{2k}(3-2^{2k+1})$$

If  $k \ge 1$  then  $3 - 2^{2k+1} < 0$  and  $\psi^{(2k)}(1) < 0$  (see formula 6.4.2 of [1]) then

$$2\psi(1) = -2\gamma_{EM} = D(0) \leqslant D(s) \leqslant D(s_{max}) = -1.146\cdots$$

where  $\gamma_{EM} = 0.577 \cdots$  is the Euler-Mascheroni constant. So

$$1 < B(s) \leq 2 + (1 - s)(2 + D(s_{max})) < 4 + D(s_{max}) < 3.$$

Since B(s) > 0 then T(s) is an increasing function and we can deduce that

$$T'(s) = \frac{1}{1-s}T(s)B(s) \le \frac{3}{1-s_{max}}T(s_{max}) < 7/2$$

The Mac-Laurin series of order 1 of T(s) is  $1+T'_s(cs)s$ , where  $c \in (0,1)$ . So it is immediate to conclude that

$$T(s) \leqslant 1 + \frac{7}{2}s.$$

We want to remark that with more work it is possible to show that

$$T(s) \leqslant 1 - Cs$$

The following proposition shows that the eventual solutions of the equation (1.15) are always bigger than  $t_3$ .

**Proposition 13.**  $t_3(s) < t_-(l,g)$  for  $g \ge 1$ .

**Proof.** From our observations, it is sufficient to show that  $t_3^2(s) < t_-^2(l,g)$  holds for  $g \ge 38$ . The propositions 10 and 11 assure that

$$t_{-}^{2} \ge 1 - Clx,$$
  
$$t_{3}^{2}(s) \le 1 + \frac{7}{2}s$$

We recall that x = ls and  $2 \leq l \leq g - 1$ . Then the result is obvious.

#### 1.4 The index and the nullity of the Costa-Hoffman-Meeks surfaces

We start recalling some results described in previous sections. We denoted by  $G_t, t \in (0, +\infty)$ , a deformation of the map G defined by (1.5). Thanks to theorem 5 Nul $(G_t) > 3$  only if t assumes special values. If  $2 \leq g \leq 37$  these values are  $t_1, t_2, t_3$ . If  $g \geq 38$  there are additional values. They are the positive solutions of the equation (1.15). We denoted them by  $t_{\pm}(l,g)$ , where  $2 \leq l \leq g - 1$ , and for definition  $t_+ \geq t_-$ . In previous section we have proved that the inequality  $t_3(s) < t_-(l,g)$  holds. S. Nayatani showed in [29] that  $t_3 > t_2$  for  $g \geq 2$ . We can conclude that no one of the  $t_{\pm}$  can be equal to  $t_2$ . As consequence Nul $(G_{t_2})$  continues to be equal to 4 also for  $g \geq 38$ , because dim  $H(G_{t_2})$  is equal to 1 for all  $g \geq 2$ .

We recall that  $M_g$  denotes the Costa-Hoffman-Meeks surface of genus g. Since the extended Gauss map of  $M_g$  is exactly  $G_{t_2}$ , and taking into account the result of S. Nayatani about the Costa surface (theorem 3) showed in [30] we have proved the following result.

**Theorem 14.** The null space of the Jacobi operator of  $M_g$  has dimension equal to 4 for all  $g \ge 1$ .

Using the definition of non degeneracy given in [34], we can also rephrase this result giving the following statement.

**Corollary 15.** The surface  $M_q$  is non degenerate for all  $g \ge 1$ .

Now we turn our attention to the results relative to the index of the map  $G_t$ . We recall that  $\Sigma_g$  denotes the compactification of  $M_g$ . S. Nayatani proved in [29] the following result.

**Theorem 16.** Let  $G: \Sigma_g \to S^2$  be the holomorphic map defined by (1.5). If  $2 \leq g \leq 37$ , then

$$\operatorname{Ind}(G_t) = \begin{cases} 2g+3 & \text{if } t \leq t_1, t_2 \leq t < t_3, t > t_3, \\ 2g+4 & \text{if } t_1 < t < t_2, \\ 2g+2 & \text{if } t = t_3. \end{cases}$$

For  $t = t_1, t_2, t_3$  we have Nul $(G_t) > 3$ , that is the kernel of  $L_{G_t}$  contains at least one non linear element. The eigenvalue associated to this function is zero. The proof of theorem 16 is based on the analysis of the behaviour of these null eigenvalues under a variation of the value of t. Let's suppose that  $t \neq t_1, t_2, t_3$  but remaining in a neighbourhood of one of these values. For example we choose  $t_1$ . Then the eigenvalue E that before the variation was associated to a non linear element of  $N(G_{t_1})$ , is not more equal to zero. To compute the index, it was necessary to understand which is the sign assumed by E, respectively for  $t > t_1$  and  $t < t_1$ . Similar considerations are applicable to the eigenvalues associated with  $t_2$  and  $t_3$ . See [29] for the details.

If  $g \ge 38$ , we have just proved that the other values for which  $\operatorname{Nul}(G_t) > 3$  are bigger than  $t_3$ . The presence of these additional values  $t_{\pm}$  does not influence the value of  $\operatorname{Ind}(G_t)$ if  $t \le t_3$ . In other terms theorem 16 continues to hold for  $g \ge 38$  if we consider  $0 < t \le t_3$ . Taking into account also the result of S. Nayatani about the Costa surface (g = 1) showed in [30], we can give the following statement

**Theorem 17.** For all  $g \ge 1$  the index of the Gauss map of  $M_g$  is equal to 2g + 3.

#### 1.5 Appendix

This section contains some additional details of the computations made by S. Nayatani.

#### 1.5.1 Divisors and Riemann-Roch theorem

Here we introduce some definitions and concepts of the algebraic geometry. See for example [5].

Let  $\Sigma_g$  be a compact Riemann surface of genus g. A divisor on  $\Sigma_g$  is a finite formal sum of integer multiples of points of  $\Sigma_g$ ,

$$D = \sum_{x \in \Sigma_g} n_x x, \quad n_x \in \mathbb{Z}, n_x = 0 \quad \text{for almost all} \quad x$$

The set of the divisors on  $\Sigma_g$  is denoted by  $\text{Div}(\Sigma_g)$ . The degree of a divisor is the integer  $\text{deg}(D) = \sum n_x$ .

Let  $\mathbb{C}(\Sigma_g)$  be the field of the meromorphic functions on  $\Sigma_g$  and let  $\mathbb{C}(\Sigma_g)^*$  be its multiplicative group of nonzero elements. Every  $f \in \mathbb{C}(\Sigma_g)^*$  has a divisor

$$\operatorname{div}(f) = \sum \nu_x(f) x,$$

where  $\nu_x(f)$  denotes the order of f at x.

Let  $\omega$  be a nonzero meromorphic differential *n*-form on  $\Sigma_g$ . Then  $\omega$  has a local representation  $\omega_x = f_x(z)(dz)^n$  about each point x of  $\Sigma_g$ , where z is the local coordinate about x and  $f_x(z) \in \mathbb{C}(\Sigma_g)^*$ . So we can define in a natural way  $\nu_x(\omega) = \nu_0(f_x)$  and also associate a divisor with a differential form:

$$\operatorname{div}(\omega) = \sum \nu_x(\omega) x.$$

A canonical divisor on  $\Sigma_g$  is a divisor of the form  $\operatorname{div}(\omega)$  where  $\omega$  is a nonzero meromorphic differential form.

Let  $D \in \operatorname{div}(\Sigma_g)$ . We denote by  $H^{0,n}(D)$  the vector space of the meromorphic differential *n*-forms  $\omega$  such that

$$\operatorname{div}(\omega) + D \ge 0.$$

In other terms, if  $D = \operatorname{div}(\eta)$ , with  $\eta$  differential form with local representation  $\eta_x = g_x(z)(dz)^n$ , then the elements of  $H^{0,n}(D)$  are the differential forms  $\omega$  having a local representation  $\omega_x = f_x(z)(dz)^n$  with  $f_x \in \mathbb{C}(\Sigma_g)$  vanishing to high enough order to make the product  $f \cdot g$  holomorphic. We set dim  $H^{0,n}(D) = \ell(D)$ .

We are ready to state the following result.

**Theorem 18** (Riemann-Roch). Let  $\Sigma_g$  be a compact Riemann surface of genus g. Let  $k_{\Sigma_g}$  be a canonical divisor on  $\Sigma$ . Then for any divisor  $D \in \text{Div}(\Sigma_g)$ ,

$$\ell(D) = \deg(D) - g + 1 + \ell(k_{\Sigma_q} - D).$$

The next result gives information about the canonical divisor and a simpler version of Riemann-Roch theorem for divisors of large enough order.

**Corollary 19.** Let  $\Sigma_q, g, D, k_{\Sigma_q}$  as above.

- $\deg(k_{\Sigma_q}) = 2g 2,$
- If  $\deg(D) > 2g 2$  then  $\ell(k_{\Sigma_g} D) = 0$ . Equivalently  $\ell(D) = \deg(D) g + 1$ .

# 1.5.2 The determination of a basis of differential forms with null residue at the ramification points

The ramification points (or branch points) of  $\gamma(w) = w$  are the zeroes of

$$\frac{dw}{dz} = (g+2)\frac{z^{g-1}(z^2 - A^2)}{(z^g(z^2 - 1))^{\frac{g}{g+1}}},$$

with  $A = \frac{g}{g+2}$ , where g denotes the genus. They are given by  $Q_0 = (0,0)$ ,  $P_{\infty} = (\infty, \infty)$ ,  $P_m = (A, B_m)$  and  $S_m = (-A, C_m)$  for  $m = 0, \ldots, g$ , where  $B_m, C_m$  denote, respectively, the *m*-th complex value of  $\sqrt[g+1]{A^g(A^2-1)}$  and  $\sqrt[g+1]{(-A)^g(A^2-1)}$ . We have set  $P_{\pm} = (\pm 1, 0)$ . We recall that

$$\hat{H}(G) = \left\{ \sigma \in H^0(2k_{\Sigma_g} + R(G)) \,|\, Res_{r_i} \frac{\sigma}{dw} = 0, i = 1, \dots, \mu \right\},\tag{1.21}$$

where  $k_{\Sigma_g}$  is a canonical divisor of  $\Sigma_g$  and  $R(G) = \sum_{i=1}^{\mu} r_i$  where  $r_i$  are the branch points of  $\gamma(w) = w$ . In our case it is given by  $R(G) = Q_0 + P_\infty + \sum_{m=0}^{g} (P_m + S_m)$ . Furthermore it holds  $\hat{H}(G) = \hat{H}(G_t)$ .

As for the canonical divisor  $k_{\Sigma_g}$ , we consider  $k_{\Sigma_g} = (g-1)P_+ + (g-1)P_-$ . We observe that  $\deg(k_{\Sigma_g}) = 2g - 2$  like stated by corollary 19.

To study the space  $\hat{H}(G_t)$  we need understand which are the members of the space  $H^0(2k_{\Sigma_g}+R(G))$ . Taking into account the definitions of  $k_{\Sigma_g}$  and R(G), then  $2k_{\Sigma_g}+R(G) = 2(g-1)P_+ + 2(g-1)P_- + Q_0 + P_\infty + \sum_{m=0}^g P_m + \sum_{m=0}^g S_m$ . We deduce that the quadratic differentials  $\sigma$  that are in  $H^0(2k_{\Sigma_g}+R(G))$  can have two possible structures:

$$z^k w^j \left(\frac{dz}{w}\right)^2,\tag{1.22}$$

$$z^{k}w^{j}\frac{1}{z\pm A}\left(\frac{dz}{w}\right)^{2}.$$
(1.23)

In fact from the definition of  $H^0$ , it follows that the quadratic differentials to consider can have a pole of order 0 (differentials of the type (1.22)) or of order 1 (differentials of the type (1.23)) at  $P_m$  and  $S_m$  for  $k = 0, \ldots, g$ . Furthermore we observe that the quadratic differential  $\left(\frac{dz}{w}\right)^2$  has

- a zero of order 1 at  $Q_0$
- a zero of order 1 at  $P_{\infty}$
- a zero of order 2(g-1) at  $P_+$  and  $P_-$ .

The quadratic differential  $\frac{1}{z \pm A} \left(\frac{dz}{w}\right)^2$  has

- a zero of order 1 at  $Q_0$
- a zero of order 2(g-1) at  $P_+$  and  $P_-$
- a zero of order g + 2 at  $P_{\infty}$

• a pole of order 1 at  $P_m$  if we consider sign "-" or at  $S_m$  if we consider the sign "+".

We will determine separately the interesting differential forms of type (1.22) and (1.23). To find the differential forms of type (1.23) it is convenient to introduce an auxiliar divisor.

$$D = Q_0 + (g+2)P_{\infty} + 2(g-1)P_+ + 2(g-1)P_-.$$

We observe that the elements of the vector space  $H^0(D)$  after the multiplication by the factor  $z \pm A$  are members of  $H^0(2k_{\Sigma_g} + R(G))$ . It is necessary to remark that to obtain a basis of  $H^0(2k_{\Sigma_g} + R(G))$ , we will not take into account the differentials of  $H^0(2k_{\Sigma_g} + R(G))$  that can be constructed from an element of  $H^0(D)$  as described above. Otherwise the number of the founded differential forms exceeds the dimension of  $H^0(2k_{\Sigma_g} + R(G))$ , that we can compute as follows. We observe that  $\deg(2k_{\Sigma_g} + R(G)) = 6g$ . Then thanks to corollary of Riemann-Roch theorem 19 we conclude that  $\dim H^0(2k_{\Sigma_g} + R(G)) = 5g + 1$ . So the basis we are looking for counts 5g + 1 elements. From the observations made above we can deduce that between the forms of type (1.22), we will consider the ones which satisfy the following conditions.

$$\begin{aligned} k(g+1) + jg &\geq -1, \\ j &\geq -2(g-1), \\ -k(g+1) - j(g+2) &\geq -1. \end{aligned}$$

These relations assure that the selected differentials of type  $z^k w^j \left(\frac{dz}{w}\right)^2$ , are holomorphic, respectively, at the points  $Q_0$ ,  $P_{\pm}$  and  $P_{\infty}$ . These differentials can be classified in three families. Each family is characterized by particular values of l and k. That is

- 1.  $j = -g + 1, \dots, 0, 1$  and k = -j,
- 2.  $j = 2 2g, \ldots, -g$  and k = -j,
- 3.  $j = 2 2g, \ldots, -g$  and k = -j 1.

As for the forms of type (1.23) we shall consider only the ones which satisfy

$$\begin{split} k(g+1)+jg \geqslant -1,\\ j \geqslant -2(g-1),\\ -k(g+1)-j(g+2) \geqslant -(g+2). \end{split}$$

These relations assure that the selected differentials of type  $z^k w^j \frac{1}{z \pm A} \left(\frac{dz}{w}\right)^2$ , are holomorphic, respectively, at the points  $Q_0$ ,  $P_{\pm}$  and  $P_{\infty}$ . We obtain that  $j = -g + 1, \ldots, 0, 1$  and k = -j + 1.

Since we are looking for a basis of a vector space we can replace each couple of differentials  $\frac{f}{z-A}\left(\frac{dz}{w}\right)^2$ ,  $\frac{f}{z+A}\left(\frac{dz}{w}\right)^2$  by an appropriate linear combination. We observe that

$$\frac{1}{z-A} \pm \frac{1}{z+A} = \begin{cases} \eta_1 = \frac{z}{z^2 - A^2} \\ \eta_2 = \frac{1}{z^2 - A^2} \end{cases}$$

So in the following we will work with the forms  $f\eta_1\left(\frac{dz}{w}\right)^2$  and  $f\eta_2\left(\frac{dz}{w}\right)^2$ , where  $f = z^k w^j$  as described above.

The 5g + 1 quadratic differentials we have found forms a basis of  $H(G_t)$ . The last step is to divide each elements of this basis by dw. After simple algebraic manipulations, we obtain the following 5g + 1 differential 1-forms:

$$\frac{w^{k}}{z^{k-1}} \frac{dz}{(z^{2}-A^{2})^{2}} \quad \text{for} \quad k = -1, 0, \dots, g-1, 
\frac{w^{k}}{z^{k}} \frac{dz}{(z^{2}-A^{2})^{2}} \quad \text{for} \quad k = -1, 0, \dots, g-1, 
\frac{w^{k}}{z^{k+1}} \frac{dz}{(z^{2}-A^{2})} \quad \text{for} \quad k = -1, 0, \dots, g-1, 
\frac{z^{k}}{w^{k+1}} \frac{dz}{(z^{2}-A^{2})} \quad \text{for} \quad k = 1, \dots, g-1, 
\frac{z^{k-1}}{w^{k+1}} \frac{dz}{(z^{2}-A^{2})} \quad \text{for} \quad k = 1, \dots, g-1.$$
(1.24)

Now it is necessary to select the 1-forms having residue equal to zero at the points  $Q_0$ ,  $P_m$ and  $S_m$  with  $m = 0, \ldots, g$ . Thanks to the properties of symmetry of the surface it is sufficient to verify the null residue condition at the points  $Q_0$ ,  $P_1 = (A, e^{\frac{2\pi i}{g+1}} \sqrt[g]{4g}(A^2 - 1))$ . In fact from the coordinates of the points  $P_m$  and  $S_m$ , we can deduce that for each  $Q \in \{P_m, S_m, k = 0, \ldots, g\}$  there exists  $n \in \{0, \ldots, 2g + 1\}$  such that  $Q = \lambda^n(P_1)$ , where  $\lambda$  is the conformal diffeomorphism described in lemma 4. So we can state that the residue of an arbitrary form  $\omega$  at the point Q is related to the residue at  $P_1$  by

$$\operatorname{Res}_Q \omega = \operatorname{Res}_{P_1}(\lambda^{n-1})^* \omega.$$

Applying this result to the differential forms of the list (1.24) and using the the definition (1.4) of  $\lambda$ , it is easy to obtain that  $\operatorname{Res}_{Q}\omega$  is equal to  $\operatorname{Res}_{P_{1}}\omega$  times a power of  $\pm\rho$ . So if  $\operatorname{Res}_{P_{1}}\omega = 0$  then  $\operatorname{Res}_{Q}\omega = 0$ .

Thanks to algebraic manipulations inspired by the simpler cases where g = 2, 3, it is possible to find 3g linear independent differential forms satisfying the null residue condition. They constitute the wanted basis.

$$\omega_k^{(1)} = \frac{z^{k-1}}{w^k} \frac{dz}{w} \quad \text{for} \quad k = 1, \dots, g-1,$$
$$\omega_k^{(2)} = \frac{z^{k-1}((k-2)z^2 - kA^2)}{w^k(z^2 - A^2)^2} dz \quad \text{for} \quad k = 0, \dots, g,$$

$$\omega_k^{(3)} = \frac{z^{k-1}((k-2)z^2 - (k)A^2)}{w^{k+1}(z^2 - A^2)^2} dz \quad \text{for} \quad k = 0, \dots, g-1$$

# 1.5.3 The equations equivalent to the condition of existence of a branched minimal surface.

Let  $\omega_1$  and  $\omega_2$  two meromorphic differential forms on  $\Sigma_g$ . We write  $\omega_1 \sim \omega_2$  if there exists a meromorphic function f on  $\Sigma_g$  such that  $\omega_2 = \omega_1 + df$ . It is possible to prove that:

$$\omega_k^{(2)} \sim -\frac{k(g+2)}{2(g+1)} \frac{z^{k-1}}{w^k} dz \quad \text{for} \quad k = 0, \dots, g,$$
  
$$\omega_k^{(3)} \sim -\frac{(g+2)(g+k+2)}{2(g+1)} \frac{z^{k-1}}{w^{k+1}} dz \quad \text{for} \quad k = 0, \dots, g-1.$$

Using these relations we get:

$$\begin{split} \int_{\tilde{\beta}} \omega_k^{(1)} &= -2i \sin \frac{(k+1)\pi}{g+1} K_k, \quad \int_{\tilde{\beta}} \omega_k^{(2)} &= -\frac{(g+2)k}{2(g+1)} 2i \sin \frac{k\pi}{g+1} I_k, \\ &\int_{\tilde{\beta}} \omega_k^{(3)} &= \frac{(g+2)(g+2-k)}{2(g+1)} 2i \sin \frac{(k+1)\pi}{g+1} K_k, \\ \int_{\tilde{\beta}} \gamma \omega_k^{(1)} &= 2i \sin \frac{k\pi}{g+1} I_k, \quad \int_{\tilde{\beta}} \gamma \omega_k^{(2)} &= \frac{(g+2)(g+2-k)}{2(g+1)} 2i \sin \frac{(k-1)\pi}{g+1} J_k, \\ &\int_{\tilde{\beta}} \gamma \omega_k^{(3)} &= -\frac{(g+2)k}{2(g+1)} 2i \sin \frac{k\pi}{g+1} I_k, \\ \int_{\tilde{\beta}} \gamma^2 \omega_k^{(1)} &= 2i \sin \frac{(k-1)\pi}{g+1} J_k, \quad \int_{\tilde{\beta}} \gamma^2 \omega_k^{(2)} &= \frac{(g+2)(2g+4-k)}{2(g+1)} 2i \sin \frac{(k-2)\pi}{g+1} L_k, \\ &\int_{\tilde{\beta}} \gamma^2 \omega_k^{(3)} &= \frac{(g+2)(g+2-k)}{2(g+1)} 2i \sin \frac{(k-1)\pi}{g+1} J_k. \end{split}$$

We recall that we must impose that  $\omega = \sum_{0}^{g-1} c_k^{(1)} \omega_k^{(1)} + \sum_{0}^{g} c_k^{(2)} \omega_k^{(2)} + \sum_{0}^{g-1} c_k^{(3)} \omega_k^{(3)}$ , where  $c_k^{(i)} \in \mathbb{C}$ , satisfies

$$\int_{\alpha} \omega = t^2 \int_{\alpha} \gamma^2(w) \omega, \quad Re \int_{\alpha} \gamma(w) \omega = 0$$

for  $\alpha = \lambda^l \circ \tilde{\beta}$  for  $l = 0, \dots, 2g-1$ . Now it is convient to introduce some additional notation. Let

$$\mathcal{L} = \begin{bmatrix} \mathcal{R}_{\theta} & 0\\ 0 & 1 \end{bmatrix}$$
(1.25)

where  $\mathcal{R}_{\theta}$  is the rotation in the plane by  $\theta = g\pi/(g+1)$ . If we denote  $\Phi(\omega) = (1 - \gamma^2, i(1 + \gamma^2), 2\gamma)\omega$ , then it is possible to prove

$$\int_{\lambda^l \circ \tilde{\beta}} \Phi(\omega) = \int_{\tilde{\beta}} \lambda^* \Phi(\omega).$$

Since we want to apply this last relation to the differential form  $\omega$ , it is convenient to remark that:

$$\lambda^* \Phi(\omega_k^{(1)}) = (-1)^k \rho^{-k} \mathcal{L} \Phi(\omega_k^{(1)}),$$
  
$$\lambda^* \Phi(\omega_k^{(2)}) = (-1)^k \rho^{-k+1} \mathcal{L} \Phi(\omega_k^{(2)}),$$
  
$$\lambda^* \Phi(\omega_k^{(3)}) = (-1)^k \rho^{-k} \mathcal{L} \Phi(\omega_k^{(3)}),$$

where  $\rho = e^{i\frac{g\pi}{g+1}}$ . Then the equations

Re 
$$\int_{\lambda^l \circ \tilde{\beta}} (1 - t^2 \gamma^2, i(1 + t^2 \gamma^2)) \omega = 0$$
, for  $l = 0, \dots, 2g - 1$ ,

are equivalent to:

$$\begin{split} ℑ\left[\sum_{k=0}^{g-1}\{(-1)^k\rho^{-k}\}^lf_k+\sum_{k=1}^g\{(-1)^k\rho^{-(k-1)}\}^lp_k\right]=\\ &t^2Im\left[\sum_{k=0,k\neq 1}^{g-1}\{(-1)^k\rho^{-k}\}^lh_k+\sum_{k=0,k\neq 2}^g\{(-1)^k\rho^{-(k-1)}\}^lq_k\right]\\ ℜ\left[\sum_{k=0}^{g-1}\{(-1)^k\rho^{-k}\}^lf_k+\sum_{k=1}^g\{(-1)^k\rho^{-(k-1)}\}^lp_k\right]=\\ &-t^2Re\left[\sum_{k=0,k\neq 1}^{g-1}\{(-1)^k\rho^{-k}\}^lh_k+\sum_{k=0,k\neq 2}^g\{(-1)^k\rho^{-(k-1)}\}^lq_k\right], \end{split}$$

 $l = 0, \ldots, 2g-1$ . These last equations can be arranged as in the systems (1.10) and (1.11). The equations

Re 
$$\int_{\lambda^l \circ \tilde{\beta}} 2t\gamma \omega = 0$$
, for  $l = 0, \dots, 2g - 1$ ,

are equivalent to:

$$Im\left[\sum_{k=1}^{g-1} \{(-1)^k \rho^{-k}\}^l d_k + \sum_{k=0, k\neq 1}^g \{(-1)^k \rho^{-(k-1)}\}^l e_k\right] = 0,$$

 $l = 0, \ldots, 2g-1$ . These last equations can be arranged as in the systems (1.12) and (1.13).

## Chapter 2

## About a family of deformations of the Costa-Hoffman-Meeks surfaces

#### 2.1 Introduction

C. Costa in [3, 4] described a genus one minimal surface with two ends asymptotic to the two ends of a catenoid and a middle end asymptotic to a plane. D. Hoffman and W.H. Meeks in [14], [15] and [16] proved the global embeddedness for the Costa surface, and generalized it for higher genus. We will denote the Costa-Hoffman-Meeks surface of genus  $k \ge 1$  by  $M_k$ . For each  $k \ge 1$  is a properly embedded minimal surface and has three ends of finite total curvature.

J. Pérez and A. Ros in [34] studied the space  $\mathcal{M}$  of minimal surfaces of finite total curvature, genus k and r ends, properly immersed in  $\mathbb{R}^3$  and with embedded horizontal ends. Given  $M \in \mathcal{M}$ , the infinitesimal deformations of M are generated by the elements of the J(M), the space of the Jacobi functions u on M, that is functions such that Lu = 0, where L denotes the Jacobi operator of M, which have logarithmic growth at the ends. They showed that dim  $J(M) \ge r + 3$ . They denoted by  $\mathcal{M}^* = \{M \in \mathcal{M} : \dim J(M) = r + 3\}$ the subspace of non degenerate surfaces and founded that it is an open subset of  $\mathcal{M}$  and a (r + 3)-dimensional real-analytic manifold.

The dimension of the space J(M) just introduced is known for  $M = M_k$  for  $k \ge 1$ . In fact thanks to the works [29] and [30] of S. Nayatani, dim  $J(M_k) = 6$ , since r = 3, but only for  $1 \le k \le 37$ . Now this result has been extended also for  $k \ge 38$  (see chapter 1). The elements of  $J(M_k)$  are the Jacobi fields associated to the horizontal translations, the rotation about the vertical direction and three functions (one for each end) whose form in a neighbourhood of an end is  $a \log |w|$ , being a the logarithmic growth. Thus, the one parameter family of deformations of these surfaces described by D. Hoffman and H.
Karcher in [13], contains all the embedded surfaces nearby  $M_k$  with a symmetry group generated by k vertical planes, up to dilations preserving the vertical direction.

In this work, following [34], we show the existence of a bigger family of immersed minimal deformations of  $M_k$  for  $k \ge 1$  having three embedded ends. These surfaces do not enjoy any property of symmetry. In fact we admit the possibility to rotate, translate and dilate any of the three ends of the surface and, in addition, to bend the two catenoidal type ends and to change the type of the middle end from a planar type end into a catenoidal type end (we recall that the planar end can be thought to be as a catenoidal type end with null vertical flux). Admitting a bigger number of deformations of  $M_k$ , one has, as consequence, the rise of dim  $J(M_k)$ . Actually it rises from the value 6 of [34] to 8.

To be more precise we will prove the following result.

**Theorem 20.** For for each possible choice of the limit values of the normal vectors of the three ends, there is, up to isometries, a 1-dimensional real analytic family of smooth minimal deformations of  $M_k$ , for  $k \ge 1$ , letting the middle planar end horizontal.

Our result is a consequence of the moduli space theory and of the implicit function theorem. We do not treat the case where also the middle planar end is not horizontal because it can be reconduced to the previous one by an isometry.

The family of surfaces described in the statement of the theorem here, contains the 1parameter family of deformations of  $M_k$ , for  $1 \le k \le 37$ , obtained by L. Hauswirth and F. Pacard in [11] bending the top and the bottom end and letting horizontal the middle planar end. All the surfaces of this family are not embedded and are symmetric with respect to the vertical plane  $x_2 = 0$  that in particular contains the axis of the catenoidal type ends (it is assumed to be the same for the two ends). The parameter is the angle between this axis and the vertical direction. This family is used in the same work to construct some new examples of minimal surfaces by a gluing technique.

One important property of the Costa-Hoffman-Meeks surface, is the non degeneracy. In section 2.3.2 we will prove that  $M_k$  is non degenerate for all  $k \ge 1$  with respect to the definition given in [34].

In [11] F. Pacard and L. Hauswirth studied the mapping properties of the Jacobi operator of  $M_k$  acting on the space of the  $C_{\delta}^{2,\alpha}$  functions defined on  $M_k$  and that are invariant under the action of the symmetry with respect to the plane  $x_2 = 0$ . In particular if  $f \in C_{\delta}^{2,\alpha}(M_k)$ , then  $f = O(e^{\delta s})$  on the catenoidal type ends. The mapping properties of the Jacobi operator (denoted by  $L_{\delta}$ ) acting on functions of  $C_{\delta}^{2,\alpha}(M_k)$  depend on the choice of  $\delta$ . The authors give another definition of non degeneracy. They define the surface  $M_k$  to be non degenerate if the operator  $L_{\delta}$  is injective for all  $\delta < -1$ .

Thanks to the works [29] and [30] of S. Nayatani and result of chapter 1, the space  $K \subset J(M_k)$  of the bounded Jacobi functions, is known to be generated by the functions  $\langle N, e_1 \rangle$ ,  $\langle N, e_2 \rangle$  and  $\langle N, e_3 \rangle$ ,  $\langle N, e_3 \times p \rangle$ , where N denotes the normal vector field about  $M_k$  and  $(e_1, e_2, e_3)$  is the canonical basis of  $\mathbb{R}^3$ . These functions are associated to 4 isometries of the ambient space: the three translations and the rotation about the  $e_3$ -axis. In [11] the authors remark that the Jacobi function  $\langle N, e_3 \times p \rangle$  associated to the rotation about the  $e_3$ -axis and the translation along the  $e_2$ -axis do not respect the mirror symmetry described above, that is they are not invariant with respect to the action of the map  $(x_1, x_2, x_3) \rightarrow (x_1, -x_2, x_3)$ . So they did not taken into account them and could conclude that  $M_k$  are non degenerate, in the sense of their definition.

The surfaces of the family described in our work do not enjoy any property of symmetry, since we admit to bend the catenoidal type ends in arbitrary directions. Then the Jacobi functions described above must be taken into account. Since the Jacobi function  $\langle N, e_3 \times p \rangle$  belongs to the space  $C^{2,\alpha}_{\delta}(M_k)$  for  $\delta = -k - 1 \leq -2$ , the property of non degeneracy does not hold any more. Actually the operator  $L_{\delta}$  acting on  $C^{2,\alpha}_{\delta}(M_k)$  is no more injective for all  $\delta < -1$ . As consequence, we can state that for all  $k \geq 1$  the Costa-Hoffman-Meeks surface  $M_k$  is degenerate in the sense of the definition given in [11].

### 2.2 Preliminaries and notation

We denote by  $X : M_k \to \mathbb{R}^3$  the conformal minimal immersion of the Costa-Hoffman-Meeks surface  $M_k$  in  $\mathbb{R}^3$ . If g and  $\eta$  are the Weierstrass data of  $M_k$ , we can write:

$$X(z) = \left(\frac{1}{2}\overline{\int g^{-1}\eta} - \frac{1}{2}\int g\eta, \operatorname{Re}\int\eta\right) \in \mathbb{C} \times \mathbb{R} = \mathbb{R}^3.$$

The meromorphic function g is the stereographic projection from the north pole of the Gauss map  $N: M_k \to \mathbb{S}^2$ . The total curvature is finite and  $M_k$  is conformally diffeomorphic to  $\overline{M_k} \setminus \{p_t, p_b, p_m\}$ , being  $\overline{M_k}$  a compact surface and  $p_i$  three points. The Weierstrass data extend in a meromorphic way at each puncture  $p_i$ . In particular the Gauss map of X(z) is well defined at  $p_i$ . The points  $p_i$  are identified with the ends and a neighbourhood of a puncture will parametrize the corresponding end. In the following we will refer to various quantities related to the three ends of the surface using the index t for the top end, the index b for the bottom end and the index m for the planar end. The Gauss map N takes the limit values (0, 0, 1) at the ends  $p_t$  and  $p_b$  and (0, 0, -1) at the end  $p_m$ .

We parametrize the ends  $p_i$  in the graph coordinate  $x = x_1 + ix_2$  on  $D_i^*(\epsilon_i) = \{x \in \mathbb{C}; 0 < |x| \leq \epsilon_i\}$  by the immersions

$$X_i(x) = \left(\frac{1}{x}, -\tilde{a}_i \ln |x| + h_i(x)\right) \in \mathbb{C} \times \mathbb{R} = \mathbb{R}^3$$

for i = t, b, where  $h_i$  is a smooth real valued function on  $D_i^*(\epsilon_i)$ . The quantities  $\tilde{a}_i$  and  $h_i(0)$  are called the logarithmic growth and the height of the end. We can observe that, for the null flux condition,  $\tilde{a}_t = -\tilde{a}_b$ . The Gauss map for an end with logarithmic growth a is given by (see p. 182, [34])

$$N(x) = Q^{-\frac{1}{2}} \left( -a\bar{x} + \bar{x}^2 \nabla_0 h, 1 \right), \qquad (2.1)$$

where  $Q = 1 + |x|^2 (a^2 + |x|^2 |\nabla_0 h|^2 - 2a \langle x, \nabla_0 h \rangle)$  and  $\bar{x}^2 \nabla_0 h$  means the product of the complex number  $\bar{x}^2$  with the gradient  $\nabla_0 h$  respect to the flat metric  $ds_0^2$  of the x-plane.

As for the planar end  $p_m$ , we will use the following parametrization

$$X_m(x) = \left(\frac{1}{x}, h_m(x)\right)$$
 on  $D_m^*(\epsilon_m)$ .

So its logarithmic growth is zero.

## 2.3 The deformation of the surface and its Jacobi operator

In this section we describe how we deform the surface  $M_k$  and we give the expressions of the immersions in  $\mathbb{R}^3$  of the three ends of the deformed surface. In subsection 2.3.1 we introduce the Jacobi operator of  $M_k$  and we study its kernel and its range.

We deform the surface  $M_k$  in the following way. Using a smooth cut-off function we glue  $X: M_k \setminus (D_t^* \cup D_b^* \cup D_m^*) \to \mathbb{R}^3$  with the parametrizations of the three ends with a different value of the logarithmic growths (that we denote with  $a_t, a_b, a_m$ ). Furthermore we rotate the ends  $p_t$  and  $p_b$ , that is we change the directions of their axes of revolution. We denote with  $F(\theta_{1,i}, \theta_{2,i})$  the frame defined by the following unit vectors:

$$e_1(\theta_{1,i}, \theta_{2,i}) = \cos \theta_{1,i} e_1 + \sin \theta_{1,i} \sin \theta_{2,i} e_2 + \sin \theta_{1,i} \cos \theta_{2,i} e_3, \tag{2.2}$$

$$e_2(\theta_{1,i}, \theta_{2,i}) = \cos \theta_{2,i} e_2 - \sin \theta_{2,i} e_3,$$

$$e_{3}(\theta_{1,i},\theta_{2,i}) = -\sin\theta_{1,i}e_{1} + \cos\theta_{1,i}\sin\theta_{2,i}e_{2} + \cos\theta_{1,i}\cos\theta_{2,i}e_{3},$$

where  $(e_1, e_2, e_3)$  denotes the canonical base of  $\mathbb{R}^3$ .

The immersions of the rotated catenoidal type ends on  $D_i^*(\epsilon_i)$  are given by

$$X_{i,\theta_{1,i},\theta_{2,i}}(x) = \frac{x_1}{|x|^2} e_1(\theta_{1,i},\theta_{2,i}) - \frac{x_2}{|x|^2} e_2(\theta_{1,i},\theta_{2,i}) + (-a_i \ln|x| + h_i(x)) e_3(\theta_{1,i},\theta_{2,i}),$$

for i = t, b. As for the planar end, we consider on  $D_m^*(\epsilon_m)$  in the canonical frame  $(e_1, e_2, e_3)$  the immersion

$$X_{m,0,0}(x) = \left(\frac{1}{x}, -a_m \ln |x| + h_m(x)\right).$$

We define  $y = (a_t, a_b, a_m, \theta_{1,t}, \theta_{1,b}, \theta_{2,t}, \theta_{2,b})$ . Thanks to the deformation we obtain a family of immersions that we denote with  $X_y : M_k \to \mathbb{R}^3$ , not necessarily minimal and depending smoothly on y. Now let  $\widetilde{N}(y) \in \mathcal{C}^{\infty}(\overline{M_k}, \mathbb{R}^3)$  be a vector field such that  $\langle \widetilde{N}(y), N \rangle = 1$  on  $\overline{M_k} \setminus (D_t^* \cup D_b^*)$  and

$$\widetilde{N}(y) = \frac{e_3(\theta_{1,i}, \theta_{2,i})}{\langle N, e_3(\theta_{1,i}, \theta_{2,i}) \rangle}$$

on  $D_i^*$  for i = t, b. We remark that we do not modify the normal vector field on  $D_m^*$  because we keep the the middle planar end horizontal. Let  $\mathcal{A}$  be a neighbourhood of  $(\tilde{a}_t, \tilde{a}_b, 0)$ (the logarithmic growths of the ends of  $M_k$ ),  $\mathcal{U}$  a neighbourhood of zero in  $\mathcal{C}^{2,\alpha}(\overline{M_k})$ . For  $y \in \mathcal{A} \times [-\epsilon, \epsilon]^4$  and a function  $u \in \mathcal{U}$ , we consider the family of immersions

$$X_y + u\widetilde{N}(y) : M_k \longrightarrow \mathbb{R}^3.$$
(2.3)

Such a family depends analytically on (y, u).

Let  $\lambda \in \mathcal{C}^{\infty}(M_k)$  be a positive function which in terms of the graph coordinate is defined by

$$\lambda(x) = \begin{cases} \frac{1}{|x|^4} & \text{on } D_t^*(\epsilon_t), \ D_b^*(\epsilon_b), \ D_m^*(\epsilon_m), \\ 1 & \text{on } M_k \setminus (D_t^*(2\epsilon_t) \cup D_b^*(2\epsilon_b) \cup D_m^*(2\epsilon_m)). \end{cases}$$
(2.4)

We know from [34] that  $d\bar{s}^2 = (1/\lambda) ds_0^2$  is a Riemannian metric on  $\overline{M_k}$ . We denote the associated area mesure by  $d\bar{A}$ . If H(y, u) is the mean curvature function of the immersion  $X_y + u\tilde{N}(y)$ , we consider the operator  $\overline{H}(y, u) = \lambda H(y, u)$ . Since at the ends  $p_t$  and  $p_b$ , the rotation does not change the value of the mean curvature, we can apply lemma 6.4 proved in [34] at each end to conclude that the operator

$$\overline{H}: \mathcal{A} \times \mathcal{U} \to \mathcal{C}^{0,\alpha}(\overline{M_k})$$

is real-analytic.

#### 2.3.1 The Jacobi operator

We define  $\tilde{y} = (\tilde{a}_t, \tilde{a}_b, 0, 0, 0, 0, 0)$  and  $\dot{y} = (\dot{a}_t, \dot{a}_b, \dot{a}_m, \dot{\theta}_{1,t}, \dot{\theta}_{2,t}, \dot{\theta}_{1,b}, \dot{\theta}_{2,b})$ . and consider a smooth curve

$$\gamma(t) = (a_t(t), a_b(t), a_m(t), \theta_{1,t}(t), \theta_{2,t}(t), \theta_{1,b}(t), \theta_{2,b}(t), u(t)),$$
(2.5)

with  $|t| < \epsilon$ , passing by  $(\tilde{y}, 0)$  with acceleration  $\gamma'(0) = (\dot{y}, u'(0))$ .

To introduce the Jacobi operator we need consider the continuous family of smooth deformations  $X_{\gamma(t)} + u(t)\tilde{N}(\gamma(t)) : M_k \to \mathbb{R}^3$  of the Costa-Hoffman-Meeks surface. Denote by H(t) the mean curvature of  $X_{\gamma(t)} + u(t)\tilde{N}(\gamma(t))$ .

If 
$$w = \langle \frac{d}{dt}|_{t=0} \left( X_{\gamma(t)} + u(t)\widetilde{N}(\gamma(t)) \right), N \rangle$$
 is the variation field, we have:  

$$\frac{d}{dt}|_{t=0} H(t) = \frac{1}{2}Lw = \frac{1}{2} \left( \Delta_{ds_0^2} w + |A|^2_{ds_0^2} w \right)$$
(2.6)

where  $|A|_{ds_0^2}$  denotes the norm of the second fundamental form computed with respect to the metric  $ds_0^2$ . L is the Jacobi operator of  $M_k$ . It can be "compactified" to obtain the operator  $\overline{L} = \Delta_{d\overline{s}^2} + |A|_{d\overline{s}^2}^2 = \lambda L$  on  $\overline{M_k}$ . The function  $\lambda$  is defined by (2.4). It is related to the differential of  $\overline{H}(t) = \lambda H(t)$  by a relation similar to (2.6).

In the following we will give the expression of a Jacobi function. To express it we need introduce additional notation. Let  $f_1, f_2, f_3$  be the functions defined by:

$$f_1(x,i) = \frac{x_1}{|x|^2} \langle N, e_3 \rangle - (-\tilde{a}_i \ln |x| + h_i(x)) \langle N, e_1 \rangle,$$
  
$$f_2(x,i) = \frac{x_2}{|x|^2} \langle N, e_3 \rangle + (-\tilde{a}_i \ln |x| + h_i(x)) \langle N, e_2 \rangle,$$
  
$$f_3(x, \dot{a}_i) = -\dot{a}_i \ln |x| \langle N, e_3 \rangle$$

in  $D_i^*(\epsilon_i)$  for i = t, b, m and  $f_n(x, i) = 0, n = 1, 2, 3$ , in  $M_k \setminus D_i^*(2\epsilon_i)$ . We recall that  $\tilde{a}_t, \tilde{a}_b$  are the logarithmic growths of the top and of the bottom end of  $M_k$ , and being the middle planar end horizontal, we assume  $\tilde{a}_m = 0$ .

**Proposition 21.** The Jacobi functions about  $M_k$  have, in  $D_i^*(\epsilon_i)$ , the following expression

$$\theta_{1,i}f_1(x,i) + \theta_{2,i}f_2(x,i) + f_3(x,\dot{a}_i) + u_i$$

for i = t, b, m, with  $\theta_{1,m} = \theta_{2,m} = 0$ ,  $\tilde{a}_m = 0$  and  $u_i \in \mathcal{C}^{2,\alpha}(D_i(\epsilon_i))$ .

**Proof.** A Jacobi function is defined by

$$\left\langle \frac{d}{dt}_{|t=0} \left( X_{\gamma(t)} + u(t)\widetilde{N}(\gamma(t)) \right), N \right\rangle.$$
(2.7)

We start observing that  $X_{\gamma(t)}$  in  $D_i^*(\epsilon_i)$  is given by

$$\frac{x_1}{|x|^2}e_1(\theta_{1,i}(t),\theta_{2,i}(t)) - \frac{x_2}{|x|^2}e_2(\theta_{1,i}(t),\theta_{2,i}(t)) + (-a_i(t)\ln|x| + h_i(x))e_3(\theta_{1,i}(t),\theta_{2,i}(t)),$$

for i = t, b and in  $D_m^*(\epsilon_m)$  by

$$\left(\frac{1}{x}, -a_m(t)\ln|x| + h_m(x)\right).$$

We recall (see (2.5)) that  $a_i(0) = \tilde{a}_i$  with  $\tilde{a}_m = 0$ ,  $\gamma'(0) = (\dot{y}, u'(0))$ . To obtain  $\frac{d}{dt}_{|t=0} X_{\gamma(t)}$  we need compute

$$\dot{e}_i = \frac{d}{dt} e_i(\varphi(t), \phi(t)) = \frac{\partial e_i(\varphi, \phi)}{\partial \varphi} \frac{d\varphi}{dt} + \frac{\partial e_i(\varphi, \phi)}{\partial \phi} \frac{d\phi}{dt}$$

and to evaluate it for t = 0. We will apply the result in the case  $\varphi(t) = \theta_{1,i}(t), \phi(t) = \theta_{2,i}(t)$ . So we suppose that  $\varphi(0) = \phi(0) = 0$ . To this aim it is useful to observe that from the equation (2.2) and the following ones we obtain

$$\left(\frac{\partial e_1(\varphi,\phi)}{\partial\varphi}\right)_{|\varphi=\phi=0} = -\left(\frac{\partial e_2(\varphi,\phi)}{\partial\phi}\right)_{|\varphi=\phi=0} = e_3,$$

$$\left(\frac{\partial e_1(\varphi,\phi)}{\partial\phi}\right)_{|\varphi=\phi=0} = \left(\frac{\partial e_2(\varphi,\phi)}{\partial\varphi}\right)_{|\varphi=\phi=0} = 0,$$

$$\left(\frac{\partial e_3(\varphi,\phi)}{\partial\varphi}\right)_{|\varphi=\phi=0} = -e_1, \quad \left(\frac{\partial e_3(\varphi,\phi)}{\partial\phi}\right)_{|\varphi=\phi=0} = e_2.$$

Then it is easy to obtain

$$\begin{split} \dot{e}_1(0) &= \varphi'(0) e_3, \\ \dot{e}_2(0) &= -\phi'(0) e_3, \\ \dot{e}_3(0) &= -\varphi'(0) e_1 + \phi'(0) e_2. \end{split}$$

Then in  $D_i^*(\epsilon_i)$ 

$$\frac{d}{dt}_{|t=0} X_{\gamma(t)} = \frac{x_1}{|x|^2} \theta'_{1,i}(0) e_3 - \frac{x_2}{|x|^2} (-\theta'_{2,i}(0) e_3) + (-a_i(0) \ln |x| + h_i(x)) (-\theta'_{1,i}(0) e_1 + \theta'_{2,i}(0) e_2) + (-a'_i(0) \ln |x|) e_3.$$

Reordering the summands we get

$$\left\langle \frac{d}{dt}_{|t=0} X_{\gamma(t)}, N \right\rangle = \theta_{1,i}'(0) f_1(x,i) + \theta_{2,i}'(0) f_2(x,i) + f_3(x,\dot{a}_i)$$

As for the last term of (2.7), we recall that u(0) = 0 (see (2.5)) and, on  $D_i^*(\epsilon_i)$ , we defined

$$\widetilde{N}(\gamma(t)) = \frac{e_3(\theta_{1,i}(t), \theta_{2,i}(t))}{\langle N, e_3(\theta_{1,i}(t), \theta_{2,i}(t)) \rangle}$$

Then  $\frac{d}{dt}u(t)\widetilde{N}(\gamma(t))$ , evaluated in t = 0, is equal to

$$u'(0)\widetilde{N}(\gamma(0)) + u(0)\frac{d}{dt}\widetilde{N}(\gamma(t))|_{t=0} = u'(0).$$

If  $u_i$  denotes the restriction of u'(0) to  $D_i^*(\epsilon_i)$  for i = t, b, m, then the result is obvious.

The following lemma shows the existing relation between the logarithmic growths and the Jacobi operator.

**Lemma 22.** Let  $U, V \in \mathcal{C}^{2,\alpha}(M_k)$  be the functions defined in  $D_i^*(\epsilon_i)$ , for  $i \in \{t, b, m\}$ , by

$$U_i(x) = \theta_{1,i} f_1(x,i) + \theta_{2,i} f_2(x,i) + f_3(x,a_i) + u_i(x)$$

and

$$V_i(x) = \varphi_{1,i} f_1(x,i) + \varphi_{2,i} f_2(x,i) + f_3(x,b_i) + v_i(x),$$

with  $\theta_{j,i}, \varphi_{j,i} \in \mathbb{R}$  and  $\tilde{a}_m = 0, \theta_{j,m} = \varphi_{j,m} = 0, j = 1, 2, and u_i, v_i \in \mathcal{C}^{2,\alpha}(D_i(\epsilon_i))$ . Then we have

$$\int_{\overline{M_k}} (U\overline{L}V - V\overline{L}U)d\overline{A} = \int_{M_k} (ULV - VLU)dA =$$
$$= 2\pi \sum_{i \in \{t,b\}} \left[ \langle \Phi_i, \nabla u_i(0) \rangle - \langle \Theta_i, \nabla v_i(0) \rangle \right] + 2\pi \sum_{i \in \{t,b,m\}} [b_i u_i(0) - a_i v_i(0)],$$
with  $\Theta_i = (\theta_{1,i}, \theta_{2,i}), \ \Phi_i = (\varphi_{1,i}, \varphi_{2,i}) \ and \ \nabla \cdot = (\partial_{x_1} \cdot, \partial_{x_2} \cdot).$ 

**Proof.** In the following we will need use the Green identity, so we must use the conformal coordinate  $z = re^{i\alpha}$  around the ends  $p_t, p_b$  and  $p_m$ . Thus for the catenoidal type ends, we have

$$\frac{1}{x} = \frac{\tilde{a}_i}{2z}(1+|z|^2+zt_i(z)) = \frac{s_i(z)}{z}$$

with  $s_i(0) = \frac{\tilde{a}_i}{2} \in \mathbb{R}$ , where  $\tilde{a}_t$ ,  $\tilde{a}_b$  are the logarithmic growths of the top end and of the bottom end of  $M_k$ .

As for the planar end

$$\frac{1}{x} = \frac{s_m(z)}{z}$$

where  $s_m(0) = c$ . We can assume that c = 1.

Since we want to find the expressions of U and V near the ends in terms of coordinates  $(r, \alpha)$ , it is useful to observe that from (2.1) for a generic end with logarithmic growth a it holds that:

$$\langle N, e_3 \rangle = Q^{-\frac{1}{2}} = \left( 1 + |x|^2 (a^2 + |x|^2 |\nabla_0 h|^2 - 2a \langle x, \nabla_0 h \rangle) \right)^{-\frac{1}{2}},$$
  
 
$$\langle N, e_1 \rangle = \langle N, e_3 \rangle \operatorname{Re}(-a\bar{x} + \bar{x}^2 \nabla_0 h),$$
  
 
$$\langle N, e_2 \rangle = \langle N, e_3 \rangle \operatorname{Im}(-a\bar{x} + \bar{x}^2 \nabla_0 h).$$

Then in a neighbourhood of each end we can write:

$$\langle N, e_3 \rangle = \left(1 + O(|x|^2)\right)^{-\frac{1}{2}} = 1 + O(|x|^2),$$
 (2.8)

$$\langle N, e_1 \rangle = \left( 1 + O(|x|^2) \right) \left( -ax_1 + O(\bar{x}^2) \right) = -ax_1 + O(|x|^2), \tag{2.9}$$

$$\langle N, e_2 \rangle = (1 + O(|x|^2)) (ax_2 + O(\bar{x}^2)) = ax_2 + O(|x|^2).$$
 (2.10)

In the coordinates  $(r, \alpha)$   $U_i$  and  $V_i$  have the following expression:

$$U_i(r) = \theta_{1,i} f_1(r,i) + \theta_{2,i} f_2(r,i) + f_3(r,a_i) + u_i(r),$$
  
$$V_i(r) = \varphi_{1,i} f_1(r,i) + \varphi_{2,i} f_2(r,i) + f_3(r,b_i) + v_i(r)$$

where

$$f_1(r,i) = \frac{\tilde{a}_i \cos \alpha}{2r} + O(r \ln r),$$
  
$$f_2(r,i) = \frac{\tilde{a}_i \sin \alpha}{2r} + O(r \ln r),$$
  
$$f_3(r,a) = -a \ln r + O(r).$$

If  $D_i(0,r)$  are conformal disks and  $M(r) = M \setminus (\bigcup_{i \in \{t,b,m\}} D_i(0,r))$ , then the conformal invariance of the integral implies:

$$I(r) = \int_{M(r)} (ULV - VLU) \, dA = \int_{\partial M(r)} \left( U \frac{\partial V}{\partial \eta} - V \frac{\partial U}{\partial \eta} \right) \, ds_0 = -\sum_{i \in \{t,b,m\}} \int_{\partial D_i(0,r)} \left( U_i \frac{\partial V_i}{\partial r} - V_i \frac{\partial U_i}{\partial r} \right) |dz|, \tag{2.11}$$

where dA is the area mesure associated to  $ds_0$ ,  $\eta$  is the exterior conormal field to the immersion along  $\partial M(r)$  and  $|dz| = rd\alpha$ . To get the lemma it will be sufficient to let r go to zero.

Of course we have for  $i \in \{t, b, m\}$ :

$$\frac{\partial U_i}{\partial r} = \theta_{1,i} \frac{\partial f_1(r)}{\partial r} + \theta_{2,i} \frac{\partial f_2(r)}{\partial r} + \frac{\partial f_3(r,a_i)}{\partial r} + \frac{\partial u_i(r)}{\partial r}$$

and a similar expression for  $\frac{\partial V_i}{\partial r}$  :

$$\frac{\partial V_i}{\partial r} = \varphi_{1,i} \frac{\partial f_1(r)}{\partial r} + \varphi_{2,i} \frac{\partial f_2(r)}{\partial r} + \frac{\partial f_3(r,b_i)}{\partial r} + \frac{\partial v_i(r)}{\partial r}.$$

As for the functions  $u_i$  and  $v_i$ , we will adopt in the neighbourhood of the origin an expression of the form (we recall that  $z = z_1 + iz_2 = re^{i\alpha}$ ):

$$l = l(0) + r \cos \alpha (\partial_{z_1} l)(0) + r \sin \alpha (\partial_{z_2} l)(0) + O(r^2).$$
(2.12)

Now we proceed with the evaluation of each summand that appears in (2.11). For  $i \in \{t, b, m\}$  we have (to simplify the notation, we will omit the dependence on r and i)

$$\begin{split} \lim_{r \to 0} \int_{\partial D_i(0,r)} \left( U_i \frac{\partial V_i}{\partial r} - V_i \frac{\partial U_i}{\partial r} \right) |dz| = \\ \lim_{r \to 0} \left( \int_{\{|z|=r\}} \left( \varphi_{1,i} u_i(z) \frac{\partial f_1}{\partial r} - \theta_{1,i} v_i(z) \frac{\partial f_1}{\partial r} \right) + \left( \frac{\partial v_i}{\partial r} \theta_{1,i} f_1 - \frac{\partial u_i}{\partial r} \varphi_{1,i} f_1 \right) + \\ + \left( \varphi_{2,i} u_i(z) \frac{\partial f_2}{\partial r} - \theta_{2,i} v_i(z) \frac{\partial f_2}{\partial r} \right) + \left( \frac{\partial v_i}{\partial r} \theta_{2,i} f_2 - \frac{\partial u_i}{\partial r} \varphi_{2,i} f_2 \right) + \\ \left( u_i(z) \frac{\partial f_3(b_i)}{\partial r} - v_i(z) \frac{\partial f_3(a_i)}{\partial r} \right) + \left( u_i \frac{\partial v_i}{\partial r} - v_i \frac{\partial v_i}{\partial r} \right) |dz| \right). \end{split}$$

We define (the expression of l is given by (2.12)):

$$G(l) = \lim_{r \to 0} \int_{\{|z|=r\}} l(r) \frac{\partial f_1}{\partial r} |dz| =$$

$$-\lim_{r \to 0} \int_{\{|z|=r\}} \left( l(0) + r(\cos\alpha(\partial_{z_1}l)(0) + \sin\alpha(\partial_{z_2}l)(0)) + O(r^2) \right) \cdot \left( \frac{\tilde{a}_i \cos\alpha}{2r^2} + O(\ln r) \right) r d\alpha = \\ = -\lim_{r \to 0} \int_{\{|z|=r\}} \frac{\tilde{a}_i \cos^2\alpha}{2} (\partial_{z_1}l)(0) d\alpha = -\frac{\pi}{2} \tilde{a}_i (\partial_{z_1}l)(0)$$

and in a similar way:

$$T(l) = \lim_{r \to 0} \int_{\{|z|=r\}} l(r) \frac{\partial f_2}{\partial r} |dz| = -\lim_{r \to 0} \int_{\{|z|=r\}} \left( l(0) + r(\cos \alpha (\partial_{z_1} l)(0) + \sin \alpha (\partial_{z_2} l)(0)) + O(r^2) \right) \cdot \left( \frac{\tilde{a}_i \sin \alpha}{2r^2} + O(\ln r) \right) r d\alpha = -\lim_{r \to 0} \int_{\{|z|=r\}} \frac{\tilde{a}_i \sin^2 \alpha}{2} (\partial_{z_2} l)(0) d\alpha = -\frac{\pi}{2} \tilde{a}_i (\partial_{z_2} l)(0).$$

Then we can conclude that for  $i \in \{t, b, m\}$  :

$$\lim_{r \to 0} \int_{\{|z|=r\}} (\varphi_{1,i}u_i(z) - \theta_{1,i}v_i(z)) \frac{\partial f_1}{\partial r}) |dz| =$$
  
$$\varphi_{1,i}G(u_i) - \theta_{1,i}G(v_i) = \frac{\pi}{2} \tilde{a}_i(\theta_{1,i}(\partial_{z_1}v_i)(0) - \varphi_{1,i}(\partial_{z_1}u_i)(0)).$$

In the same way we get

$$\lim_{r \to 0} \int_{\{|z|=r\}} \left(\varphi_{2,i} u_i(z) - \theta_{2,i} v_i(z)\right) \frac{\partial f_2}{\partial r} |dz| = \varphi_{2,i} T(u_i) - \theta_{2,i} T(v_i) = \frac{\pi}{2} \tilde{a}_i (\theta_{2,i} (\partial_{z_2} v_i)(0) - \varphi_{2,i} (\partial_{z_2} u_i)(0)).$$

We define another couple of functions:

$$R(l) = \lim_{r \to 0} \int_{\{|z|=r\}} \frac{\partial l}{\partial r} f_1 |dz| =$$
$$\lim_{r \to 0} \int_{\{|z|=r\}} (\cos \alpha (\partial_{z_1} l)(0) + \sin \alpha (\partial_{z_2} l)(0) + O(r)) \cdot \left(\frac{\tilde{a}_i \cos \alpha}{2r} + O(r \ln r)\right) r d\alpha =$$
$$\lim_{r \to 0} \int_{\{|z|=r\}} \frac{\tilde{a}_i \cos^2 \alpha}{2} (\partial_{z_1} l)(0) d\alpha = \frac{\pi}{2} \tilde{a}_i (\partial_{z_1} l)(0)$$

and

$$F(l) = \lim_{r \to 0} \int_{\{|z|=r\}} \frac{\partial l}{\partial r} f_2 |dz| = \lim_{r \to 0} \int_{\{|z|=r\}} (\cos \alpha (\partial_{z_1} l)(0) + \sin \alpha (\partial_{z_2} l)(0) + O(r)) \cdot \left(\frac{\tilde{a}_i \sin \alpha}{2r} + O(r \ln r)\right) r d\alpha = \lim_{r \to 0} \int_{\{|z|=r\}} (\cos \alpha (\partial_{z_1} l)(0) + \sin \alpha (\partial_{z_2} l)(0) + O(r)) \cdot \left(\frac{\tilde{a}_i \sin \alpha}{2r} + O(r \ln r)\right) r d\alpha = \lim_{r \to 0} \int_{\{|z|=r\}} (\cos \alpha (\partial_{z_1} l)(0) + \sin \alpha (\partial_{z_2} l)(0) + O(r)) \cdot \left(\frac{\tilde{a}_i \sin \alpha}{2r} + O(r \ln r)\right) r d\alpha$$

$$\lim_{r \to 0} \int_{\{|z|=r\}} \frac{\tilde{a}_i \sin^2 \alpha}{2} (\partial_{z_2} l)(0) d\alpha = \frac{\pi}{2} \tilde{a}_i (\partial_{z_2} l)(0).$$

Then we find:

$$\lim_{r \to 0} \int_{\{|z|=r\}} \left( \frac{\partial v_i}{\partial r} \theta_{1,i} f_1 - \frac{\partial u_i}{\partial r} \varphi_{1,i} f_1 \right) |dz| =$$
  
$$\theta_{1,i} R(v_i) - \varphi_{1,i} R(u_i) = \frac{\pi}{2} \tilde{a}_i (\theta_{1,i} (\partial_{z_1} v_i) (0) - \varphi_{1,i} (\partial_{z_1} u_i) (0)).$$

Analogously:

$$\lim_{r \to 0} \int_{\{|z|=r\}} \left( \frac{\partial v_i}{\partial r} \theta_{2,i} f_2 - \frac{\partial u_i}{\partial r} \varphi_{2,i} f_2 \right) |dz| = \theta_{2,i} F(v_i) - \varphi_{2,i} F(u_i) = \frac{\pi}{2} \tilde{a}_i (\theta_{2,i} (\partial_{z_2} v_i)(0) - \varphi_{2,i} (\partial_{z_2} u_i)(0)).$$

As for the fifth summand, we have

$$\lim_{r \to 0} \int_{\{|z|=r\}} \left( u_i(z) \frac{\partial f_3(b_i)}{\partial r} - v_i(z) \frac{\partial f_3(a_i)}{\partial r} \right) |dz| = -\lim_{r \to 0} \int_{\{|z|=r\}} \left( \left( u_i(0) + r \cos \alpha (\partial_{z_1} u_i)(0) + r \sin \alpha (\partial_{z_2} u_i)(0) + O(r^2) \right) \frac{b_i}{r} - \left( v_i(0) + r \cos \alpha (\partial_{z_1} v_i)(0) + r \sin \alpha (\partial_{z_2} v_i)(0) + O(r^2) \right) \frac{a_i}{r} \right) r d\alpha = -2\pi (b_i u_i(0) - a_i v_i(0)).$$

To finish we show that

$$\lim_{r \to 0} \int_{\{|z|=r\}} \left( u_i \frac{\partial v_i}{\partial r} - v_i \frac{\partial v_i}{\partial r} \right) |dz| = 0.$$

In fact:

$$\lim_{r \to 0} \int_{\{|z|=r\}} u_i \frac{\partial v_i}{\partial r} |dz| = \lim_{r \to 0} \int_{\{|z|=r\}} \left( (u_i(0) + O(r))(\cos \alpha (\partial_{z_1} v_i)(0) + \sin \alpha (\partial_{z_2} v_i)(0) + O(r)) \right) r d\alpha = 0.$$

A similar result is true for the other summand.

If we collect the previous results, we find that for i=t,b,m :

$$\lim_{r \to 0} \int_{\partial D_i(0,r)} \left( U_i \frac{\partial V_i}{\partial r} - V_i \frac{\partial U_i}{\partial r} \right) |dz| =$$

$$-\pi \tilde{a}_i \left[ (\varphi_{1,i}(\partial_{z_1} u_i)(0) - \theta_{1,i}(\partial_{z_1} v_i)(0)) + (\varphi_{2,i}(\partial_{z_2} u_i)(0) - \theta_{2,i}(\partial_{z_2} v_i)(0)) \right] \\ -2\pi (b_i u_i(0) - a_i v_i(0)) = 0,$$

with  $\theta_{j,m} = \varphi_{j,m} = 0$  for j = 1, 2.

In conclusion we have:

$$\lim_{r \to 0} \int_{M(r)} \left( ULV - VLU \right) dA = \pi \tilde{a}_i \sum_{i \in \{t, b\}} \left[ \varphi_{1,i}(\partial_{z_1} u_i)(0) - \theta_{1,i}(\partial_{z_1} v_i)(0) \right] + \pi \tilde{a}_i \sum_{i \in \{t, b\}} \left[ \varphi_{2,i}(\partial_{z_2} u_i)(0) - \theta_{2,i}(\partial_{z_2} v_i)(0) \right] + 2\pi \sum_{i \in \{t, b, m\}} \left[ b_i u_i(0) - a_i v_i(0) \right].$$

We must do a change of variables to return in the graph coordinate. It is sufficient to observe that  $\tilde{}$ 

$$\frac{1}{x} = \frac{\tilde{a}_i}{2z} + O(1)$$

at each catenoidal type end. Then we get

$$\partial_{z_1} u_i(0) = \frac{2}{\tilde{a}_i} \partial_{x_1} u_i(0) \quad \partial_{z_2} u_i(0) = \frac{2}{\tilde{a}_i} \partial_{x_2} u_i(0)$$

and the same equations involving the functions  $v_i$ . After a change of sign we can conclude

$$\int_{M_k} (ULV - VLU) dA =$$

$$2\pi \sum_{i \in \{t,b\}} [\varphi_{1,i}(\partial_{z_1} u_i)(0) - \theta_{1,i}(\partial_{z_1} v_i)(0)] +$$

$$+2\pi \sum_{i \in \{t,b\}} [\varphi_{2,i}(\partial_{z_2} u_i)(0) - \theta_{2,i}(\partial_{z_2} v_i)(0)] + 2\pi \sum_{i \in \{t,b,m\}} [b_i u_i(0) - a_i v_i(0)].$$

Reordering the terms, we get the statement of the lemma.

## 2.3.2 The properties of the kernel and of the range of the Jacobi operator

Let  $B = B(M_k) \subset C^{2,\alpha}(M_k)$  the space of functions v such that their expression in a neighbourhood of  $p_i$ , with i = t, b, m, is

$$\theta_{1,i}f_1(x,i) + \theta_{2,i}f_2(x,i) + f_3(x,a_i) + v_i(x), \qquad (2.13)$$

in the graph coordinate x (here we use the same notation of the lemma 22) with  $v_i \in \mathcal{C}^{2,\alpha}(\{|x| \leq \epsilon\})$ . We recall that  $\tilde{a}_m = 0, \theta_{j,m} = 0$  for j = 1, 2.

We are interested to study the kernel and the image of the "compactified" Jacobi operator  $\overline{L}: B(M_k) \to \mathcal{C}^{0,\alpha}(\overline{M_k})$ . We define the following subspaces of the Banach space B:

$$J = J(M_k) = \ker(\overline{L}), \quad K = K(M_k) = J \cap \mathcal{C}^{2,\alpha}(\overline{M_k}), \quad K_0 = K_0(\overline{M_k}) = \overline{L}(B)^{\perp}.$$

The elements of the space K are the Jacobi functions on  $M_k$  bounded at the ends. From the previous definitions it follows that

$$\overline{L}: B(M_k) = J \oplus J^{\perp} \to \overline{L}(B) \oplus K_0.$$

Lemma 23. In the situation described above, it holds that:

- 1.  $K_0 = \{v \in K; \partial_{x_i}v_j(0) = 0, \text{ for } i = 1, 2, j = t, b, \text{ and } v_j(0) = 0, \text{ for } j = t, b, m\} = \{\langle N, e_3 \times p \rangle\}$
- 2.  $\dim J = 7 + \dim K_0 = 8.$

#### Proof.

1. Given  $v \in K$ , we have  $v \in K_0$  if and only if  $\int_{\overline{M_k}} v \overline{L} U d\overline{A} = 0 \ \forall U \in B$ . Here we continue to use the notation of lemma 22. Then we suppose that U on a neighbourhood of the end  $p_i$ , i = t, b, m has the following expression:

$$\varphi_{1,i}f_1(x,i) + \varphi_{2,i}f_2(x,i) + f_3(x,b_i) + u_i(x),$$

with  $\tilde{a}_m = 0$ ,  $\varphi_{j,m} = 0$ , j = 1, 2. Then by the lemma 22 we get

$$\int_{\overline{M_k}} U\bar{L}vd\bar{A} = \int_{M_k} ULvdA =$$
$$= 2\pi \sum_{i \in \{t,b\}} \left[ \langle \Phi_i, \nabla u_i(0) \rangle - \langle \Theta_i, \nabla v_i(0) \rangle \right] + 2\pi \sum_{i \in \{t,b,m\}} \left[ a_i u_i(0) - b_i v_i(0) \right]$$

for each  $u_i$ . This is equivalent to

$$2\pi \sum_{i \in \{t,b\}} \langle \Theta_i, \nabla v_i(0) \rangle + 2\pi \sum_{i \in \{t,b,m\}} b_i v_i(0) = 0.$$

This gives  $K_0$ . Now we have to determine the Jacobi fields that are the generators of the space K. Thanks to the works [29] and [30] of S. Nayatani and result contained in 1, for all  $k \ge 1$ , the bounded Jacobi fields are associated to the following isometries

of the ambient space: the three translations along the coordinate axes and the rotation about the vertical axis  $e_3$ . The space  $K_0$  is generated only by the Jacobi functions which satisfy the conditions just proved. Making use of (2.8), (2.9) and (2.10) with the appropriate values of the logarithmic growths, we want to determine which of the following functions belongs to  $K_0 : \langle N, e_3 \rangle, \langle N, e_1 \rangle, \langle N, e_2 \rangle$ . We find the following relations:

$$\partial_{x_j} \langle N, e_3 \rangle (p_i) = 0, \quad \partial_{x_j} \langle N, e_1 \rangle (p_i) = -\tilde{a}_i \delta_{1,j}, \quad \partial_{x_j} \langle N, e_2 \rangle (p_i) = \tilde{a}_i \delta_{2,j},$$

with j = 1, 2. So we can conclude that these functions do not belong to the space  $K_0$ .

Now we consider the Jacobi function associated to the rotation about the vertical axis, that is  $\langle N, e_3 \times p \rangle = det(e_3, p, N)$ , where  $p = (s_1, s_2, s_3)$  denotes the position vector. We observe that its expression is given by

$$s_1 \langle N, e_2 \rangle - s_2 \langle N, e_1 \rangle = \frac{x_1}{|x|^2} (a_i x_2 + O(|x|^2)) - \left(-\frac{x_2}{|x|^2}\right) (-a_i x_1 + O(|x|^2)) = O(x).$$

Then it is clear that  $K_0$  is generated by this Jacobi function.

2. We consider the space  $V \subset B$  of the functions defined on the disks  $D_i^*(\epsilon_i)$  by

$$\theta_{1,i}f_1(x,i) + \theta_{2,i}f_2(x,i) + f_3(x,a_i)$$

It is a 7-dimensional space: in fact a function in V is determined by the values of the following parameters:  $a_t, a_b, a_m, \theta_{1,t}, \theta_{1,b}, \theta_{2,t}, \theta_{2,b}$ . The spaces B and V can be decomposed in the following way:

$$B = V \oplus \mathcal{C}^{2,\alpha} \quad V = V_1 \oplus V_2,$$

where  $V_1 = \{f \in V : \overline{L}f \in \overline{L}(\mathcal{C}^{2,\alpha})\}$  and  $V_2$  is a supplementary space. Then we have  $\overline{L}(B) = \overline{L}(V_2) \oplus \overline{L}(\mathcal{C}^{2,\alpha})$ . Since  $K_0 = \overline{L}(B)^{\perp}$  and  $K = \overline{L}(\mathcal{C}^{2,\alpha})$  we deduce

$$\dim K_0 = \operatorname{codim} \bar{L}(B) = \operatorname{codim} \bar{L}(\mathcal{C}^{2,\alpha}) - \dim \bar{L}(V_2) =$$
$$\dim K - \dim V_2 = \dim K - \dim V + \dim V_1,$$

that is

$$\dim K_0 = \dim K - 7 + \dim V_1.$$
 (2.14)

Now we consider the restriction to  $J = \ker(\overline{L})$  of the projection  $\pi : B \to V$ . It is clear that  $\ker(\pi_{|J}) = K = J \cap \mathcal{C}^{2,\alpha}$ , then given a function  $f \in J$  such that f = v + uwith  $v \in V$  and  $u \in \mathcal{C}^{2,\alpha}$ , we have

$$0 = \bar{L}f = \bar{L}v + \bar{L}u$$

and  $\pi(f) = v \in V_1$ . Furthermore, for any  $v \in V_1$  there exists  $v' \in C^{2,\alpha}$  such that  $\overline{L}v = \overline{L}v'$ , that is  $v - v' \in J$ . Then  $\pi(J) = V_1$  and

$$\dim J = \dim \ker(\pi_{|J}) + \dim \operatorname{Im}(\pi_{|J}) = \dim K + \dim V_1.$$
(2.15)

From the equations (2.14) and (2.15) we get

$$\dim J = 7 + \dim K_0 = 8.$$

**Remark 24.** In [34] a minimal surface is defined to be non degenerate if the space K of the Jacobi fields induced by the isometries of the ambient space contains the space  $K_0$ . Thanks to the result of lemma 23 we can conclude that the Costa-Hoffman-Meeks surface  $M_k$  is non degenerate for all  $k \ge 1$  with respect to this definition.

### 2.4 The proof of the main result

We consider again the immersion  $X_y + u\tilde{N}(y)$  (see (2.3)) and its mean curvature function H(y, u), where  $y = (a_t, a_b, a_m, \theta_{1,t}, \theta_{1,b}, \theta_{2,t}, \theta_{2,b})$ . We denote with  $e_3(y)$  the unit vector defined in  $D_i^*(\epsilon_i)$ , for i = t, b, m, by  $e_3(\theta_{1,i}, \theta_{2,i})$  for i = t, b and by  $e_3(0, 0)$  for i = m. We denote with  $ds_{y,u}^2$  the metric induced by  $X_y + u\tilde{N}(y)$ . In the following, the subindex  $\cdot_{y,u}$  denotes that the corresponding object is computed with respect to the metric  $ds_{y,u}^2$ . We recall that  $\mathcal{A}$  and  $\mathcal{U}$  denote, respectively a neighbourhood of  $(\tilde{a}_t, \tilde{a}_b, 0)$  and a neighbourhood of zero in  $\mathcal{C}^{2,\alpha}(\overline{M_k})$ .

**Lemma 25.** The mean curvature function H(y, u), with  $(y, u) \in \mathcal{A} \times [-\epsilon, \epsilon]^4 \times \mathcal{U}$ , is orthogonal to  $\langle N_{y,u}, e_3(y) \times p_{y,u} \rangle$ , where  $N_{y,u}$  is the Gauss map of  $X_y + u\tilde{N}(y)$  and  $p_{y,u}$  is the position vector.

**Proof.** Given  $\epsilon > 0$  small we consider a compact domain  $M_k(\epsilon)$  of  $M_k$  obtained by removing the disks  $D_i(\epsilon)$  which parametrize the ends. We have:

$$2\int_{M(\epsilon)} H(y,u)\langle N_{y,u}, e_3(y) \times p_{y,u} \rangle dA_{y,u} = -\int_{M(\epsilon)} \Delta_{y,u} \det(N_{y,u}, X_y + u\widetilde{N}(y), e_3(y)) dA_{y,u} = -\int_{\partial M(\epsilon)} \det(\eta_{y,u}, X_y + u\widetilde{N}(y), e_3(y)) ds_{y,u},$$

where  $\eta_{y,u}$  is the exterior conormal field to  $X_y + u\widetilde{N}(y)$ .

We parametrize  $\partial M_k(\epsilon)$  by three disjoint copies of the boundary of a disk of radius  $\epsilon$ . On this boundary we assume  $x = \epsilon e^{i\psi}$ ,  $\psi \in [0, 2\pi]$ . In each neighbourhood of the ends we will use the appropriate frame to simplify the computation. That is, in the neighbourhood of the end  $p_l$ , l = t, b, m, we will adopt the frame  $F(\theta_{1,l}, \theta_{2,l})$  with  $\theta_{1,m} = \theta_{2,m} = 0$ . We set

$$\alpha_{\epsilon,l}(y) = (X_y + u\widetilde{N}(y))(\epsilon e^{i\psi}) = \left(\frac{1}{x}, v_l\right)$$

with  $v_l = -a_l \ln |x| + h_l(x)$ . Then

$$\eta_{y,u,l}ds_{y,u} = -\alpha'_{\epsilon,l} \times N_{y,u}d\psi$$

where  $\alpha'_{\epsilon,l} = \left(\frac{-i}{x}, \langle ix, \nabla_0 v_l \rangle\right)$ , for l = t, b, m, is the derivative of  $\alpha_{\epsilon,l}$  with respect to  $\psi$ ,  $i = \sqrt{-1}$  and  $\nabla_0$  denotes the gradient computed with respect to  $ds_0^2$ , the flat metric of the *x*-plane. It is possible to prove that (see p. 194 [34])

$$\eta_{y,u,l}ds_{y,u} = \left[\left(\frac{1}{x}, a_l\right) + \left(\frac{O(\epsilon^2)}{x} + O(\epsilon^2), O(\epsilon)\right)\right]d\psi.$$

Then

$$\int_{|x|=\epsilon} \det(X_y + u\widetilde{N}(y), e_3(y), \eta_{y,u}) ds_{y,u} = \sum_{l \in \{t,b,m\}} \int_{|x|=\epsilon} \det\left(\left(\frac{1}{x}, v_l\right), (0,1), \left(\frac{1}{x}, a_l\right) + \left(\frac{O(\epsilon^2)}{x} + O(\epsilon^2), O(\epsilon)\right)\right) d\psi.$$

It is easy to conclude that this expression converges to zero as  $\epsilon$  goes to zero.

And to finish, here it is our main result.

We set  $c = (0, \tilde{a}_t, \tilde{a}_b, 0, 0, 0, 0, 0, 0)$  and we consider the function  $v \in B$  given by (2.13). We recall that  $\tilde{y} = (\tilde{a}_t, \tilde{a}_b, 0, 0, 0, 0, 0)$ .

**Theorem 26.** For each possible choice of the limit values of the normal vectors of the three ends, there is, up to isometries, a 1-dimensional real-analytic family of smooth minimal deformations of  $M_k$ , for  $k \ge 1$ , letting the planar end horizontal.

**Proof.** We consider the map

$$F: \mathbb{R} \times \mathcal{A} \times [-\epsilon, \epsilon]^4 \times \mathcal{U} \longrightarrow \mathcal{C}^{0,\alpha}(\overline{M_k})$$
$$(r, y, u) \longrightarrow \overline{H}(y, u) + r \langle N_{y,u}, e_3(y) \times p_{y,u} \rangle$$

where  $N_{y,u}$  is the Gauss map of  $X_y + u\widetilde{N}(y)$ . The map F is real analytic. Since the values  $r = 0, y = \tilde{y}, u = 0$  parametrize the Costa-Hoffman-Meeks surface  $M_k$ , the differential of F at c,

$$DF_c: \mathbb{R}^8 \times B(M_k) \longrightarrow \mathcal{C}^{0,\alpha}(\overline{M_k})$$

is given by

$$DF_c(r, y, u) = \frac{1}{2}\overline{L}(v_y + u) + r\langle N_{y,u}, e_3(y) \times p_{y,u} \rangle.$$

Since  $\overline{L}(B)^{\perp} = K_0$ , we have Ker $DF = \{0\} \times J$ . The differential for (r, y, u) = c is surjective and its kernel has dimension 8. Using the implicit function theorem we find a neighborhood  $\mathcal{W}$  of c in  $\mathbb{R}^8 \times \mathcal{U}$  such that  $\mathcal{V} = F^{-1}(0) \cap \mathcal{W}$  is a real analytic 8-dimensional manifold. Thanks to the orthogonality between the mean curvature function and  $\langle N_{y,u}, e_3(y) \times p_{y,u} \rangle$ proved in the lemma 25, we can conclude that  $\mathcal{V}$  contains only minimal immersions, hence  $\mathcal{V} \subset \{0\} \times \mathcal{A} \times [-\epsilon, \epsilon]^4 \times \mathcal{U}.$ 

To complete the proof, it remains to observe that up to now we have considered the choice of the parameters  $a_t, a_b, a_m, \theta_{1,t}, \theta_{2,t}, \theta_{1,b}, \theta_{2,b}$  arbitrary. But it's necessary that the null flux condition is satisfied. In our case the flux is given by the sum of the flux of three catenoidal ends. So we have that the sum of three vectors must be the null vector. The direction and the length of each vector are respectively given by the direction of axis of revolution and by the logarithmic growth of the respective catenoidal end. It's easy to understand that these three vectors belong to a same vertical plane, that is we must have always  $\theta_{2,t} = \theta_{2,b}$ . The common value of these angles determines the orientation of this plane (see (2.2)). Furthermore the flux triangle described by the three vectors is uniquely determined by three of the remaining parameters (the logarithmic growths  $a_t, a_b, a_m$  and the angles  $\theta_{1,t}, \theta_{1,b}$ ). It is clear that the choice of the limit values of the normal vectors (in other words of the angles  $\theta_{1,t}, \theta_{1,b}$ ) of the three ends determines in unique way, up to a dilation, the flux triangle. So we can conclude that for each possible choice of the flux triangle, there exists a smooth 1-parameter family of minimal surfaces that are deformations of the surface  $M_k$ . 

## Chapter 3

# A Costa-Hoffman-Meeks type surface in $\mathbb{H}^2\times\mathbb{R}$

## 3.1 Introduction

In the last years the study of the minimal surfaces in the product spaces  $M \times \mathbb{R}$  with  $M = \mathbb{H}^2, \mathbb{S}^2$  has been becoming more and more active. The development of the theory of the minimal surfaces in these spaces started with [36] by H. Rosenberg and continued with [25] and [26] by W. H. Meeks and H. Rosenberg. In [32] B. Nelli and H. Rosenberg showed the existence in  $\mathbb{H}^2 \times \mathbb{R}$  of a rich family of examples including helicoids, catenoids and, solving Plateau problems, of higher topological type examples inspired by the theory of minimal surfaces in  $\mathbb{R}^3$ .

C. Costa in [3, 4] and D. Hoffman and W.H. Meeks in [14], [15] and [16] described in  $\mathbb{R}^3$  a minimal surface of genus  $k \ge 1$  with two ends asymptotic to the two ends of a catenoid and a middle end asymptotic to a plane. We will denote the Costa-Hoffman-Meeks surface of genus  $k \ge 1$  by  $M_k$ .

The aim of this work is to show the existence in the space  $\mathbb{H}^2 \times \mathbb{R}$  of a family of surfaces inspired to  $M_k$ . We shall prove the following result

**Theorem 27.** For all  $k \ge 1$  there exists in  $\mathbb{H}^2 \times \mathbb{R}$  a minimal surface of genus k with three horizontal ends: two catenoidal type ends and a middle planar end.

The construction is based on a gluing procedure. We consider a rescaled version of a compact part of a Costa-Hoffman-Meeks type surface, such that it can be contained in a cylindrical neighbourhood of  $\{0, 0\} \times \mathbb{R} \subset \mathbb{H}^2 \times \mathbb{R}$  of sufficiently small radius. Actually it's possible to prove that, in the same set, the mean curvature of such a surface with respect the standard metric of  $\mathbb{H}^2 \times \mathbb{R}$ , up to an infinitesimal term, can be expressed in

terms of the euclidean one. The main result is proved by a gluing procedure (see for example [11]) usually adopted to construct in  $\mathbb{R}^3$  new examples starting from known minimal surfaces. In particular we glue the surface described above along its three boundary curves to two minimal graphs that are respectively asymptotic to an upper half catenoid and a lower half catenoid defined in  $\mathbb{H}^2 \times \mathbb{R}$  and to a minimal graph asymptotic to  $\mathbb{H}^2 \times \{0\}$ .

The author wishes to thank L. Hauswirth for invaluable conversations.

## 3.2 Minimal graphs in $\mathbb{H}^2 \times \mathbb{R}$

In this work we shall consider the disk model for  $\mathbb{H}^2$ . Let  $x_1, x_2$  denote the coordinates in  $\mathbb{H}^2$  and  $x_3$  the coordinate in  $\mathbb{R}$ . Then the space  $\mathbb{H}^2 \times \mathbb{R}$  is endowed with the metric

$$d\sigma^2 = \frac{dx_1^2 + dx_2^2}{F} + dx_3^2,$$

where

$$F = \frac{1}{4} \left( 1 - x_1^2 - x_2^2 \right)^2 = \frac{1}{4} \left( 1 - r^2 \right)^2.$$

We denote with  $H_u$  the mean curvature of the graph of the function u over a domain in  $\mathbb{H}^2$ . Its expression is

$$2H_u = F \operatorname{div}\left(\frac{\nabla u}{\sqrt{1+F|\nabla u|^2}}\right),\tag{3.1}$$

where div denotes the divergence in  $\mathbb{R}^2$ . For the details of the computation see [32].

Let  $\Sigma_u$  be the graph of the function u. In this section we want to obtain the expression of the mean curvature of the surface  $\Sigma_{u+v}$  that is the normal graph of the function vover  $\Sigma_u$  and close to it. We shall show how it follows from (3.1) that the linearized mean curvature operator, that we denote with  $L_u$ , is given by:

$$F \operatorname{div}\left(\frac{\nabla v}{\sqrt{1+F|\nabla u|^2}} - F\nabla u \frac{\nabla u \cdot \nabla v}{\sqrt{(1+F|\nabla u|^2)^3}}\right).$$
(3.2)

Furthermore we shall give the expression of  $H_{u+v}$ , the mean curvature of the graph of the function u + v, in terms of the mean curvature of  $\Sigma_u$ , that is  $H_u$ . In the following we shall restrict our attention to two cases: the plane, that is u = 0, and (in section 3.6) a part of catenoid defined on the domain  $\{(r, \theta) \in \mathbb{H}^2 | r \in [r_{\epsilon}, 1)\}$ , where  $r_{\epsilon} = \epsilon^{\frac{1}{2}}/2$ .

Here we shall show that:

$$2H_{u+v} = 2H_u + L_u v + FQ_u(\sqrt{F}\nabla v, \sqrt{F}\nabla^2 v), \qquad (3.3)$$

where  $Q_u$  is an operator with bounded coefficients if  $r \in [r_{\epsilon}, 1)$  which satisfies

$$Q_u(0,0) = \nabla Q_u(0,0) = 0.$$

To show this, we start observing that:

$$\frac{1}{\sqrt{1+F|\nabla(u+v)|^2}} = \frac{1}{\sqrt{1+F|\nabla u|^2}} - F\frac{\nabla u \cdot \nabla v}{\sqrt{(1+F|\nabla u|^2)^3}} + Q_{u,1}(v).$$
(3.4)

The operator  $Q_{u,1}(v)$  has the following expression

$$\frac{-F|\nabla v|^2}{\left(1+F|\nabla(u+\bar{t}v)|^2\right)^{3/2}} + \frac{3F^2\left(\nabla u\cdot\nabla v+\bar{t}|\nabla v|^2\right)^2}{\left(1+F|\nabla(u+\bar{t}v)|^2\right)^{5/2}},\tag{3.5}$$

with  $\bar{t} \in (0,1)$ , and it satisfies  $Q_1(0) = \nabla Q_1(0) = 0$ . To prove (3.4) it's sufficient to set

$$f(t) = \frac{1}{\sqrt{1+F|\nabla(u+tv)|^2}}$$

and to write down the Taylor's series of order one of this function and to evaluate it in t = 1. That is  $f(1) = f(0) + f'(0) + \frac{1}{2}f''(\bar{t})$ , with  $\bar{t} \in (0, 1)$ . We insert (3.4) in the expression that defines  $2H_{u+v}$  to get

$$F \operatorname{div}\left(\frac{\nabla(u+v)}{\sqrt{1+F|\nabla u|^2}} - F\nabla(u+v)\frac{\nabla u \cdot \nabla v}{\sqrt{(1+F|\nabla u|^2)^3}} + \nabla(u+v)Q_{u,1}(v)\right) = 2H_u + F\operatorname{div}\left(\frac{\nabla v}{\sqrt{1+F|\nabla u|^2}} - F\nabla u\frac{\nabla u \cdot \nabla v}{\sqrt{(1+F|\nabla u|^2)^3}}\right) + FQ_u(\sqrt{F}\nabla v,\sqrt{F}\nabla^2 v).$$

Since we assume that  $\Sigma_u$  is a minimal surface, we will consider  $H_u = 0$ .

### 3.3 The linearized operator about the hyperbolic plane

Now we restrict our attention to the case of the minimal surfaces close to  $\mathbb{H}^2 \times \{0\}$ , that is the graph of the function u = 0. In this case we obtain immediately from (3.2) that  $L_{u=0} = F\Delta_{eucl}$ , where  $\Delta_{eucl}$  denotes the Laplace operator in the euclidean metric. In this section we will study the mapping properties of the operator  $\mathcal{L} = \Delta_{eucl}$ . In the following we will use the polar coordinates  $r, \theta$ . In particular our aim is to solve in an unique way the problem:

$$\begin{cases} \mathcal{L}w = f & \text{in } S^1 \times [r_0, 1) \\ w|_{r=r_0} = \varphi \end{cases}$$

with  $r_0 \in (0, 1)$ , considering a convenient normed functions space for w, f and  $\varphi$ , so that the norm of w is bounded by the one of f. Since

$$r^{2}\mathcal{L} = \left(r^{2}\frac{\partial^{2}}{\partial r^{2}} + r\frac{\partial}{\partial r} + \frac{\partial^{2}}{\partial \theta^{2}}\right)$$

has separated variables, let us firstly consider the operator

$$L_{\theta} = \partial_{\theta\theta}^2,$$

which acts on  $2\pi$ -periodic even functions. It is uniformly elliptic and self-adjoint. In particular,  $L_{\theta}$  has discrete spectrum  $(\lambda_i)_{i \ge 0} = i^2$ . Each eigenvalue  $\lambda_i$  has multiplicity one. We denote by  $e_i(\theta)$  the eigenfunction associated to  $\lambda_i$ , normalized so that

$$\int_0^{2\pi} (e_i(\theta))^2 \, d\theta = 1.$$

The Hilbert basis  $\{e_i\}_{i\in\mathbb{N}}$  of the space of  $2\pi$ -periodic even functions in  $L^2(S^1)$  induces the following Fourier decomposition of  $L^2$  functions  $g = g(\theta, r)$  which are  $2\pi$ -periodic and even in the variable  $\theta$ ,

$$g(\theta, r) = \sum_{i \ge 0} g_i(r) e_i(\theta).$$

From this, we deduce that the operator  $\mathcal{L}$ , can be decomposed as  $\mathcal{L} = \sum_{i \ge 0} L_i$ , being

$$r^{2}L_{i} = \left(r^{2}\frac{\partial^{2}}{\partial r^{2}} + r\frac{\partial}{\partial r} - \lambda_{i}\right), \text{ for every } i \ge 0.$$

From the observations made above we get that the potential of  $r^2L_i$  is

$$P_i := -\lambda_i = -i^2. \tag{3.6}$$

Now we can prove a lemma which assures that  $\mathcal{L}$  is injective.

**Lemma 28.** Given  $0 < r_0 < r_1 < 1$ , let w be a solution of  $\mathcal{L}w = 0$  on  $S^1 \times [r_0, r_1]$  such that  $w(\cdot, r_0) = w(\cdot, r_1) = 0$ . Then w = 0.

**Proof:** We can decompose  $w = \sum_{i \ge 0} w_i(r) e_i(\theta)$ . Since the potential  $P_i$  of the operator  $L_i$  is negative or zero for every  $i \ge 0$  (see (3.6)) and the operator  $L_i$  is uniformly elliptic, the maximum principle holds. Then we get the lemma 28 from the hypothesis on the boundary conditions.

#### 3.3.1 The mapping properties of the Laplace operator

**Definition 29.** Given  $\ell \in \mathbb{N}$ ,  $\alpha \in (0,1)$ ,  $\mu \in \mathbb{R}$  and the closed interval I, we define

 $\mathcal{C}^{\ell,\alpha}_{\mu}(S^1 \times I)$ 

to be the space of functions  $w := w(\theta, r)$  in  $\mathcal{C}_{loc}^{\ell, \alpha}(S^1 \times I)$  for which the following norm is finite

$$||w||_{\mathcal{C}_{\mu}^{\ell,\alpha}} := ||(1-r^2)^{-\mu}w||_{\mathcal{C}^{\ell,\alpha}(S^1 \times I)}$$

**Proposition 30.** Given  $\mu > 0$ ,  $r_0 \in (0, 1)$ , there exists an operator

$$\begin{array}{rcl} G_{r_0}: & \mathcal{C}^{0,\alpha}_{\mu}(S^1 \times [r_0,1)) & \longrightarrow & \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_0,1)) \\ & f & \longmapsto & w := G_{r_0}(f) \end{array}$$

satisfying the following statements

- (i)  $\mathcal{L} w = f \text{ on } S^1 \times [r_0, 1),$
- (*ii*) w = 0 on  $S^1 \times \{r_0\},\$
- (iii)  $||w||_{\mathcal{C}^{2,\alpha}_{\mu}(S^1\times[r_0,1))} \leq c ||f||_{\mathcal{C}^{0,\alpha}_{\mu}(S^1\times[r_0,1))}$ , for some constant c > 0 which does not depend on  $r_0$ .

**Proof:** As consequence of the lemma 28, the operator  $\mathcal{L} = \Delta_{eucl}$  is injective. Hence, Fredholm alternative let us assure that there exists, an unique  $w \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_0, r_1])$ , with  $w(\theta, r)$  satisfying:

$$\begin{cases} \Delta_{eucl} w = f \quad \text{on } S^1 \times [r_0, r_1] \\ w(\cdot, r_0) = w(\cdot, r_1) = 0. \end{cases}$$
(3.7)

We want to prove the following assertion.

**Assertion 31.** For every  $0 < r_0 < r_1 < 1$ ,  $f \in \mathcal{C}^{0,\alpha}_{\mu}(S^1 \times [r_0, r_1])$  and  $w \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_0, r_1])$ satisfying (3.7), it exists a constant c such that

$$\|w\|_{\mathcal{C}^{0,\alpha}_{\mu}(S^{1}\times[r_{0},r_{1}])} \leqslant c \, \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}(S^{1}\times[r_{0},r_{1}])}.$$

We suppose by contradiction that the assertion 31 is false, that is it does not exist a universal costant for which the previous estimate holds. Then, for each  $n \in \mathbb{N}$ , there exist  $r_{1,n} > r_{0,n}$  and  $f_n, w_n$  satisfying (3.7) (with  $r_{0,n}, r_{1,n}, f_n, w_n$  instead of  $r_0, r_1, f, w$ ) such that

$$||f_n||_{\mathcal{C}^{0,\alpha}_{\mu}(S^1 \times [r_{0,n}, r_{1,n}])} = 1 \text{ and } ||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}(S^1 \times [r_{0,n}, r_{1,n}])} \to +\infty, \text{ when } n \to \infty.$$

Since  $\Omega_n := S^1 \times [r_{0,n}, r_{1,n}]$  is a compact set,  $A_n := \sup_{S^1 \times [r_{0,n}, r_{1,n}]} (1 - r^2)^{-\mu} |w_n|$  is achieved at a point  $(\theta_n, r_n) \in S^1 \times [r_{0,n}, r_{1,n}]$ . We define

$$\widetilde{w}_n(\theta, r) := \frac{(1 - r_n^2)^{-\mu}}{\|w_n\|_{\mathcal{C}^{0,\alpha}_\mu(\Omega_n)}} w_n(\theta, rr_n),$$

and

$$\widetilde{f}_n(\theta, r) := \frac{(1 - r_n^2)^{-\mu}}{\|f_n\|_{\mathcal{C}^{0,\alpha}_\mu(\Omega_n)}} f_n(\theta, rr_n),$$

for all  $(\theta, r) \in S^1 \times I_n$ , with  $I_n = [r_{0,n}/r_n, r_{1,n}/r_n]$ . Clearly,  $A_n \leq ||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_n)}$ , and

$$|\widetilde{w}_n(\theta, r)| \leq (1 - r^2)^{\mu} \frac{(1 - (rr_n)^2)^{-\mu} |w_n(\theta, rr_n)|}{A_n} \leq (1 - r^2)^{\mu}.$$

On the other hand,  $(1-r^2)^{-\mu} |\nabla \widetilde{w}_n| \leq \|\widetilde{w}_n\|_{\mathcal{C}^{2,\alpha}_{\mu}(S^1 \times I_n)} \leq \frac{\|w_n\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Omega_n)}}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_n)}}$ . Thanks to Schauder estimates, we obtain

$$||w_n||_{C^{2,\alpha}_{\mu}(\Omega_n)} \leq c \left( ||f_n||_{C^{0,\alpha}_{\mu}(\Omega_n)} + ||(1-r^2)^{-\mu}w_n||_{C^0(\Omega_n)} \right) = c \left(1 + A_n\right).$$

Hence,

$$|\nabla \widetilde{w}_{n}| \leq c \, (1-r^{2})^{\mu} \frac{1+\|w_{n}\|_{\mathcal{C}^{0,\alpha}_{\mu}}(\Omega_{n})}{\|w_{n}\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_{n})}} \leq c \, (1-r^{2})^{\mu}.$$

The sets  $I_n$  converge to a nonempty interval  $I_{\infty}$ . Since the sequences  $(\tilde{w}_n)_n$  and  $(\nabla \tilde{w}_n)_n$  are uniformly bounded, Ascoli-Arzelà theorem assures that a subsequence of  $(\tilde{w}_n)_n$  converges on compact sets of  $S^1 \times I_{\infty}$  to a function  $w_{\infty}$  that vanishes on  $S^1 \times \partial I_{\infty}$ .

on compact sets of  $S^1 \times I_{\infty}$  to a function  $w_{\infty}$  that vanishes on  $S^1 \times \partial I_{\infty}$ . We note that  $\sup_{S^1 \times I_n} (1 - r^2)^{-\mu} |\widetilde{w}_n| = \frac{A_n}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_n)}}$ , which does not converge to zero. In fact  $A_n \to \infty$ ,  $\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_n)} \leq \|w_n\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Omega_n)} \leq c'(1 + A_n)$  then

$$\frac{A_n}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Omega_n)}} \ge \frac{A_n}{c'(1+A_n)} \to \frac{1}{c'} > 0.$$

In particular, it holds

$$0 < \sup_{S^1 \times I_{\infty}} (1 - r^2)^{-\mu} |w_{\infty}| \le 1.$$
(3.8)

In the same way it's possible to prove that a subsequence of  $(\tilde{f}_n)_n$  converges on compact sets of  $S^1 \times I_\infty$  to the function  $f_\infty \equiv 0$  since, if  $n \to \infty$ ,

$$\sup_{S^1 \times I_n} (1 - r^2)^{-\mu} |\tilde{f}_n| = \frac{1}{A_n} \to 0.$$

Then the limit function  $w_{\infty}$  must satisfy the differential equation

$$\Delta_{eucl} w_{\infty} = 0$$

on  $S^1 \times I_{\infty}$  with null boundary conditions on  $\partial I_{\infty}$ . Furthermore  $r = 1 \in I_{\infty}$ , then  $w_{\infty}(\theta, 1) = 0$ . So, also in the case  $\partial I_{\infty} = \emptyset$ , we can conclude that  $w_{\infty}(\theta, r) = 0$ . This function does not satisfy (3.8), a contradiction. This proves the assertion 31.

Thanks to Schauder estimates, we know that the function introduced in the assertion 31 is uniformly bounded and likewise its gradient. Hence Ascoli-Arzelà theorem assure us that we can take the limit as  $r_1 \rightarrow 1$  in a sequence of solutions which are defined on  $S^1 \times [r_0, r_1]$ . This proves the existence of a solution of  $\Delta_{eucl} w = f$  defined on  $S^1 \times [r_0, 1)$  for which it holds

$$\|w\|_{\mathcal{C}^{0,\alpha}_{\mu}(S^{1}\times[r_{0},1])} \leqslant c \,\|f\|_{\mathcal{C}^{0,\alpha}_{\mu}(S^{1}\times[r_{0},1])}.$$

Now it is sufficient to use again Schauder estimates to obtain the estimates for the derivatives.

## 3.4 A family of minimal surfaces close to the hyperbolic plane

In this section we will show the existence of normal minimal graphs over the plane,  $C_m$ , of equation  $x_3 = 0$  which are asymptotic to it. We will reformulate the problem to use Schauder fixed point theorem. We recall that  $r_{\epsilon} = \epsilon^{1/2}/2$ . We know already that the graph of the function v, denoted with  $\Sigma_v$ , is minimal, if and only if the function v is a solution of

$$F\left(\mathcal{L}v + Q_0\left(\sqrt{F}\nabla v, \sqrt{F}\nabla^2 v\right)\right) = 0.$$
(3.9)

This equation is a simplified version (since u = 0) of (3.3) introduced in section 3.2. The operator  $Q_0$  has bounded coefficients for  $r \in [r_{\epsilon}, 1)$ . Its expression is div  $(\nabla v Q_{0,1})$  where  $Q_{u,1}$  is given by (3.5).

Now let's consider a function  $\varphi \in \mathcal{C}^{2,\alpha}(S^1)$  which is even with respect to  $\theta$ ,  $L^2$ -orthogonal to  $e_0$  and  $e_1$  and such that  $\|\varphi\|_{\mathcal{C}^{2,\alpha}} \leq k\epsilon$ . We define

$$w_{\varphi}(\cdot, \cdot) := \mathcal{H}_{r_{\epsilon}, \varphi}(\cdot, \cdot),$$

where  $\mathcal{H}$  is the operator of harmonic extension introduced in proposition 46. In order to solve the equation (3.9), we choose  $\mu \in (0, 1)$  and look for v of the form  $v = w_{\varphi} + w$  where

 $w \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  and  $v = \varphi$  on  $S^1 \times \{r_{\epsilon}\}$ . Using proposition 30, we can rephrase this problem as a fixed point problem

$$w = S(\varphi, w) \tag{3.10}$$

where the nonlinear mapping S which depends on  $\epsilon$  and  $\varphi$  is defined by

$$S(\varphi, w) := -G_{r_{\epsilon}} \left( \mathcal{L} w_{\varphi} + Q_0 \left( w_{\varphi} + w \right) \right),$$

where the operator G is defined in proposition 30. To prove the existence of a fixed point for (3.10) we need the following result that states that S is a contracting mapping:

**Lemma 32.** There exist some constants  $c_k > 0$  and  $\epsilon_k > 0$ , such that

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^2 \tag{3.11}$$

and, for all  $\epsilon \in (0, \epsilon_k)$ 

$$\|S(\varphi, v_2) - S(\varphi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq \frac{1}{2} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu}}$$

for all  $v_1, v_2 \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  such that  $||v_i||_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant 2c_k \epsilon^2$ .

**Proof.** We know from proposition 30 that  $\|G_{r_{\epsilon}}(f)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}$ , then

$$|S(\varphi, 0)||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c ||\mathcal{L} w_{\varphi} + Q_0(w_{\varphi})||_{\mathcal{C}^{0,\alpha}_{\mu}} \leq c \left( ||\mathcal{L} w_{\varphi}||_{\mathcal{C}^{0,\alpha}_{\mu}} + ||Q_0(w_{\varphi})||_{\mathcal{C}^{0,\alpha}_{\mu}} \right).$$

So we need find the estimates for the two summands above. We recall that  $\|\varphi\|_{2,\alpha} \leq k\epsilon$ . Since  $\mu \in (0, 1)$  and thanks to proposition 46 we obtain

$$\|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c \|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{1}} \leqslant c \|\varphi\|_{\mathcal{C}^{2,\alpha}} \leqslant ck\epsilon.$$

Being  $w_{\varphi}$  a harmonic function we have  $\mathcal{L} w_{\varphi} = 0$ . The last term is estimated by

$$\|Q_0\left(w_{\varphi}\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c_k \epsilon^2$$

In fact

$$\|Q_0(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c \|w\|_{2,\alpha,\mu}^2 \leqslant c \|\varphi\|_{2,\alpha}^2 \leqslant c_k \epsilon^2.$$

Then we can conclude

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^2.$$

As for the second estimate, we consider

$$S(\varphi, v_2) - S(\varphi, v_1) = G_{r_{\epsilon}} \left( Q_0 \left( w_{\varphi} + v_2 \right) - Q_0 \left( w_{\varphi} + v_1 \right) \right)$$

and consequently

$$\|S(\varphi, v_2) - S(\varphi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \|Q_0(w_{\varphi} + v_2) - Q_0(w_{\varphi} + v_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}}$$

We observe that from the considerations made above it follows that

$$\begin{aligned} ||Q_0 (w_{\varphi} + v_2) - Q_0 (w_{\varphi} + v_1)||_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c_1 ||v_2 - v_1||_{\mathcal{C}^{2,\alpha}_{\mu}} ||w_{\varphi}||_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon ||v_2 - v_1||_{\mathcal{C}^{2,\alpha}_{\mu}}. \end{aligned}$$

Then

$$\|S(\varphi, v_2) - S(\varphi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon ||v_2 - v_1||_{\mathcal{C}^{2,\alpha}_{\mu}}.$$

**Theorem 33.** Let be  $B := \{ w \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1)) \mid ||w||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq 2c_k \epsilon^2 \}$ . Then the nonlinear mapping S defined above has a unique fixed point v in B.

**Proof:** The previous lemma shows that, if  $\epsilon$  is chosen small enough, the nonlinear mapping S is a contraction mapping from the ball B of radius  $2c_k\epsilon^2$  in  $\mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  into itself. This value follows from the estimate of the norm of S(0). Consequently thanks to Schauder fixed point theorem, S has an unique fixed point w in this ball.  $\Box$ 

We have proved the existence of a minimal surface, denoted with  $S_m$ , which is close to  $\mathbb{H}^2 \times \{0\}$ , and close to its boundary is the vertical graph over the annulus  $B_{2r_{\epsilon}} - B_{r_{\epsilon}}$  of a function which can be expanded as

$$\mathcal{H}_{r_{\epsilon},\varphi}(r,\theta) + \bar{V}_m(r,\theta), \quad \text{with} \quad ||\bar{V}_m||_{\mathcal{C}^{2,\alpha}_u} \leqslant c\epsilon.$$

The function  $V_m$  depends non linearly on  $\epsilon, \varphi$ . Furthermore it satisfies

$$\|\bar{V}_m(\epsilon,\varphi)(r_\epsilon\cdot) - \bar{V}_m(\epsilon,\varphi')(r_\epsilon\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leqslant c\epsilon \|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}(S^1)}.$$

## 3.5 The catenoid in $\mathbb{H}^2 \times \mathbb{R}$

The catenoid in the space  $\mathbb{H}^2 \times \mathbb{R}$  can be obtained by the revolution around the  $x_3$  axis, {0,0}  $\times \mathbb{R}$ , of an appropriate curve  $\gamma$  (see [32]). We consider a vertical geodesic plane containing the origin of  $\mathbb{H}^2$  and the curve  $\gamma$ . Let r be the euclidean distance between the point of  $\gamma$  at height t and the  $x_3$  axis: we denote with r = r(t) a parametrization of  $\gamma$ . The surface obtained by revolution of  $\gamma$ , is minimal if and only if r = r(t) satisfies the following differential equation:

$$4r(t)\frac{\partial^2 r}{\partial t^2} - 4\left(\frac{\partial r}{\partial t}\right)^2 - (1 - r(t)^4) = 0.$$
(3.12)

A first integral for this equation is:

$$\left(\frac{\partial r}{\partial t}\right)^2 = Cr^2 - \frac{1+r^4}{4} \tag{3.13}$$

with  $C > \frac{1}{2}$ . It is possible to prove that the function r(t) has a minimum value  $r_{min}$  given by:

$$r_{min} = \sqrt{\frac{2C+1}{2}} - \sqrt{\frac{2C-1}{2}} < 1.$$

Since we assume  $C = \frac{1}{4\epsilon^2}$ , we get

$$r_{min} = \sqrt{\frac{2C+1}{2}} - \sqrt{\frac{2C-1}{2}} = \sqrt{C} \left( 1 + \frac{1}{4C} - 1 + \frac{1}{4C} + \mathcal{O}(\frac{1}{C^2}) \right) = \frac{1}{2\sqrt{C}} + O\left(\frac{1}{C^{3/2}}\right) = \epsilon + \mathcal{O}(\epsilon^3).$$

We denote with  $C_t$  and  $C_b$ , respectively, the part of the catenoid contained in  $\mathbb{H}^2 \times \mathbb{R}^+$ and  $\mathbb{H}^2 \times \mathbb{R}^-$ .

We set

$$t_{\epsilon} = -\epsilon \ln \epsilon$$

We need find the parametrization of  $C_t$  and  $C_b$  as graphs on the horizontal plane respectively for  $t \in [t_{\epsilon} - \epsilon, t_{\epsilon} + \epsilon]$  and  $t \in [-t_{\epsilon} - \epsilon, -t_{\epsilon} + \epsilon]$ .

**Lemma 34.** For  $\epsilon > 0$  small enough, we have

$$r_{\epsilon}(t) = \epsilon \cosh \frac{t}{2\epsilon} + \mathcal{O}(\epsilon^3 e^{\frac{t}{2\epsilon}}) \text{ and } \partial_t r_{\epsilon}(t) = \frac{1}{2} \sinh \frac{t}{2\epsilon} + \mathcal{O}(\epsilon^2 e^{\frac{t}{2\epsilon}})$$

for  $t \in [0, t_{\epsilon} + \epsilon]$ . Moreover if  $t \in [t_{\epsilon} - \epsilon, t_{\epsilon} + \epsilon]$ , we derive

$$r_{\epsilon}(t) = \mathcal{O}(\epsilon^{1/2}) \text{ and } \partial_t r_{\epsilon} = \mathcal{O}(\epsilon^{-1/2})$$

**Proof.** We define the function v(t) such that  $r_{\epsilon}(t) = r_{\epsilon}(0) \cosh v(t)$ , with v(0) = 0 and  $r_{\epsilon}(0)$  which satisfies

$$Cr_{\epsilon}^{2}(0) - \frac{1 + r_{\epsilon}^{4}(0)}{4} = 0,$$

from which

$$\frac{1}{4} = Cr_{\epsilon}^2(0) - \frac{r_{\epsilon}^4(0)}{4}.$$
(3.14)

Plugging  $r_{\epsilon}(t)$  in (3.13) and using (3.14), we have

$$(\partial_t v)^2 = C - \frac{r_{\epsilon}^2(0)}{4} (1 + \cosh^2 v(t))$$

and under the hypothesis that  $v(t) = g(\frac{t}{2\epsilon})$  and

$$\frac{t}{2\epsilon} \leqslant g\left(\frac{t}{2\epsilon}\right) \leqslant \frac{t}{2\epsilon} + 1$$

we obtain that  $(\partial_t v)^2 = C + \mathcal{O}(\epsilon^2 e^{\frac{t}{2\epsilon}})$  and then  $v(t) = \sqrt{Ct} + \mathcal{O}(\epsilon^3 e^{\frac{t}{2\epsilon}})$ . We remark a posteriori that  $\frac{t}{2\epsilon} \leq v(t) \leq \frac{t}{2\epsilon} + 1$  holds for  $t \in [0, t_{\epsilon} + \epsilon], \epsilon > 0$  small enough and then

$$r_{\epsilon}(t) = r_{\epsilon}(0)\cosh v(t) = \epsilon \cosh\left(\frac{t}{2\epsilon}\right) + \mathcal{O}(\epsilon^{3}e^{\frac{t}{2\epsilon}})$$
(3.15)

where we use that  $r_{\epsilon}(0) = r_{min} = \epsilon + \mathcal{O}(\epsilon^3)$ . Now we assume that  $t \in [t_{\epsilon} - \epsilon, t_{\epsilon} + \epsilon]$ , then  $r_{\epsilon}(t) = \mathcal{O}(\epsilon^{1/2})$  and  $\partial_t r_{\epsilon}(t) = \frac{1}{2} \sinh\left(\frac{t}{2\epsilon}\right) + \mathcal{O}(\epsilon^2 e^{\frac{t}{2\epsilon}}) = \mathcal{O}(\epsilon^{-1/2})$ .

Now we can prove a lemma that give us the parametrization of the part of catenoid whose height t belongs to a neighbourhood of  $t_{\epsilon}$ .

**Lemma 35.** For  $\epsilon > 0$ , small enough and  $t \in [t_{\epsilon} - \epsilon, t_{\epsilon} + \epsilon]$ , the catenoid can be parametrized on an annulus of  $C_{\epsilon} = \left\{ re^{i\theta}; \frac{r_{\epsilon}}{\sqrt{e}} \leqslant r \leqslant r_{\epsilon}\sqrt{e} \right\}$  by the graph of the function  $U_t(r, \theta)$  which satisfies

$$U_t(r,\theta) = 2\epsilon \ln \frac{2r}{\epsilon} + v_t(r,\theta), \qquad (3.16)$$
$$\partial_r U_t(r,\theta) = \frac{2\epsilon}{r} + v_{t,d}(r,\theta).$$

with  $v_t(r,\theta) = \mathcal{O}(\epsilon^2)$  and  $v_{t,d}(r,\theta) = \mathcal{O}(\epsilon^{5/2})$ .

**Proof:** The results follow easily from the hypothesis and the equation (3.15).

It's easy to understand that the parametrization of the catenoid for values of t in a neighbourhood of  $-t_{\epsilon}$  is obtained by a change of the sign in the expression 3.16. Similar consideration are true for the derivative. Then we obtain

$$U_b(r,\theta) = -2\epsilon \ln \frac{2r}{\epsilon} + v_b(r,\theta),$$
$$\partial_r U_b(r,\theta) = -\frac{2\epsilon}{r} + v_{b,d}(r,\theta)$$

with  $v_b(r,\theta) = \mathcal{O}(\epsilon^2)$  and  $v_{b,d}(r,\theta) = \mathcal{O}(\epsilon^{5/2})$ .

## 3.6 A family of minimal surfaces close to a catenoid on $S^1 \times [r_{\epsilon}, 1)$

In this section we want to show the existence of minimal normal graphs over the catenoid defined in  $S^1 \times [r_{\epsilon}, 1) \subset \mathbb{H}^2$  and asymptotic to it. We know that the graph of the function u + v is minimal, being u the function whose graph is the catenoid, if and only if v is a solution of the equation

$$H_{u+v} = 0$$
 (3.17)

whose expression is given by (3.3). The explicit expression of  $L_u v$  is

$$F\left(\frac{1}{\sqrt{A}}\Delta v + \frac{\partial}{\partial r}\left(\frac{1}{\sqrt{A}}\right)\frac{\partial v}{\partial r} - \frac{1}{A^{\frac{3}{2}}}\frac{\partial u}{\partial r}\frac{\partial}{\partial r}\left(F\frac{\partial u}{\partial r}\right)\frac{\partial v}{\partial r} - F\frac{\partial u}{\partial r}\frac{\partial}{\partial r}\left(\frac{1}{A^{\frac{3}{2}}}\frac{\partial u}{\partial r}\frac{\partial v}{\partial r}\right)\right), \quad (3.18)$$

where  $F = \frac{1}{4}(1 - r^2)^2$ ,

$$A = 1 + F|\nabla u|^2 = \frac{(4C-2)r^2}{4Cr^2 - 1 - r^4}$$

and

$$\frac{\partial u}{\partial r} = \pm \frac{2}{\sqrt{4Cr^2 - 1 - r^4}},$$

as it is easy to obtain using (3.13). It's useful to observe that since we assume  $C = \frac{1}{4\epsilon^2}$ and  $r_{\epsilon} = \epsilon^{\frac{1}{2}}/2$ , we have that, for  $r \in [r_{\epsilon}, 1)$ ,  $A = 1 + \mathcal{O}(\epsilon)$ ,  $\frac{\partial u}{\partial r} = \mathcal{O}(\sqrt{\epsilon})$ ,

$$\frac{\partial A}{\partial r} = \frac{(8C-4)(-r+r^5)}{(4Cr^2-1-r^4)^2} = \mathcal{O}(\sqrt{\epsilon})$$

and

$$\frac{\partial^2 u}{\partial r^2} = \frac{(8Cr - 4r^3)}{\sqrt{(4Cr^2 - 1 - r^4)^3}} = \mathcal{O}(1).$$

Taking into account these estimates, we can conclude that it holds

$$\bar{L}_{u}v := \sqrt{A} \left( \frac{\partial}{\partial r} \left( \frac{1}{\sqrt{A}} \right) \frac{\partial v}{\partial r} - \frac{1}{A^{\frac{3}{2}}} \frac{\partial u}{\partial r} \frac{\partial}{\partial r} \left( F \frac{\partial u}{\partial r} \right) \frac{\partial v}{\partial r} - F \frac{\partial u}{\partial r} \frac{\partial}{\partial r} \left( \frac{1}{A^{\frac{3}{2}}} \frac{\partial u}{\partial r} \frac{\partial v}{\partial r} \right) \right) = \mathcal{O}(\sqrt{\epsilon}) \frac{\partial v}{\partial r} + \mathcal{O}(\sqrt{\epsilon}) \frac{\partial^{2} v}{\partial^{2} r}.$$
(3.19)

Then we can write  $\sqrt{A}L_u v = F\left(\mathcal{L}v + \bar{L}_u v\right)$ .

We remark that we have already studied the mapping properties of the operator  $\mathcal{L}$  in section 3.3.

Then the graph of a function v over  $\Sigma_u$  is minimal if and only if v is a solution of the following equation

$$\mathcal{L}v + \bar{L}_u v + \sqrt{A}Q_u(v) = 0. \tag{3.20}$$

Thanks to the observations on the functions A and  $u_r$ , we can conclude that  $Q_u$  has bounded coefficients in  $[r_{\epsilon}, 1)$ . Now we consider a function  $\varphi \in \mathcal{C}^{2,\alpha}(S^1)$  which is even with respect to  $\theta$ ,  $L^2$ -orthogonal to  $e_0$  and  $e_1$  and such that  $\|\varphi\|_{\mathcal{C}^{2,\alpha}} \leq k\epsilon$ . We define

$$w_{\varphi}(\cdot, \cdot) := \mathcal{H}_{r_{\epsilon}, \varphi}(\cdot, \cdot)$$

where the operator  $\mathcal{H}$  has been introduced in proposition 46. In order to solve the equation (3.20), we choose  $\mu \in (0, 1)$  and look for v of the form  $v = w_{\varphi} + w$  where  $w \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  and  $v = \varphi$  on  $S^1 \times \{r_{\epsilon}\}$ . We can rephrase this problem as a fixed point problem, that is

$$w = S(\varphi, w) \tag{3.21}$$

where the nonlinear mapping S is defined by

$$S(\varphi, w) := -G_{r_{\epsilon}} \left( \mathcal{L}w_{\varphi} + \bar{L}_{u}(w_{\varphi} + w) + \sqrt{A}Q_{u}(w_{\varphi} + w) \right)$$

where the operator G is defined in proposition 30. To prove the existence of a solution for (3.21) we need the following result which states that S is a contracting mapping.

**Lemma 36.** There exist some constants  $c_k > 0$  and  $\epsilon_k > 0$ , such that

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^{3/2} \tag{3.22}$$

and, for all  $\epsilon \in (0, \epsilon_k)$ 

$$\|S(\varphi, w_2) - S(\varphi, w_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq \frac{1}{2} \|w_2 - w_1\|_{\mathcal{C}^{2,\alpha}_{\mu}}$$

for all  $w_1, w_2 \in \mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  such that  $||w_i||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq 2c_k \epsilon^{\frac{3}{2}}$ .

**Proof.** We know from the proposition 30 that  $\|G_{r_{\epsilon}}(f)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}$ . Then

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \|\mathcal{L}w_{\varphi} + \bar{L}_{u}(w_{\varphi}) + \sqrt{AQ_{u}(w_{\varphi})}\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leq c \left(\|\mathcal{L}w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}} + \|\bar{L}_{u}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}} + \|Q_{u}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}}\right).$$

Here we have used the fact that  $A = 1 + \mathcal{O}(\epsilon)$ .

So we need to find the estimates of each summand. We recall that  $\|\varphi\|_{\mathcal{C}^{2,\alpha}} \leq k\epsilon$ . For all  $\mu \in (0,1)$ ,  $\|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq \|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{1}}$ , and thanks to proposition 46 we get that

$$\|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c \|\varphi\|_{\mathcal{C}^{2,\alpha}(S^{1})} \leqslant ck\epsilon$$

and  $\mathcal{L} w_{\varphi} = 0$ . We use (3.19) for finding the estimate of  $\bar{L}_u w_{\varphi}$ . We obtain

$$\|\bar{L}_u(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c\epsilon^{1/2} \|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^{3/2}.$$

The last term is estimated observing that

$$\|Q_u(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c \|w_{\varphi}\|^2_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^2.$$

Putting together all these estimates we get

$$\|S(\varphi, w_{\varphi})\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c_k \epsilon^{3/2}.$$

As for the second estimate, we observe that

$$S(\varphi, w_2) - S(\varphi, w_1) = -G_{r_{\epsilon}} \left( \bar{L}_u(w_{\varphi} + w_2) + \sqrt{A}Q_u(w_{\varphi} + w_2) \right) + G_{r_{\epsilon}} \left( \bar{L}_u(w_{\varphi} + w_1) + \sqrt{A}Q_u(w_{\varphi} + w_1) \right)$$

and

$$\|S(\varphi, w_2) - S(\varphi, w_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c\|\bar{L}_u(w_{\varphi} + w_2) - \bar{L}_u(w_{\varphi} + w_1) + Q_u(w_{\varphi} + w_2) - Q_u(w_{\varphi} + w_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}} = c\|\bar{L}_u(w_2 - w_1) + Q_u(w_{\varphi} + w_2) - Q_u(w_{\varphi} + w_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leq \leq \|\bar{L}_u(w_2 - w_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}} + \|Q_u(w_{\varphi} + w_1) - Q_u(w_{\varphi} + w_2)\|_{\mathcal{C}^{0,\alpha}_{\mu}}.$$

We observe that from the considerations above it follows that

$$\|\bar{L}_u(w_2 - w_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c\epsilon^{1/2} ||w_2 - w_1||_{\mathcal{C}^{2,\alpha}_{\mu}}$$

and

$$\begin{aligned} ||Q_u (w_{\varphi} + w_1) - Q_u (w_{\varphi} + w_2)||_{\mathcal{C}^{0,\alpha}_{\mu}} &\leq c ||w_2 - w_1||_{\mathcal{C}^{2,\alpha}_{\mu}} ||w_{\varphi}||_{\mathcal{C}^{2,\alpha}_{\mu}} \\ &\leq c_k \epsilon ||w_2 - w_1||_{\mathcal{C}^{2,\alpha}_{\mu}}. \end{aligned}$$

Then

$$||S(\varphi, w_2) - S(\varphi, w_1)||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c\epsilon^{1/2} ||w_2 - w_1||_{\mathcal{C}^{2,\alpha}_{\mu}}.$$

**Theorem 37.** Let be  $B := \{ w \in C^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1)) \mid ||w||_{2,\alpha,\mu} \leq 2c_k \epsilon^{3/2} \}$ . Then the nonlinear mapping S defined above has a unique fixed point v in B.

**Proof:** The previous lemma shows that, if  $\epsilon$  is chosen small enough, the nonlinear mapping S is a contraction mapping from the ball B of radius  $2c_k\epsilon^{3/2}$  in  $\mathcal{C}^{2,\alpha}_{\mu}(S^1 \times [r_{\epsilon}, 1))$  into itself. This value follows from the estimate of the norm of  $S(\varphi, 0)$ . Consequently thanks to Schauder fixed point theorem, S has a unique fixed point w in this ball.  $\Box$ 

We have proved the existence of a minimal surface  $S_t$ , which is close to the part of catenoid  $C_t$  introduced in section 3.5 and close to its boundary is a graph over the annulus  $B_{2r_e} - B_{r_e}$  of the function which can be expanded as

$$2\epsilon \ln\left(\frac{2r}{\epsilon}\right) + \mathcal{H}_{r_{\epsilon},\varphi}(r,\theta) + \bar{V}_t(r,\theta),$$

with  $\|\bar{V}_t\|_{\mathcal{C}^{2,\alpha}_u} \leq c\epsilon^{3/2}$ .

Analogously we can show the existence of a minimal surface  $S_b$ , which is close to the part of catenoid denoted by  $C_b$  introduced in section 3.5 and close to its boundary is a graph over the annulus  $B_{2r_{\epsilon}} - B_{r_{\epsilon}}$  of the function

$$-2\epsilon \ln\left(\frac{2r}{\epsilon}\right) - \mathcal{H}_{r_{\epsilon},\varphi}(r,\theta) - \bar{V}_{b}(r,\theta),$$

with  $\|\bar{V}_b\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c\epsilon^{3/2}$ . The functions  $\bar{V}_t, \bar{V}_b$  depend non linearly on  $\epsilon, \varphi$ . Furthermore they satisfy

$$\|\bar{V}(\epsilon,\varphi)(r_{\epsilon}\cdot)-\bar{V}(\epsilon,\varphi')(r_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_{1}-B_{1/2})} \leqslant c\epsilon^{\frac{1}{2}}\|\varphi-\varphi'\|_{\mathcal{C}^{2,\alpha}(S^{1})}.$$

## 3.7 The mean curvature in a neighbourhood of $\{0, 0\} \times \mathbb{R}$ in $\mathbb{H}^2 \times \mathbb{R}$

We recall that the model we use for  $\mathbb{H}^2 \times \mathbb{R}$  the disk model endowed with the metric  $g_{hyp}$ . If  $x_1, x_2$  denote the coordinates in  $\mathbb{H}^2$  and  $x_3$  the coordinate in  $\mathbb{R}$ , then

$$g_{hyp} = \frac{dx_1^2 + dx_2^2}{F} + dx_3^2,$$

where

$$F = \frac{1}{4} \left( 1 - x_1^2 - x_2^2 \right)^2 = \frac{1}{4} \left( 1 - r^2 \right)^2.$$

In this section we want to express the mean curvature  $H_{hyp}$  of a surface, contained in a cylindrical neighbourhood C of  $\{0,0\} \times \mathbb{R}$  in  $\mathbb{H}^2 \times \mathbb{R}$  in terms of the mean curvature  $H_e$  computed with respect to the euclidean metric  $g_{eucl} = dx_1^2 + dx_2^2 + dx_3^2$ .

Let  $\bar{g}$  be the metric defined on  $\mathbb{R}^3$  by  $4dx_1^2 + 4dx_2^2 + dx_3^2$ . We consider the map f:  $(\mathbb{R}^3, g_{eucl}) \to (\mathbb{R}^3, \bar{g})$  defined by

$$(x_1, x_2, x_3) \to \left(\frac{x_1}{2}, \frac{x_2}{2}, x_3\right).$$
 (3.23)

It is easy to see that it is an isometric embedding. That is the pull-back of the metric  $\bar{g}$  by f equals  $g_{eucl}$ .

We denote by  $\overline{\nabla}$  the riemannian connection on  $(\mathbb{R}^3, \overline{g})$ . Let X, Y be two vector fields on  $(\mathbb{R}^3, g_{eucl})$ . We define  $\overline{X} := f_*(X), \overline{Y} := f_*(Y)$ , Then the connection defined on  $(\mathbb{R}^3, g_{eucl})$ . by

$$\nabla_X Y = \bar{\nabla}_{\bar{X}} \bar{Y},\tag{3.24}$$

coincides with the riemannian connection relative to the metric  $g_{eucl}$ . If  $N_e$  denotes the normal vector to a surface  $\Sigma$  with respect to the euclidean metric, the mean curvature  $H_e(\Sigma)$  of  $\Sigma$  in  $(\mathbb{R}^3, g_{eucl})$  is defined to be half the trace of the mapping

$$X \to -[\nabla_X N_{eucl}]^T,$$

where  $[\cdot]^T$  denotes the projection on  $T\Sigma$ . We set  $\bar{\Sigma} = f(\Sigma)$ . Similarly if  $\bar{N}$  denotes the normal vector to  $\bar{\Sigma}$  with respect the metric  $\bar{g}$ , the mean curvature  $\bar{H}(\bar{\Sigma})$  of  $\bar{\Sigma}$  in  $(\mathbb{R}^3, \bar{g})$  is defined to be half the trace of the mapping

$$X \to -[\bar{\nabla}_X \bar{N}]^T,$$

where  $[\cdot]^T$  denotes the projection on  $T\overline{\Sigma}$ . Thanks to the relation (3.24) and from the fact that  $T\overline{\Sigma} = f_*T\Sigma$  it is clear that

$$\bar{H}(\bar{\Sigma}) = H_e(\Sigma). \tag{3.25}$$

Now we turn our attention toward the space  $\mathbb{H}^2 \times \mathbb{R}$ . Now our aim is to find a relation between the mean curvature with respect to the metric  $g_{hyp}$  of a surface contained in a cylindrical neighbourhood C of  $\{0,0\} \times \mathbb{R}$  in  $\mathbb{H}^2 \times \mathbb{R}$  and the mean curvature of the same surface seen in the riemannian manifold  $(C, \bar{g})$ .

The Christoffel symbols,  $\Gamma_{ij}^k$ , associated to  $g_{hyp}$  all vanish except

$$\Gamma_{11}^{1} = \Gamma_{21}^{2} = \Gamma_{12}^{2} = -\Gamma_{22}^{1} = \frac{2x^{1}}{1 - r^{2}},$$
  
$$\Gamma_{12}^{1} = \Gamma_{22}^{2} = \Gamma_{21}^{1} = -\Gamma_{11}^{2} = \frac{2x^{2}}{1 - r^{2}}.$$

Let  $\partial_1 = \frac{1}{2} \frac{\partial}{\partial x_1}$ ,  $\partial_2 = \frac{1}{2} \frac{\partial}{\partial x_2}$ ,  $\partial_1 = \frac{\partial}{\partial x_3}$  be the elements of a basis of the tangent space. Now, if  $X = \sum_i X^i \partial_i$  and  $Y = \sum_j Y^j \partial_j$  are two tangent vector fields, the expression of the covariant derivative in  $(\mathbb{H}^2 \times \mathbb{R}, g_{hyp})$  is given by

$$\nabla^h_X Y = \sum_k \left( \sum_{i,j} X^i Y^j \Gamma^k_{ij} + X(Y^k) \right) \partial_k.$$

It is clear that

$$\nabla^h_X Y = \bar{\nabla}_X Y + \sum_{k=1}^2 \sum_{i,j} X^i Y^j \Gamma^k_{ij} \partial_k.$$
(3.26)

It is possible to show that  $\sum_{k=1}^{2} \sum_{i,j} X^{i} Y^{j} \Gamma_{ij}^{k} \partial_{k}$  is the vector whose components with respect the basis  $(\partial_{1}, \partial_{2}, \partial_{3})$  are given by

$$\begin{bmatrix} \frac{Y^{1}x_{1}}{\sqrt{F}} + \frac{Y^{2}x_{2}}{\sqrt{F}} & \frac{Y^{1}x_{2}}{\sqrt{F}} - \frac{Y^{2}x_{1}}{\sqrt{F}} & 0\\ -\frac{Y^{1}x_{2}}{\sqrt{F}} + \frac{Y^{2}x_{1}}{\sqrt{F}} & \frac{Y^{1}x_{1}}{\sqrt{F}} + \frac{Y^{2}x_{2}}{\sqrt{F}} & 0\\ 0 & 0 & 0 \end{bmatrix} \begin{bmatrix} X_{1} \\ X_{2} \\ X_{3} \end{bmatrix}$$

The mean curvature  $H_{hyp}(S)$  of a surface S in  $(\mathbb{H}^2 \times \mathbb{R}, g_{hyp})$  is defined to be half the trace of the mapping

$$X \to -[\nabla_X N_{hyp}]^T,$$

where  $[\cdot]^T$  denotes the projection onto the tangent bundle of the surface and  $N_{hyp} = (N_1, N_2, N_3)$  the normal vector to S. Thanks to (3.26) we get the relation

$$[\nabla^h_X N_{hyp}]^T = [\bar{\nabla}_X N_{hyp}]^T + \sum_{k=1}^2 \sum_{i,j} X^i N^j \Gamma^k_{ij} \partial_k.$$
(3.27)

Now we observe that the normal vector  $\bar{N}$  to S with respect the metric  $\bar{g}$  does not coincide with  $N_{hyp}$ . Since  $g_{hyp}$  is asymptotic to the metric  $\bar{g}$ , it is reasonable to think that  $N_{hyp}(S)$ can be related to  $\bar{N}(S)$ . If we suppose that  $N_{hyp}(S) = \frac{(v_1, v_2, v_3)}{|(v_1, v_2, v_3)|g_{hyp}|}$  and  $\bar{N}(S) = \frac{(v_1, v_2, v_3)}{|(v_1, v_2, v_3)|g}$ then it is possible to show that

$$N_{hyp}(S) = 2\sqrt{F}\bar{N}(S)\left(1 + \mathcal{O}(r^2)\right).$$

Now we insert this result into (3.27) and we compute the opposite of half the trace. We obtain

$$H_{hyp}(S) = (\bar{H}(S) - 2(x_1N_1 + x_2N_2))(1 + \mathcal{O}(r^2)).$$
(3.28)

Thanks to (3.25) we know that  $\bar{H}(S) = H_e(f^{-1}(S))$ . We have proved the following result

**Proposition 38.** Let S be a surface contained in a sufficiently small cylindrical neighbourhood C of  $\{0,0\} \times \mathbb{R} \subset \mathbb{H}^2 \times \mathbb{R}$  endowed with the metric  $g_{hyp}$ . If  $N = \sum_{i=1}^{3} N_i \partial_i$  and  $H_{hyp}(\cdot)$  denote respectively the normal vector to S and the mean curvature with respect to  $g_{hyp}$ ,  $H_e(\cdot)$  the mean curvature with respect to  $g_e$  and f is the map defined by (3.23), then

$$H_{hyp}(S) = (H_e(f^{-1}(S)) - 2(x_1N_1 + x_2N_2))(1 + \mathcal{O}(r^2)).$$
(3.29)

### **3.8** A rescaled Costa-Hoffman-Meeks type surface

In this section we will describe the surface obtained by rescaling of the Costa-Hoffmann-Meeks surface of genus  $k \ge 1$ ,  $M_k$ , (see C. Costa [3], [4] and D. Hoffman and W. H. Meeks [15], [16]) and we will study the mapping properties of its Jacobi operator. We denote by  $M_{k,\epsilon}$  the image of  $M_k$  by an homothety of parameter  $2\epsilon$ . We will adapt to our situation some of the analytical tools used in [11] to show the existence of a family of minimal surfaces close to  $M_k$  with one planar end and two slightly bent catenoidal ends by an angle  $\xi \in (-\xi_0, \xi_0), \xi_0 > 0$  and small enough. We denote an element of this family by  $M_k(\xi)$ , then  $M_k(\xi)|_{\xi=0} = M_k$ .

#### 3.8.1 The Costa-Hoffman-Meeks surface.

We start by giving a brief description of the surface  $M_k$ . After suitable rotation and translation,  $M_k$  enjoys the following properties.

- 1. It has one planar end  $E_m$  asymptotic to the  $x_3 = 0$  plane, one top end  $E_t$  and one bottom end  $E_b$  that are respectively asymptotic to the upper end and to the lower end of a catenoid with  $x_3$ -axis of revolution. The planar end  $E_m$  is located between the two catenoidal ends.
- 2. It is invariant under the action of the rotation of angle  $\frac{2\pi}{k+1}$  about the  $x_3$ -axis, under the action of the symmetry with respect to the  $x_2 = 0$  plane and under the action of the composition of a rotation of angle  $\frac{\pi}{k+1}$  about the  $x_3$ -axis and the symmetry with respect to the  $x_3 = 0$  plane.
- 3. It intersects the  $x_3 = 0$  plane in k + 1 straight lines, which intersect themselves at the origin with angles equal to  $\frac{\pi}{k+1}$ . The intersection of  $M_k$  with the plane  $x_3 = const (\neq 0)$  is a single Jordan curve. The intersection of  $M_k$  with the upper half space  $x_3 > 0$  (resp. with the lower half space  $x_3 < 0$ ) is topologically an open annulus.

We denote with  $X_i$ , with i = t, b, m, the parametrization of the end  $E_i$  and with  $X_{i,\epsilon}$  the parametrization of the corresponding end  $E_{i,\epsilon}$  of  $M_{k,\epsilon}$ .

Now we give a local description of the surface  $M_{k,\epsilon}$  near its ends and we introduce coordinates that we will use.

The planar end. The planar end  $E_{m,\epsilon}$  of the surface  $M_{k,\epsilon}$  can be parametrized by

$$X_{m,\epsilon}(x) := \left(\frac{2\epsilon x}{|x|^2}, 2\epsilon u_m(x)\right) \in \mathbb{R}^3,$$
(3.30)

where  $x \in \overline{B}_{\rho_0}(0) - \{0\} \subset \mathbb{R}^2$  and the function  $u_m$  tends to 0 like  $u_m(x) = \mathcal{O}(|x|^{k+1})$ . Here  $\rho_0 > 0$  is fixed small enough. We will assume  $\rho_0 = \mathcal{O}(\epsilon^{\frac{1}{4}})$ . The minimal surface equation has the following form

$$\frac{|x|^4}{4\epsilon^2} \operatorname{div}\left(\frac{\nabla u}{(1+|x|^4\,|\nabla u|^2)^{1/2}}\right) = 0. \tag{3.31}$$

It can be shown (see [11]) that the function  $u_m$  can be extended at the origin continuously using Weierstrass representation. We can prove that  $u_m \in C^{2,\alpha}(\bar{B}_{\rho_0})$ .

If we linearize in u = 0 the nonlinear equation (3.31) we obtain the expression an operator which is close, up to a multiplication by  $4\epsilon^2$ , to the Jacobi operator about the plane, that is  $\mathcal{L}_{\mathbb{R}^2} = |x|^4 \Delta$ . To be more precise, the linearization of (3.31) gives

$$L_u = \frac{|x|^4}{4\epsilon^2} \operatorname{div} \left( \frac{\nabla v}{\sqrt{1+|x|^4|\nabla u|^2}} - |x|^4 \nabla u \frac{\nabla u \cdot \nabla v}{\sqrt{(1+|x|^4|\nabla u|^2)^3}} \right).$$

We will give the expression of  $H_{u+v}$ , the mean curvature of the graph of the function u+v, in terms of the mean curvature of  $\Sigma_u$ , that is  $H_u$ . In the following we shall restrict our attention to the planar case, that is u = 0, on a domain of the form  $\{(r, \theta) \in B_{r_0}(0) | r \in [r_1, r_2]\}$ . Here we shall show that

$$2H_{u+v} = 2H_u + L_u v + \frac{|x|^4}{4\epsilon^2} Q_u(\sqrt{|x|^4}\nabla v, \sqrt{|x|^4}\nabla^2 v), \qquad (3.32)$$

where  $Q_u$  satisfies

$$Q_u(0,0) = \nabla Q'_u(0,0).$$

To show (3.32), we start observing that:

$$\frac{1}{\sqrt{1+|x|^4|\nabla(u+v)|^2}} = \frac{1}{\sqrt{1+|x|^4|\nabla u|^2}} - |x|^4 \frac{\nabla u \cdot \nabla v}{\sqrt{(1+|x|^4|\nabla u|^2)^3}} + Q_{u,1}(v)$$
(3.33)

where the function  $Q_{u,1}$  satisfies  $Q_{u,1}(0) = \nabla Q_{u,1}(0) = 0$ . The proof of that is very close to the one that appears in section 3.2: it's necessary only to replace F by  $|x|^4$ . So we can omit some details. Secondly we observe that  $2H_{u+v}$  is given by

$$\frac{|x|^4}{4\epsilon^2} \operatorname{div}\left(\frac{\nabla(u+v)}{\sqrt{1+|x|^4|\nabla u|^2}} - |x|^4 \nabla(u+v) \frac{\nabla u \cdot \nabla v}{\sqrt{(1+|x|^4|\nabla u|^2)^3}} + \nabla(u+v)Q_{u,1}(v)\right).$$
From this it follows the wanted expression.

Since we assume that  $\Sigma_u$  is a minimal surface, we will consider  $H_u = 0$ .

Following what we have done in section 3.6 replacing F by  $|x|^4$  we get the expression of the minimal surfaces equation that we will use in the following sections:

$$\frac{|x|^4}{4\epsilon^2} \left( \mathcal{L}v + \bar{L}_u v + Q_u(\sqrt{|x|^4}\nabla v, \sqrt{|x|^4}\nabla^2 v) \right) = 0.$$
(3.34)

The catenoidal ends. We denote by  $X_c$  the parametrization of the standard catenoid C whose axis of revolution is the  $x_3$ -axis. Its expression is

$$X_c(s,\theta) := (\cosh s \, \cos \theta, \cosh s \, \sin \theta, s) \in \mathbb{R}^3$$

where  $(s, \theta) \in \mathbb{R} \times S^1$ . The unit normal vector field about C is given by

$$n_c(s,\theta) := \frac{1}{\cosh s} (\cos \theta, \sin \theta, -\sinh s).$$

The catenoid C may be divided in two pieces, denoted  $C_{\pm}$ , which are defined as the image by  $X_c$  of  $(\mathbb{R}^{\pm} \times S^1)$ . For any  $\epsilon > 0$ , we define the catenoid  $C_{\epsilon}$  as the image of C by an homothety of parameter  $2\epsilon$ . We denote with  $X_{c,\epsilon} := 2\epsilon X_c$  its parametrization. Of course, by this transformation, to  $C_{\pm}$ , correspond two surfaces denoted  $C_{\epsilon,\pm}$ .

Up to some dilation, we can assume that the two ends  $E_{t,\epsilon}$  and  $E_{b,\epsilon}$  of  $M_{k,\epsilon}$  are asymptotic to some translated copy of the catenoid parametrized by  $X_{c,\epsilon}$  in the vertical direction. Therefore,  $E_{t,\epsilon}$  and  $E_{b,\epsilon}$  can be parametrized, respectively, by

$$X_{t,\epsilon} := X_{c,\epsilon} + w_t \, n_c + \sigma_{t,\epsilon} \, e_3 \tag{3.35}$$

for  $(s, \theta) \in (s_0, \infty) \times S^1$ ,

$$X_{b,\epsilon} := X_{c,\epsilon} - w_b \, n_c - \sigma_{b,\epsilon} \, e_3 \tag{3.36}$$

for  $(s, \theta) \in (-\infty, -s_0) \times S^1$ , where  $\sigma_{t,\epsilon}, \sigma_{b,\epsilon} \in \mathbb{R}$ , functions  $w_t, w_b$  tend exponentially fast to 0 as s goes to  $\infty$  reflecting the fact that the ends are asymptotic to a catenoidal end.

In section 3 of [24] it is given the expression of the mean curvature operator about of a surface close to a rescaled standard catenoid. We can adapt this result to our situation. We obtain that the surface parametrized by  $X_{c,\epsilon} + w n_c$  is minimal if and only if the function w satisfies the minimal surface equation  $H_w = 0$ , where

$$H_w = -\frac{1}{4\epsilon^2} \mathbb{L}_C w + \frac{1}{2\epsilon \cosh^2 s} Q_{2,\epsilon} \left( \frac{w}{2\epsilon \cosh s}, \frac{\nabla w}{2\epsilon \cosh s}, \frac{\nabla^2 w}{2\epsilon \cosh s} \right) + \frac{1}{2\epsilon \cosh s} \frac{1}{2\epsilon \cosh s} + \frac{1}{2\epsilon \cosh s} \frac{1}{2\epsilon \cosh s} \frac{1}{2\epsilon \cosh s} + \frac{1}{2\epsilon \cosh s} \frac{1}{2\epsilon \cosh$$

$$\frac{1}{2\epsilon \cosh s} Q_{3,\epsilon} \left( \frac{w}{2\epsilon \cosh s}, \frac{\nabla w}{2\epsilon \cosh s}, \frac{\nabla^2 w}{2\epsilon \cosh s} \right).$$
(3.37)

Here  $\mathbb{L}_C$  is the Jacobi operator about the catenoid, that is

$$\mathbb{L}_C w = \frac{1}{\cosh^2 s} \left( \frac{\partial^2 w}{\partial s^2} + \frac{\partial^2 w}{\partial \theta^2} + \frac{2w}{\cosh^2 s} \right)$$

and  $Q_{2,\epsilon}$  and  $Q_{3,\epsilon}$  are functions which are bounded in  $\mathcal{C}^k(\mathbb{R} \times S^1)$  for all k, uniformly in  $\epsilon$ . They satisfy

$$Q_{2,\epsilon}(0,0,0) = Q_{3,\epsilon}(0,0,0) = 0$$
 and  $\nabla Q_{2,\epsilon}(0,0,0) = \nabla Q_{3,\epsilon}(0,0,0) = 0$ , (3.38)

$$\nabla^2 Q_{3,\epsilon}(0,0,0) = 0. \tag{3.39}$$

We will write for short

$$Q_{\epsilon}(w_{\Phi}) = \frac{1}{2\epsilon \cosh^2 s} Q_{2,\epsilon} \left( \frac{w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla^2 w_{\Phi}}{2\epsilon \cosh s} \right) + \frac{1}{2\epsilon \cosh s} Q_{3,\epsilon} \left( \frac{w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla^2 w_{\Phi}}{2\epsilon \cosh s} \right).$$
(3.40)

For all  $r < \rho_0$  and  $s > s_0$ , we define

$$M_{k,\epsilon}(s,r) := M_{k,\epsilon} - \left[ X_{t,\epsilon}((s,\infty) \times S^1) \cup X_{b,\epsilon}((-\infty,-s) \times S^1) \cup X_{m,\epsilon}(B_r(0)) \right].$$
(3.41)

The parametrizations of the three ends of  $M_{k,\epsilon}$  induce a decomposition of  $M_{k,\epsilon}$  into slightly overlapping components: a compact piece  $M_{k,\epsilon}(s_0+1,\rho_0/2)$  and three noncompact pieces  $X_{t,\epsilon}((s_0,\infty) \times S^1), X_{b,\epsilon}((-\infty,-s_0) \times S^1)$  and  $X_{m,\epsilon}(\bar{B}_{\rho_0}(0))$ .

We define a weighted space of functions on  $M_{k,\epsilon}$ .

**Definition 39.** Given  $\ell \in \mathbb{N}$ ,  $\alpha \in (0,1)$  and  $\Delta \in \mathbb{R}$ , the space  $\mathcal{C}^{\ell,\alpha}_{\delta}(M_{k,\epsilon})$  is defined to be the space of functions in  $\mathcal{C}^{\ell,\alpha}_{loc}(M_{k,\epsilon})$  for which the following norm is finite

$$\|w\|_{\mathcal{C}^{\ell,\alpha}_{\delta}(M_{k,\epsilon})} := \|w\|_{\mathcal{C}^{\ell,\alpha}(M_{k,\epsilon}(s_{0}+1,\rho_{0}/2))} + \|w \circ X_{m,\epsilon}\|_{\mathcal{C}^{\ell,\alpha}(B_{\rho_{0}}(0))}$$
  
+ 
$$\sup_{s \geqslant s_{0}} e^{-\delta s} \left(\|w \circ X_{t,\epsilon}\|_{\mathcal{C}^{\ell,\alpha}([s,s+1] \times S^{1})} + \|w \circ X_{b,\epsilon}\|_{\mathcal{C}^{\ell,\alpha}([-s-1,-s] \times S^{1})}\right)$$

and which are invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane.

We remark that there is no weight on the middle end. In fact we compactify this end and we consider a weighted space of functions defined on a two ended surface. We will perturb the surface  $M_{k,\epsilon}$  by the normal graph of a function  $u \in C^{2,\alpha}_{\delta}(M_{k,\epsilon})$ . In particular the middle end  $E_{m,\epsilon}$  will be just translated in the vertical direction.

#### 3.8.2 The Jacobi operator

The Jacobi operator about  $M_{k,\epsilon}$  is

$$\mathbb{L}_{M_{k,\epsilon}} := \Delta_{M_{k,\epsilon}} + |A_{M_{k,\epsilon}}|^2$$

where  $|A_{M_{k,\epsilon}}|$  is the norm of the second fundamental form on  $M_{k,\epsilon}$ .

In the parametrization of the ends introduced above, the volume forms  $dvol_{M_{k,\epsilon}}$  can be written as  $\gamma_t ds d\theta$  and  $\gamma_b ds d\theta$  near the catenoidal type ends and as  $\gamma_m dx_1 dx_2$  near the middle end. Now we can define globally on  $M_{k,\epsilon}$  a smooth function

$$\gamma: M_{k,\epsilon} \longrightarrow [0,\infty)$$

that is identically equal to  $4\epsilon^2$  on  $M_{k,\epsilon}(s_0 - 1, 2\rho_0)$  and equal to  $\gamma_t$  (resp.  $\gamma_b, \gamma_m$ ) on the end  $E_{t,\epsilon}$  (resp.  $E_{b,\epsilon}, E_m$ ). They are defined in such a way that on  $X_{t,\epsilon}((s_0, \infty) \times S^1)$  and on  $X_{b,\epsilon}((-\infty, s_0) \times S^1)$  we have

$$\gamma \circ X_{t,\epsilon}(s,\theta) \sim 4\epsilon^2 \cosh^2 s$$
 and  $\gamma \circ X_{b,\epsilon}(s,\theta) \sim 4\epsilon^2 \cosh^2 s$ .

Finally on  $X_{m,\epsilon}(B_{\rho_0})$ , we have

$$\gamma \circ X_m(x) \sim \frac{4\epsilon^2}{|x|^4}.$$

It is possible to check that:

$$\mathcal{L}_{\epsilon,\delta} : \ \mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon}) \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon}) w \longmapsto \gamma \mathbb{L}_{M_{k,\epsilon}}(w)$$

is a bounded linear operator. The subscript  $\delta$  is meant to keep track of the weighted space over which the Jacobi operator is acting. Observe that, the function  $\gamma$  is here to counterbalance the effect of the conformal factor  $\frac{1}{\sqrt{|g_{M_{k,\epsilon}}|}}$  in the expression of the Laplacian in the coordinates we use to parametrize the ends of the surface  $M_{k,\epsilon}$ . This is precisely what is needed to have the operator defined from the space  $C^{2,\alpha}_{\delta}(M_{k,\epsilon})$  into the target space  $C^{0,\alpha}_{\delta}(M_{k,\epsilon})$ .

To have a better grasp of what is going on, let us linearize the nonlinear equation (3.37) at w = 0. We get the expression of the Jacobi operator about the rescaled catenoid  $C_{\epsilon}$ 

$$\mathbb{L}_{C_{\epsilon}} := \frac{1}{4\epsilon^2 \cosh^2 s} \left( \partial_s^2 + \partial_{\theta}^2 + \frac{2}{\cosh^2 s} \right).$$

We can observe that the operator  $4\epsilon^2 \cosh^2 s \mathbb{L}_{C_{\epsilon}}$  maps the space  $(\cosh s)^{\delta} \mathcal{C}^{2,\alpha}((s_0,\infty) \times S^1)$ into the space  $(\cosh s)^{\delta} \mathcal{C}^{0,\alpha}((s_0,\infty) \times S^1)$ . Similarly, if we linearize the nonlinear equation (3.31) at u = 0, we obtain the expression of the Jacobi operator about the plane times  $4\epsilon^2$ .

$$\mathbb{L}_{\mathbb{R}^2_{\epsilon}} := \frac{|x|^4}{4\epsilon^2} \,\Delta.$$

Again, the operator  $\frac{4\epsilon^2}{|x|^4} \mathbb{L}_{\mathbb{R}^2_{\epsilon}} = \Delta$  clearly maps the space  $\mathcal{C}^{2,\alpha}(\bar{B}_{\rho_0})$  into the space  $\mathcal{C}^{0,\alpha}(\bar{B}_{\rho_0})$ . Now, the function  $\gamma$  plays, for the ends of the surface  $M_{k,\epsilon}$ , the role played by the function  $\cosh^2 s$  for the ends of the standard catenoid and the role played by the function  $|x|^{-4}$  for the plane. Since the Jacobi operator about  $M_{k,\epsilon}$  is asymptotic to  $\mathbb{L}_{\mathbb{R}^2_{\epsilon}}$  at  $E_{m,\epsilon}$  and is asymptotic to  $\mathbb{L}_{C_{\epsilon}}$  at  $E_{t,\epsilon}$  and  $E_{b,\epsilon}$ , we conclude that the operator  $\mathcal{L}_{\epsilon,\delta}$  maps  $\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon})$  into  $\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})$ .

Now we recall the notion of non degeneracy introduced in [11].

**Definition 40.** The surface  $M_{k,\epsilon}$  is said to be non degenerate if  $\mathcal{L}_{\epsilon,\delta}$  is injective for all  $\delta < -1$ .

It useful to observe that a duality argument in the weighted Lebesgue spaces, implies that

 $(\mathcal{L}_{\epsilon,\delta} \text{ is injective}) \Leftrightarrow (\mathcal{L}_{\epsilon,-\delta} \text{ is surjective})$ 

if  $\delta \notin \mathbb{Z}$ . See [27] and [18] for more details.

The non degeneracy of  $M_{k,\epsilon}$  is related to the mapping properties of  $\mathcal{L}_{\epsilon,\delta}$  and to the kernel of this operator. From the observations made above, it follows that at the catenoidal type ends and at the middle planar end the Jacobi operators of  $M_{k,\epsilon}$  and  $M_k$  are respectively asymptotic to  $\mathbb{L}_C$  and  $\mathbb{L}_{C_{\epsilon}}$  which coincide up to a multiplication by  $4\epsilon^2$ . So we could transpose the all results about the surface  $M_k(0)$  contained in [11] related to the study of its mean curvature operator, to the surface  $M_{k,\epsilon}$ , including non degeneracy.

The Jacobi fields. It is known that a smooth one parameter group of isometries containing the identity generates a Jacobi field, that is a solution of the equation  $\mathbb{L}_{M_{k,\epsilon}} u = 0$ . These solutions are generated by the following one parameter groups of isometries: the vertical translations, the translations along the  $x_1$ -axis, the dilations. See [11] for details.

The group of vertical translations generated by the Killing vector field  $\Xi(p) = e_3$  gives rise to the Jacobi field

$$\Phi^{0,+}(p) := n(p) \cdot e_3.$$

The vector field  $\Xi(p) = p$  that is associated to the one parameter group of dilation generates a Jacobi fields

$$\Phi^{0,-}(p) := n(p) \cdot p.$$

The Killing vector field  $\Xi(p) = e_1$  that generates the group of translations along the  $x_1$ -axis is associated to a Jacobi field

$$\Phi^{1,+}(p) := n(p) \cdot e_1.$$

Finally, we denote by

$$\Phi^{1,-}(p) := n(p) \cdot (e_2 \times p)$$

the Jacobi field associated to the Killing vector field  $\Xi(p) = e_2 \times p$  that generates the group of rotations about the  $x_2$ -axis.

The Jacobi equation has other solutions which are not taken into account because in the difference with the four Jacobi fields just introduced they are not invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane.

With these notations, we define the deficiency space

$$\mathcal{D} := \operatorname{Span}\{\chi_t \, \Phi^{j,\pm}, \chi_b \, \Phi^{j,\pm} : j = 0, 1\}$$

where  $\chi_t$  is a cutoff function that is identically equal to 1 on  $X_{t,\epsilon}((s_0 + 1, \infty) \times S^1)$ , identically equal to 0 on  $M_{k,\epsilon} - X_{t,\epsilon}((s_0, \infty) \times S^1)$  and that is invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane. Also, we agree that

$$\chi_b(\cdot) := \chi_t(-\cdot).$$

Clearly

$$\tilde{\mathcal{L}}_{\epsilon,\delta}: \ \mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon}) \oplus \mathcal{D} \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon}) \\
w \longmapsto \gamma \mathbb{L}_{M_{k,\epsilon}}(w)$$

is a bounded linear operator. The linear decomposition Lemma proved in [23] for constant mean curvature surfaces (see also [18] for minimal hypersurfaces) can be adapted to our situation and thanks to the results of S. Nayatani contained in [29],[30] and extended in chapter 1, stating that any bounded Jacobi field respecting the mirror symmetry with respect to the  $x_2 = 0$  plane is linear combination of  $\Phi^{0,\pm}$  and  $\Phi^{1,+}$ , we get the following result

**Proposition 41.** All bounded Jacobi fields on  $M_{k,\epsilon}$  that are invariant with respect to the  $x_2 = 0$  plane are linear combination of  $\Phi^{0,\pm}$  and  $\Phi^{1,+}$  and choose  $\delta \in (1,2)$ . Then the operator  $\tilde{\mathcal{L}}_{\epsilon,\delta}$  is surjective and has a kernel of dimension 4. Moreover, there exists  $G_{\epsilon,\delta}$  a right inverse for  $\tilde{\mathcal{L}}_{\epsilon,\delta}$  whose norm is bounded.

### 3.9 An infinite dimensional family of minimal surfaces which are close to a compact part of a rescaled Costa-Hoffman-Meeks type surface in $\mathbb{H}^2 \times \mathbb{R}$ .

We recall that in section 3.7 we found that the mean curvature with respect to the metric  $g_{hyp}$  of a surface  $\bar{S}$  contained in a cylindrical neighbourhood of  $\{0,0\} \times \mathbb{R}$  of sufficiently small radius in  $\mathbb{H} \times \mathbb{R}$  can be expressed in terms of the euclidean mean curvature of a surface S, which is related to  $\bar{S}$  by the relation  $S = f(\bar{S})$ , being f the map defined by (3.23).

In this section we will apply this result to prove the existence of a family of minimal surfaces close to the surface  $\overline{M}_{k,\epsilon} = f(M_{k,\epsilon})$  contained in a cylindrical neighbourhood of radius  $r_{\epsilon} = \sqrt{\epsilon}/2$  of  $\{0,0\} \times \mathbb{R}$ . We remark that the image by the map  $f^{-1}$  of a surface contained in such a domain is contained in a cylinder of radius  $2r_{\epsilon}$ .

We start giving the statement of a result that can be easily obtained by [11], lemma 2.2. It describes the region of the surface  $M_{k,\epsilon}$  which can be parametrized by a graph on a annular neighbourhood of  $2r_{\epsilon}$  contained in the  $x_3 = 0$  plane.

**Lemma 42.** There exists  $\epsilon_0 > 0$  such that, for all  $\epsilon \in (0, \epsilon_0)$  an annular part of the ends  $E_{t,\epsilon}$ ,  $E_{b,\epsilon}$  and  $E_{m,\epsilon}$  of  $M_{k,\epsilon}$  can be written as vertical graphs over the horizontal plane of the functions

$$U_t(r,\theta) = \sigma_{t,\epsilon} + 2\epsilon \ln\left(\frac{2r}{\epsilon}\right) + f_t(r,\theta), \qquad (3.42)$$
$$\partial_r U_t(r,\theta) = \left(\frac{2\epsilon}{r}\right) + f_{t,d}(r,\theta), \qquad (3.43)$$
$$U_b(r,\theta) = -\sigma_{b,\epsilon} - 2\epsilon \ln\left(\frac{2r}{\epsilon}\right) + f_b(r,\theta), \qquad (3.43)$$
$$\partial_r U_b(r,\theta) = -\left(\frac{2\epsilon}{r}\right) + f_{b,d}(r,\theta),$$

where  $f_i = \mathcal{O}(\epsilon^2)$ ,  $f_{i,d} = \mathcal{O}(\epsilon^{5/2})$ , i = t, b. As for the parametrization of the planar end, it satisfies

$$U_m(r,\theta) = f_m(r,\theta) = \mathcal{O}(\left(\frac{r}{2\epsilon}\right)^{-(k+1)})$$

$$\partial_r U_m(r,\theta) = f_{m,d}(r,\theta) = \mathcal{O}(\left(\frac{r}{2\epsilon}\right)^{-k}).$$
(3.44)

Here  $(r, \theta)$  are the polar coordinates in the  $x_3 = 0$  plane. The functions  $\mathcal{O}(\epsilon)$  are defined in the annulus  $B_{4\epsilon^{1/2}} - B_{\epsilon^{1/2}/4}$  and are bounded in  $\mathcal{C}_b^{\infty}$  topology by a constant (independent on  $\epsilon$ ) multiplied by  $\epsilon$ , where the partial derivatives are computed with respect to the vector fields  $r \partial_r$  and  $\partial_{\theta}$ .

Then  $M_{k,\epsilon}$  has two ends  $E_{t,\epsilon}$  and  $E_{b,\epsilon}$  which are graphs over the  $x_3 = 0$  plane of functions  $U_t$  and  $U_b$  defined on the annulus  $B_{4\epsilon^{1/2}} - B_{\epsilon^{1/2}/4}$ .

Taking into account the definition of  $\overline{M}_{k,\epsilon} = f(M_{k,\epsilon})$ , it is clear that a lemma with identic statement can be proved also for this surface.

We set  $s_{\epsilon} = -\frac{1}{2} \ln \epsilon$  and we define  $M_{k,\epsilon}^{T}$  to be equal to  $M_{k,\epsilon}$  from which we have removed the image of  $(s_{\epsilon}, +\infty) \times S^{1}$  by  $X_{t,\epsilon}$ , the image of  $(-\infty, -s_{\epsilon}) \times S^{1}$  by  $X_{b,\epsilon}$  and the image of  $B_{\rho_{\epsilon}}(0)$  by  $X_{m,\epsilon}$  with  $\rho_{\epsilon} := 2\epsilon \frac{1}{2r_{\epsilon}} = 2\epsilon^{1/2}$ . We set  $\bar{M}_{k,\epsilon}^{T} = f(M_{k,\epsilon}^{T})$ . In this section we will prove the existence of a family of surfaces close to  $\bar{M}_{k,\epsilon}^{T}$ . To this aim we will use proposition 38 and we will follow the work [11].

First, we modify the parametrization of the ends  $E_{t,\epsilon}$ ,  $E_{b,\epsilon}$  for appropriates values of s, so that, when  $r \in [\epsilon^{1/2}/4, 4\epsilon^{1/2}]$  the curves corresponding to the image of

$$\theta \to (2\epsilon r \cos \theta, 2\epsilon r \sin \theta, U_t(r, \theta)), \quad \theta \to (2\epsilon r \cos \theta, 2\epsilon r \sin \theta, U_b(r, \theta))$$

correspond, respectively, up to a vertical translation to the curves  $s = -2\epsilon \ln(2r)$  and  $s = 2\epsilon \ln(2r)$ .

The curve  $\theta \to (2\epsilon r \cos \theta, 2\epsilon r \sin \theta, U_m(r, \theta))$  corresponds to  $\rho = \frac{2\epsilon}{r}$ .

The second step is the modification of unit normal vector field on  $M_{k,\epsilon}$  into a transverse unit vector field  $\tilde{n}_{\epsilon}$  in such a way that it coincides with the normal vector field  $n_{\epsilon}$  on  $M_{k,\epsilon}$ , is equal to  $e_3$  on the graph over  $B_{2\epsilon^{1/2}} - B_{3\epsilon^{1/2}/8}$  of the functions  $U_t$  and  $U_b$  and interpolate smoothly between the different definitions of  $\tilde{n}_{\epsilon}$  in different subsets of  $M_{k,\epsilon}^T$ .

The graph of a function u, using the vector field  $\tilde{n}_{\epsilon}$ , will be a minimal surface if and only if u is a solution of a second order nonlinear elliptic equation of the form

$$\mathbb{L}_{M_{h,\epsilon}^{T}} u = \tilde{L}_{\epsilon} u + Q_{\epsilon} (u)$$

where  $\mathbb{L}_{M_{k,\epsilon}^T}$  is the Jacobi operator about  $M_{k,\epsilon}^T$ ,  $Q_{\epsilon}$  is a nonlinear second order differential operator and  $\tilde{L}_{\epsilon}$  is a linear operator which takes into account the change of the parametrization and of the change of the normal vector field. In [11] it is proved that this last operator has coefficients uniformly bounded by a constant times  $\epsilon^2$ . Now, we consider three functions  $\varphi_t, \varphi_b, \varphi_m \in C^{2,\alpha}(S^1)$  which are even, with respect to  $\theta$ ,  $\varphi_t, \varphi_b$  are  $L^2$  orthogonal to 1 and  $\cos \theta$  while  $\varphi_m$  is  $L^2$  orthogonal to 1. Assume that they satisfy

$$\|\varphi_t\|_{\mathcal{C}^{2,\alpha}} + \|\varphi_b\|_{\mathcal{C}^{2,\alpha}} + \|\varphi_m\|_{\mathcal{C}^{2,\alpha}} \leqslant \kappa \epsilon.$$

We set  $\Phi := (\varphi_t, \varphi_b, \varphi_m)$  and we define  $w_{\Phi}$  to be the function equal to

- 1.  $\chi_+ H_{\varphi_t}(s_{\epsilon} s, \cdot)$  on the image of  $X_{t,\epsilon}$  where  $\chi_+$  is a cut-off function equal to 0 for  $s \leq s_0 + 1$  and identically equal to 1 for  $s \in [s_0 + 2, s_{\epsilon}]$
- 2.  $\chi_{-} H_{\varphi_{b}}(s + s_{\epsilon}, \cdot)$  on the image  $X_{b,\epsilon}$  where  $\chi_{-}$  is a cut-off function equal to 0 for  $s \ge -s_{0} 1$  and identically equal to 1 for  $s \in [-s_{\epsilon}, -s_{0} 2]$
- 3.  $\chi_m \tilde{H}_{\rho_{\epsilon},\varphi_m}(\cdot, \cdot)$  on the image of  $X_{m,\epsilon}$ , where  $\chi_m$  is a cut-off function equal to 0 for  $r \ge \rho_0$  and identically equal to 1 for  $\rho \in [\rho_{\epsilon}, \rho_0/2]$
- 4. zero on the remaining part of the surface  $M_{k,\epsilon}^T$ .

We recall that the operators  $\tilde{H}$  and H have been introduced respectively in Propositions 48 and 47.

We would like to prove that, under appropriates hypotheses, the graph about  $\overline{M}_{k,\epsilon}^T$  of the function  $\overline{u} = f(u)$  with  $u = w_{\Phi} + v$ , is a minimal surface. If we denote it by  $\overline{\Sigma}_{\overline{u}}$ , this is equivalent to solve the equation:

$$H_{hyp}(\bar{\Sigma}_{\bar{u}}) = 0.$$

If we define  $\Sigma_u = f^{-1}(\bar{\Sigma}_{\bar{u}})$  and denote by  $N_u = (N_1(u), N_2(u), N_3(u))$  the unit normal vector to  $\bar{\Sigma}_{\bar{u}}$ , thanks to proposition 38 we can write the equation to solve as

$$(H_e(\Sigma_u) - 2(x_1N_1(u) + x_2N_2(u)))(1 + \mathcal{O}(r^2)) = 0.$$

 $H_e(\Sigma_u)$  is the mean curvature of the graph of the function u about  $M_{k,\epsilon}^T$ . Taking into account that  $u = w_{\Phi} + v$ , its expression is given by

$$\mathbb{L}_{M_{k,\epsilon}^T}(w_{\Phi}+v) - \tilde{L}_{\epsilon}(w_{\Phi}+v) - Q_{\epsilon}(w_{\Phi}+v).$$

To simplify the notation we set  $-2(x_1N_1(u) + x_2N_2(u)) = P(w_{\Phi} + v)$ . The resolution of the previous equation is obtained thanks to the one of the following fixed point problem:

$$v = T(\Phi, v) \tag{3.45}$$

with

$$T(\Phi, v) = G_{\epsilon,\delta} \circ \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon}(w_{\Phi} + v) + P(w_{\Phi} + v) - \mathbb{L}_{M_{k,\epsilon}^{T}} w_{\Phi} + Q_{\epsilon}(w_{\Phi} + v) \right) \right)$$

where  $\delta \in (1, 2)$ , the operator  $G_{\epsilon,\delta}$  is defined in proposition 41 and  $\mathcal{E}_{\epsilon}$  is a linear extension operator such that

$$\mathcal{E}_{\epsilon}: \mathcal{C}^{0,\alpha}_{\delta}(M^T_{k,\epsilon}) \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon}),$$

where  $\mathcal{C}^{0,\alpha}_{\delta}(M^T_{k,\epsilon})$  denotes the space of functions of  $\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})$  restricted to  $M^T_{k,\epsilon}$ . It is defined by  $\mathcal{E}_{\epsilon}v = v$  in  $M^T_{k,\epsilon}$ ,  $\mathcal{E}_{\epsilon}v = 0$  in the image of  $[s_{\epsilon}+1, +\infty) \times S^1$  by  $X_{t,\epsilon}$ , in the image of  $(-\infty, -s_{\epsilon}-1) \times S^1$  by  $X_{b,\epsilon}$  and in the image of  $B_{\rho_{\epsilon}/2} \times S^1$  by  $X_{m,\epsilon}$ . Finally  $\mathcal{E}_{\epsilon}v$  is an interpolation of these values in the remaining part of  $M_{k,\epsilon}$  such that, for example,

$$(\mathcal{E}_{\epsilon}v) \circ X_{t,\epsilon}(s,\theta) = ((1+s_{\epsilon}-s)v) \circ X_{t,\epsilon}(s_{\epsilon},\theta)$$

for  $(s, \theta) \in [s_{\epsilon}, s_{\epsilon} + 1] \times S^1$ 

$$(\mathcal{E}_{\epsilon}v) \circ X_{b,\epsilon}(s,\theta) = ((1+s_{\epsilon}+s)v) \circ X_{b,\epsilon}(s_{\epsilon},\theta)$$

for  $(s, \theta) \in [-s_{\epsilon} - 1, -s_{\epsilon}] \times S^1$  and

$$(\mathcal{E}_{\epsilon}v) \circ X_{m,\epsilon}(\rho,\theta) = ((\frac{2}{\rho_{\epsilon}}\rho - 1)v) \circ X_{m,\epsilon}(\rho_{\epsilon},\theta)$$

for  $(\rho, \theta) \in [\rho_{\epsilon}/2, \rho_{\epsilon}] \times S^1$ .

**Remark 43.** From the definition of  $\mathcal{E}_{\epsilon}$ , if  $supp v \cap (B_{\rho_0} - B_{\rho_{\epsilon}}) \neq \emptyset$  then

$$\|(\mathcal{E}_{\epsilon}v) \circ X_{m,\epsilon}\|_{\mathcal{C}^{0,\alpha}(B_{\rho_0})} \leqslant c\epsilon^{-\alpha} \|v \circ X_{m,\epsilon}\|_{\mathcal{C}^{0,\alpha}(B_{\rho_0})}$$

This phenomenon of explosion of the norm does not occur near the catenoidal type ends:

$$\|(\mathcal{E}_{\epsilon}v) \circ X_{t,\epsilon}\|_{\mathcal{C}^{0,\alpha}([s_{\epsilon}+1,+\infty)\times S^{1})} \leqslant c \|v \circ X_{t,\epsilon}\|_{\mathcal{C}^{0,\alpha}([s_{\epsilon}+1,+\infty)\times S^{1})}$$

A similar equation holds for the bottom end. In the following we will assume  $\alpha > 0$  and near to zero.

The existence of a solution  $v \in \mathcal{C}^{2,\alpha}_{\delta}(M^T_{k,\epsilon})$  for the equation (3.45) is a consequence of the following result which proves that T is a contracting mapping.

**Lemma 44.** There exist constants  $c_{\kappa} > 0$  and  $\epsilon_{\kappa} > 0$ , such that

$$\|T(\Phi,0)\|_{\mathcal{C}^{2,\alpha}_{\kappa}} \leqslant c_{\kappa} \epsilon^2 \tag{3.46}$$

and, for all  $\epsilon \in (0, \epsilon_{\kappa})$ 

$$\|T(\Phi, v_2) - T(\Phi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon})} \leq \frac{1}{2} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon}^T)}$$

for all  $v_1, v_2 \in \mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon}^T)$  and satisfying  $\|v\|_{\mathcal{C}^{2,\alpha}_{\delta}} \leq 2 c_{\kappa} \epsilon^2$ 

**Proof.** We recall that the Jacobi operator associated to  $M_{k,\epsilon}$ , is asymptotic to the operator  $\mathbb{L}_{C,\epsilon}$  near the catenoidal ends, and it is asymptotic to the laplacian near of the planar end. The function  $w_{\Phi}$  is identically zero far from the ends where the explicit expression of  $\mathbb{L}_{M_{k,\epsilon}}$  is not known: this is the reason of our particular choice in the definition of  $w_{\Phi}$ . Then from the definition of  $w_{\Phi}$  and thanks to proposition 41 we obtain the estimate

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma\mathbb{L}_{M_{k,\epsilon}^{T}}w_{\Phi}\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} &\leqslant ||\gamma_{t}\mathbb{L}_{C,\epsilon}(w_{\Phi}\circ X_{t,\epsilon})||_{\mathcal{C}_{\delta}^{0,\alpha}([s_{0}+1,s_{\epsilon}]\times S^{1})} + \\ + ||\gamma_{b}\mathbb{L}_{C,\epsilon}(w_{\Phi}\circ X_{b,\epsilon})||_{\mathcal{C}_{\delta}^{0,\alpha}[-s_{\epsilon},-s_{0}-1]\times S^{1})} &\leqslant c \left|\left|\frac{2}{\cosh^{2}s}w_{\Phi}\circ X_{t,\epsilon}\right|\right|_{\mathcal{C}_{\delta}^{0,\alpha}[s_{0}+1,s_{\epsilon}]\times S^{1})} \\ &\qquad c\epsilon\|w_{\Phi}\circ X_{t,\epsilon}\|_{\mathcal{C}_{\delta}^{2,\alpha}[s_{0}+1,s_{\epsilon}]\times S^{1})} \leqslant c_{\kappa}\,\epsilon^{2+\frac{\delta}{2}}.\end{aligned}$$

Using the properties of  $\tilde{L}_{\epsilon}$ , we obtain

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma \tilde{L}_{\epsilon} w_{\Phi}\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} \leqslant c\epsilon^{2} \|w_{\Phi} \circ X_{t,\epsilon}\|_{\mathcal{C}^{2,\alpha}_{\delta}([s_{0}+1,s_{\epsilon}]\times S^{1})} + c\epsilon^{2} \|w_{\Phi} \circ X_{m,\epsilon}\|_{(\mathcal{C}^{2,\alpha}[\rho_{\epsilon},\rho_{0}/2]\times S^{1})} \leqslant c_{\kappa} \epsilon^{2}. \end{aligned}$$

The estimate of  $\|\mathcal{E}_{\epsilon}(\gamma P(w_{\Phi}))\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})}$  is related to the estimate of the horizontal components of the normal vector to surface at the catenoidal type ends and the middle planar end and to the definition of the function  $\gamma$ . It is possible to show that  $\|\mathcal{E}_{\epsilon}(\gamma P(w_{\Phi}))\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} \leq \epsilon^{2}$ .

As for the last term, we recall that the operator  $Q_{\epsilon}$  has two different expressions if we consider the catenoidal type end and the middle planar end (see equation (3.37) and (3.32)). In particular we assume that at the middle planar end,  $Q_{\epsilon}$  keeps track also of the operator  $\bar{L}_u$ , for  $u = u_m$  (see (3.30)) that appears in the expression of the mean curvature operator given by (3.34). It holds that  $\|\bar{L}_{u_m} w_{\Phi}\|_{\mathcal{C}^{2,\alpha}_{\delta}([\rho_{\epsilon},\rho_0/2]\times S^1)} \leq c\epsilon \|w_{\Phi} \circ X_{m,\epsilon}\|_{\mathcal{C}^{2,\alpha}_{\delta}([\rho_{\epsilon},\rho_0/2]\times S^1)} \leq c_k \epsilon^2$ . We find the following estimate

$$\left\|\mathcal{E}_{\epsilon}\left(\gamma Q_{\epsilon}\left(w_{\Phi}\right)\right)\right\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} \leqslant c_{k}\epsilon^{2}.$$

It is convenient to recall that the expression of  $Q_{\epsilon}$  at the catenoidal type ends is given by (3.40). So we obtain

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma Q_{\epsilon}\left(w_{\Phi}\right)\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} &\leqslant c\epsilon \|Q_{2,\epsilon}\left(\frac{w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla w_{\Phi}}{2\epsilon \cosh s}, \frac{\nabla^{2} w_{\Phi}}{2\epsilon \cosh s}\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} \\ &\leqslant c\epsilon \|\frac{w_{\Phi}}{2\epsilon \cosh s} \circ X_{t,\epsilon}\|_{\mathcal{C}_{\delta}^{2,\alpha}([s_{0}+1,s_{\epsilon}]\times S^{1})}^{2} + c\epsilon \|\frac{w_{\Phi}}{2\epsilon \cosh s} \circ X_{b,\epsilon}\|_{\mathcal{C}_{\delta}^{2,\alpha}([-s_{\epsilon},-s_{0}-1]\times S^{1})}^{2} + \\ &+ c\epsilon^{1-\alpha} \|w_{\Phi} \circ X_{m,\epsilon}\|_{\mathcal{C}^{2,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})}^{2} + \|\bar{L}_{u_{m}}w_{\Phi}\|_{\mathcal{C}^{2,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})}^{2} \leqslant c_{k}\epsilon^{2} \end{aligned}$$

As for the second estimate, we recall that

$$T(\Phi, v) := G_{\epsilon,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon} (w_{\Phi} + v) + P(w_{\Phi} + v) - \mathbb{L}_{M_{k,\epsilon}^{T}} w_{\Phi} + Q_{\epsilon} (w_{\Phi} + v) \right) \right) \right).$$

Then

$$T(\Phi, v_2) - T(\Phi, v_1) = G_{\epsilon,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon} (w_{\Phi} + v_2) + P(w_{\Phi} + v_2) - \mathbb{L}_{M_{k,\epsilon}^T} w_{\Phi} + Q_{\epsilon} (w_{\Phi} + v_2) \right) \right) \right) - G_{\epsilon,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon} (w_{\Phi} + v_1) + P(w_{\Phi} + v_1) - \mathbb{L}_{M_{k,\epsilon}^T} w_{\Phi} + Q_{\epsilon} (w_{\Phi} + v_1) \right) \right) \right)$$
and

and

$$\begin{aligned} \|T(\Phi, v_{2}) - T(\Phi, v_{1})\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon})} &\leq c \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}_{\epsilon}(w_{\Phi} + v_{2}) + P(w_{\Phi} + v_{2}) - \tilde{L}_{\epsilon}(w_{\Phi} + v_{1}) - P(w_{\Phi} + v_{1}) + Q_{\epsilon} \left(w_{\Phi} + v_{2}\right) - Q_{\epsilon} \left(w_{\Phi} + v_{1}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} &\leq \\ &\leq \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}(v_{2} - v_{1})\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} + \|\mathcal{E}_{\epsilon} \left(\gamma \left(P(w_{\Phi} + v_{2}) - P(w_{\Phi} + v_{1})\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})} + \\ &+ \|\mathcal{E}_{\epsilon} \left(\gamma \left(Q_{\epsilon} \left(w_{\Phi} + v_{1}\right) - Q_{\epsilon} \left(w_{\Phi} + v_{2}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k,\epsilon})}.\end{aligned}$$

We observe that from the considerations above it follows that

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma\left(\tilde{L}_{\epsilon}(v_{2}-v_{1})\right)\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} \leqslant c\epsilon^{2}||v_{2}-v_{1}||_{\mathcal{C}_{\delta}^{2,\alpha}(M_{k,\epsilon}^{T})},\\ \|\mathcal{E}_{\epsilon}\left(\gamma\left(P(w_{\Phi}+v_{2})-P(w_{\Phi}-v_{1})\right)\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} \leqslant c\epsilon^{2}||v_{2}-v_{1}||_{\mathcal{C}_{\delta}^{2,\alpha}(M_{k,\epsilon}^{T})}.\end{aligned}$$

and

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma\left(Q_{\epsilon}\left(w_{\Phi}+v_{1}\right)-Q_{\epsilon}\left(w_{\Phi}+v_{2}\right)\right)\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})} \\ \leqslant c\epsilon\|v_{2}-v_{1}\|_{\mathcal{C}_{\delta}^{2,\alpha}(M_{k,\epsilon}^{T})}\|\frac{w_{\Phi}}{2\epsilon\cosh s}\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k,\epsilon})}+\|\bar{L}_{u_{m}}(v_{2}-v_{1})\|_{\mathcal{C}^{2,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})} \\ \leqslant \left(c_{k}\epsilon^{\frac{3}{2}+\frac{\delta}{2}}+c\epsilon\right)\|v_{2}-v_{1}\|_{\mathcal{C}_{\delta}^{2,\alpha}(M_{k,\epsilon}^{T})}.\end{aligned}$$

Then

$$\|T(\Phi, v_2) - T(\Phi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon})} \leq c\epsilon \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k,\epsilon}^T)}.$$

This argument provides a minimal surface  $\bar{M}_{k,\epsilon}^T(\Phi)$  which is close to  $\bar{M}_{k,\epsilon}^T$  and has three boundaries. This surface is, close to its upper and lower boundary, a vertical graph over the annulus  $B_{r_{\epsilon}} - B_{r_{\epsilon}/2}$ , whose parametrization is, respectively, given by

$$\bar{U}_t(r,\theta) = \sigma_{t,\epsilon} + 2\epsilon \ln(2r) + H_{\varphi_t}(s_\epsilon + \ln 2r, \theta) + V_t(r,\theta),$$

$$\bar{U}_b(r,\theta) = -\sigma_{b,\epsilon} - 2\epsilon \ln(2r) - H_{\varphi_b}(\ln 2r + s_{\epsilon},\theta) + V_b(r,\theta),$$

where  $s_{\epsilon} = -\frac{1}{2} \ln \epsilon$ . Nearby the middle boundary the surface is a vertical graph whose parametrization is

$$\bar{U}_m(r,\theta) = \tilde{H}_{\rho_{\epsilon},\varphi_m}\left(\frac{2\epsilon}{r},\theta\right) + V_m(r,\theta).$$

The boundaries of the surface correspond to  $r = r_{\epsilon} = \epsilon^{1/2}/2$ . All the functions  $V_i$ , i = t, b, m, depend non linearly on  $\epsilon, \varphi$ . The functions  $V_i(\epsilon, \varphi_i)$ , for i = t, b, satisfy  $\|V_i(\epsilon, \varphi_i)(r_{\epsilon})\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leq c\epsilon$  and

$$\|V_i(\epsilon,\varphi)(r_{\epsilon}\cdot) - V_i(\epsilon,\varphi')(r_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leqslant c\epsilon^{1-\frac{\delta}{2}} \|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}}.$$

The function  $V_m(\epsilon, \varphi_m)$  satisfies  $\|V_m(\epsilon, \varphi_m)(r_{\epsilon})\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1-B_{1/2})} \leq c\epsilon$  and

$$\|V_m(\epsilon,\varphi)(r_{\epsilon}\cdot) - V_m(\epsilon,\varphi')(r_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leq c\epsilon \|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}}.$$

#### 3.10 The matching of Cauchy data

In this section we shall complete the proof of Theorem 27.

We need introduce additional notation. Given an even function  $f \in C^{2,\alpha}(S^1)$  with the following Fourier expansion

$$f(\theta) = \sum_{n \in \mathbb{N}} a_n e_i(n\theta) = \sum_{n \in \mathbb{N}} a_n \cos(n\theta),$$

then we denote with  $\pi''(f)$  the function

$$\sum_{n \geqslant 2} a_n \cos(n\theta)$$

and with  $\pi'(f)$  the function

$$a_0 + a_1 \cos(\theta).$$

In section 3.6 we have obtained the minimal surfaces  $S_t, S_b$  which are graphs on  $D = B_1 \setminus B_{r_{\epsilon}} \subset \mathbb{H}^2$  and are asymptotic, respectively, to the parts of catenoid denoted  $C_t$  and  $C_b$  introduced in section 3.5. In section 3.4 we have obtained a surface  $S_m$  which is a minimal graph on D and is asymptotic to  $C_m := \mathbb{H}^2 \times \{0\}$ .

In other words we are able to solve, for  $\Phi = (\varphi_t, \varphi_b, \varphi_m) \in [\pi'' C^{2,\alpha}(S^1)]^3$  and  $\|\varphi_i\|_{\mathcal{C}^{2,\alpha}} \leq \epsilon$ a system of minimal surface equations, that we write for short as

$$\begin{cases} L_c u_S = Q_c(u_S) & \text{on } C_t, C_b \\ L_p u_S = Q_p(u_S) & \text{on } C_m \\ \pi'' u_S = (\varphi_t - \pi'' v_t, \varphi_b - \pi'' v_b, \varphi_m) & \text{on } (\partial C_t, \partial C_b, \partial C_m) \end{cases}$$

Here the index p and c mean that the corresponding minimal surface equation is relative, respectively, to a plane or a catenoid. So it is possible to construct three minimal graphs about the plane and two part of a catenoid in  $\mathbb{H}^2 \times \mathbb{R}$ . The functions  $v_t, v_b, v_{t,d}, v_{b,d}$  are introduced in (3.16) and in following equations.

The parametrizations of the surfaces  $S_t, S_b$  that are denoted by  $X'_{i,\Phi}$  with i = t, b, satisfy in a neighbourhood of their boundaries

$$X'_{i,\Phi} = 2\epsilon \ln\left(\frac{2r}{\epsilon}\right) + \varphi_i + \pi'(u_S + v_i)$$
$$\partial_r X'_{i,\Phi} = \frac{2\epsilon}{r} + v_{i,d} + \partial_r u_S$$

and a surface  $S_m$ , whose parametrization  $X'_{m,\Phi}$ , in a neighbourhood of its boundary, satisfies

$$X'_{m,\Phi} = \varphi'_m + \pi'(u_S)$$
$$\partial_r X'_{m,\Phi} = \partial_r u_S$$

Now we can define

$$\begin{split} E_{\epsilon} &: [\pi'' C^{2,\alpha}(S^1)]^3 & \longrightarrow \quad C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3 \\ \Phi & \longrightarrow \quad [(X'_{t,\Phi}, X'_{b,\Phi}, X'_{m,\Phi}), (\partial_r X'_{t,\Phi}, \partial_r X'_{b,\Phi}, \partial_r X'_{m,\Phi})]_{|r_{\epsilon}}. \end{split}$$

Finally, in section 3.9 we have obtained the surface  $\overline{M}_{k,\epsilon}^{T}(\Phi)$ , whose boundary consists in three curves on  $\partial B_{r_{\epsilon}}$ . In fact we solved, for  $\Phi = (\varphi_t, \varphi_b, \varphi_m) \in [\pi'' C^{2,\alpha}(S^1)]^3$  and  $\|\varphi_i\|_{\mathcal{C}^{2,\alpha}} \leq \epsilon$  the problem

$$\begin{cases} H_{hyp}(\Sigma_{u_C}) &= 0 & \text{on } \bar{M}_{k,\epsilon} \\ \pi'' u_C &= (\varphi_t - \pi'' f_t, \varphi_b - \pi'' f_b, \varphi_m - \pi'' f_m) & \text{on } \partial \bar{M}_{k,\epsilon} \end{cases}$$

The functions  $f_t$ ,  $f_b$ ,  $f_m$ ,  $f_{t,d}$ ,  $f_{b,d}$ ,  $f_{m,d}$  are introduced in (3.42), (3.43) and (3.44). The parametrization of the surface  $\Sigma_{u_C}$  near the catenoidal type ends are denoted by  $X_{i,\Phi}$  with i = t, b. In a neighbourhood of the boundaries of  $\Sigma_{u_C}$ , they satisfy

$$X_{i,\Phi} = 2\epsilon \ln \frac{2r}{\epsilon} + \varphi_i + \pi'(u_C + f_i)$$
$$\partial_r X_{i,\Phi} = \frac{2\epsilon}{r} + f_{i,d} + \partial_r u_C$$

The parametrization of the middle planar end,  $X_{m,\Phi}$ , in a neighbourhood of its boundary, satisfies

$$X_{m,\Phi} = \varphi_m + \pi' (u_C + f_m)$$
$$\partial_r X_{m,\Phi} = f_{m,d} + \partial_r u_C.$$

We define

$$\begin{array}{rcl} F_{\epsilon} : [\pi'' C^{2,\alpha}(S^1)]^3 & \longrightarrow & C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3 \\ \Phi & \longrightarrow & [(X_{t,\Phi}, X_{b,\Phi}, X_{m,\Phi}), (\partial_r X_{t,\Phi}, \partial_r X_{b,\Phi}, \partial_r X_{m,\Phi})]_{|r_{\epsilon}} \end{array}$$

We set  $C_{\epsilon} := E_{\epsilon} - F_{\epsilon}$ .

We want to prove that the surfaces  $S_t, S_b, S_m$  and  $\overline{M}_{k,\epsilon}^T$  can be glued along their boundaries to obtain a  $C^{\infty}$ -surface. Firstly we will show that these surface correspond in a  $C^1$  way along the boundaries curves. This is true if it exists  $\Psi = (\psi_1, \psi_2, \psi_3)$  such that  $C_{\epsilon}(\Psi) = 0$ . The existence of the appropriates boundary functions is proven in the following theorem. Finally, to show that the surface is  $C^{\infty}$ , it is sufficient to apply the regularity theory. That completes the proof of Theorem 27.

**Theorem 45.** There exists  $\epsilon_0$  such that, for each  $0 < \epsilon < \epsilon_0$ , there exists  $\Psi = (\psi_1, \psi_2, \psi_3) \in [\pi'' C^{2,\alpha}(S^1)]^3$  which solves  $C_{\epsilon}(\Psi) = 0$ .

**Proof.** We consider the harmonic extensions of  $\psi_i$ , i = 1, 2, 3, on the ends of  $\overline{M}_{k,\epsilon}$ , that is

- 1.  $\bar{w}_t = \chi_+ H_{\psi_1}(s_{\epsilon} s, \cdot)$ , on the upper end
- 2.  $\bar{w}_b = \chi_- H_{\psi_2}(s + s_{\epsilon}, \cdot)$ , on the lower end
- 3.  $\bar{w}_m = \chi_m \tilde{H}_{R_{\epsilon},\psi_3}(\cdot,\cdot)$  on the middle end (the definition of the map f is given by (3.23)),

and its harmonic extensions

- 1.  $w_t = \mathcal{H}_{r_{\epsilon},\psi_1}$  on  $S_t$
- 2.  $w_b = -\mathcal{H}_{r_{\epsilon},\psi_2}$  on  $S_b$
- 3.  $w_m = \mathcal{H}_{r_{\epsilon},\psi_3}$  on  $S_m$

(see section 3.9 for the definitions of the cut-off functions). We recall that the operators  $\mathcal{H}$ , H and  $\tilde{H}$  have been introduced respectively in proposition 46, 47 and 48. We consider the following maps

$$E_0 : [\pi'' C^{2,\alpha}(S^1)]^3 \longrightarrow C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3$$
$$\Psi \longrightarrow [(w_t, w_b, w_m), (\partial_r w_t, \partial_r w_b, \partial_r w_m)]_{|r_e}$$

and

$$F_0: [\pi'' C^{2,\alpha}(S^1)]^3 \longrightarrow C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3$$
$$\Psi \longrightarrow [(\bar{w}_t, \bar{w}_b, \bar{w}_m), (\partial_r \bar{w}_t, \partial_r \bar{w}_b, \partial_r \bar{w}_m)]_{|r_e|}$$

Now using Fourier expansion of the function, we can see that  $C_0 = E_0 - F_0$  has an inverse which is bounded independently of  $\epsilon$ . In particular, the equation  $C_0(\Psi) = 0$  has the unique solution  $\Psi = (0, 0, 0)$ . Now we consider  $(C_{\epsilon} - C_0)(\Psi)$ , whose expression is

$$(\pi'(u_S - u_C) + \pi'(v_t - f_t), \pi'(u_S - u_C) + \pi'(v_b - f_b), \pi'(u_S - u_C) + \pi'(-f_m),$$
  
$$\partial_r(u_S - w_t) - \partial_r(u_C - \bar{w}_t) + v_{t,d} - f_{t,d},$$
  
$$\partial_r(u_S - w_b) - \partial_r(u_C - \bar{w}_b) + v_{b,d} - f_{b,d},$$
  
$$\partial_r(u_S - w_m) - \partial_r(u_C - \bar{w}_m) - f_{m,d}).$$

It is easy to prove that

$$||(C_{\epsilon} - C_0)(\Psi)||_{C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3} \leqslant \epsilon.$$

In order to solve  $C_{\epsilon}(\Psi) = 0$ , we find a fixed point for the mapping

$$D_{\epsilon}(\Psi) := C_0^{-1}\left( (C_{\epsilon} - C_0)(\Psi) \right).$$

### 3.11 Appendix

The results contained in this section are about the existence of some harmonic extension operators. The first one gives the harmonic extension of a function on  $\mathbb{H}^2 \setminus D_{r_0}$ .

**Proposition 46.** If  $\mu \in (0, 1)$  there exists an operator

$$\mathcal{H}_{r_0}: C^{2,\alpha}(S^1) \longrightarrow C^{2,\alpha}_1(S^1 \times [r_0, 1)),$$

such that for every function  $\varphi(\theta) \in C^{2,\alpha}(S^1)$ , which is  $L^2$ -orthogonal to  $e_0, e_1$ , the function  $w_{\varphi} = \mathcal{H}_{r_0,\varphi}$  solves

$$\begin{cases} \Delta_{eucl} w_{\varphi} = 0 & on \quad S^{1} \times [r_{0}, 1) \\ w_{\varphi} = \varphi & on \quad S^{1} \times \{r_{0}\}. \end{cases}$$

Moreover,

$$||\mathcal{H}_{r_{0},\varphi}||_{C_{1}^{2,\alpha}(S^{1}\times[r_{0},1))} \leqslant c \, ||\varphi||_{C^{2,\alpha}(S^{1})},\tag{3.47}$$

for some constant c > 0.

**Proof.** We observe that

$$\Delta_{eucl}(1-r^2)^{\mu} = \left(-4\mu(1-r^2) + 4\mu(\mu-1)r^2\right) \left(1-r^2\right)^{\mu-2} = \left(-4\mu + 4\mu^2 r^2\right) \left(1-r^2\right)^{\mu-2} \leqslant 0$$

if  $\mu \in (0, 1]$ . So  $(1 - r^2)^{\mu}$  it's a superharmonic function. Then the function  $||\varphi||_{C^{2,\alpha}(S^1)}(1 - r^2)$  can be used as barrier function and we can apply the Perron method. We can conclude that the solution, that we denote with  $w_{\varphi}$ , exists and satisfies

$$|w_{\varphi}| \leqslant c ||\varphi||_{C^{2,\alpha}(S^1)} (1 - r^2).$$

Using the initial assumption we can write

$$(1-r^2)^{-1}|w_{\varphi}| \leqslant c|\varphi|.$$

The estimates for the derivatives of  $w_{\varphi}$  are obtained by Schauder estimates. We can conclude that

$$||w_{\varphi}||_{C^{2,\alpha}_{1}} \leqslant c||\varphi||_{C^{2,\alpha}(S^{1})}.$$

Now we give the statement of a result whose proof is contained in [7]. It gives the harmonic extension of a function on a half catenoid.

**Proposition 47.** There exists an operator

$$H: \mathcal{C}^{2,\alpha}(S^1) \longrightarrow \mathcal{C}^{2,\alpha}_{-2}([0,+\infty) \times S^1),$$

such that for all  $\varphi \in C^{2,\alpha}(S^1)$ , even function and orthogonal to  $e_i$ , i = 0, 1 in the  $L^2$ -sense, the function  $w = H(\varphi)$  solves

$$\begin{cases} (\partial_s^2 + \partial_\theta^2) w = 0 & in \quad S^1 \times [0, +\infty) \\ \\ w = \varphi & on \quad S^1 \times \{0\} \end{cases}$$

Moreover

$$\|H(\varphi)\|_{\mathcal{C}^{2,\alpha}_{-2}([0,+\infty)\times S^1)} \leqslant c \, \|\varphi\|_{\mathcal{C}^{2,\alpha}(S^1)},$$

for some constant c > 0.

The following result gives a harmonic extension of a function on  $\mathbb{R}^2 \setminus D_{\bar{\rho}}$ .

Proposition 48. There exists an operator

$$\tilde{H}_{\bar{\rho}}: C^{2,\alpha}(S^1) \longrightarrow C^{2,\alpha}_{-2}(S^1 \times [\bar{\rho}, +\infty)),$$

such that for each even function  $\varphi(\theta) \in C^{2,\alpha}(S^1)$ , which is  $L^2$ -orthogonal to the constant function and  $\cos \theta$ , then  $w_{\varphi} = \tilde{H}_{\bar{\rho},\varphi}$  solves

$$\begin{cases} \Delta w_{\varphi} = 0 & on \quad S^{1} \times [\bar{\rho}, +\infty) \\ w_{\varphi} = \varphi & on \quad S^{1} \times \{\bar{\rho}\}. \end{cases}$$

Moreover,

 $||\tilde{H}_{\bar{\rho},\varphi}||_{C^{2,\alpha}_{-2}(S^1 \times [\bar{\rho}, +\infty))} \leqslant c \, ||\varphi||_{C^{2,\alpha}(S^1)},\tag{3.48}$ 

for some constant c > 0.

**Proof.** We consider the decomposition of the function  $\varphi$  with respect to the basis  $\{\cos(i\theta)\}$ , that is

$$\varphi = \sum_{i=2}^{\infty} \varphi_i \cos(i\theta).$$

Then the solution  $w_{\varphi}$  is given by

$$w_{\varphi}(\rho,\theta) = \sum_{i=2}^{\infty} \left(\frac{\bar{\rho}}{\rho}\right)^{i} \varphi_{i} \cos(i\theta).$$

Since  $\frac{\bar{\rho}}{\rho} \leqslant 1$ , then  $\left(\frac{\bar{\rho}}{\rho}\right)^i \leqslant \left(\frac{\bar{\rho}}{\rho}\right)$ , we can conclude that  $|w(r,\theta)| \leqslant c\rho^{-2}|\varphi(\theta)|$  and then  $||w_{\varphi}||_{C^{2,\alpha}_{-2}} \leqslant c||\varphi||_{C^{2,\alpha}}$ .

## Chapter 4

# Singly periodic minimal surfaces with arbitrary nonzero genus and infinitely many ends

#### 4.1 Introduction

In 1988, H. Karcher [19, 20] defined a family of doubly periodic minimal surfaces, called toroidal halfplane layers, with genus one and four horizontal Scherk-type ends<sup>1</sup> in the quotient. In 1989, Meeks and Rosenberg [25] developed a general theory for doubly periodic minimal surfaces having finite topology in the quotient, and used an approach of minimax type to obtain the existence of a family of doubly periodic minimal surfaces, also with genus one and four horizontal Scherk-type ends in the quotient. These Karcher's and Meeks and Rosenberg's surfaces have been generalized in [35], constructing a 3-parameter family  $\mathcal{K} = \{M_{\sigma,\alpha,\beta}\}_{\sigma,\alpha,\beta}$  of surfaces, called KMR examples (sometimes, they are also referred in the literature as toroidal halfplane layers). Such examples have been classified by Pérez, Rodríguez and Traizet [33] as the only doubly periodic minimal surfaces with genus one and finitely many parallel (Scherk-type) ends in the quotient. The possible limits of KMR examples are: the catenoid, the helicoid, any singly or doubly periodic Scherk minimal surface, any Riemann minimal example or another KMR example.

Each  $M_{\sigma,\alpha,\beta}$  has an horizontal period  $T_1$  (the period at the ends) and a non horizontal period  $T_2$  coming from homology. We denote by  $\widetilde{M}_{\sigma,\alpha,\beta}$  the lifting of  $M_{\sigma,\alpha,\beta}$  to  $\mathbb{S}^1 \times \mathbb{R}^2$  by forgetting the period  $T_2$ . The surface  $\widetilde{M}_{\sigma,\alpha,\beta}$ , that we go on calling KMR example, has genus zero, infinitely many parallel Scherk-type ends, and two limit ends. We consider in this work KMR examples near the catenoidal limit, so  $\sigma \to 0$  or equivalently  $T_1 \to \infty$ 

 $<sup>^{1}</sup>$ A horizontal *Scherk-type* end is an end asymptotic to a horizontal half-plane, invariant by one of the period vectors of the surface.

and  $T_2 \to \infty$ , and with  $\alpha = 0$  or  $\beta = 0$ .

In 1982, C. Costa [3, 4] discovered a genus one minimal surface with two catenoidal ends and one planar middle end, which is embedded outside a ball in  $\mathbb{R}^3$ . D. Hoffmann and W.H. Meeks [14] proved the global embeddedness for this Costa's example, and generalized it for bigger genus. For each  $k \geq 1$ , we will denote by  $M_k$  such Costa-Hoffmann-Meeks' surface, which is a properly embedded minimal surface of genus k and three ends: one middle end asymptotic to the plane  $\{x_3 = 0\}$ , one top catenoidal end and one bottom catenoidal end. L. Hauswirth and F. Pacard [11] have obtained a deformation of  $M_k$ , for  $1 \leq k \leq 37$  (in these cases,  $M_k$  is nondegenerate [29],[30]), by bending the catenoidal ends. We denote such deformed examples by  $M_k(\xi)$ . It is known that their construction extends for  $k \geq 38$  thanks to the result showed in chapter 1.

V. Ramos Batista [2] has constructed a singly periodic Costa minimal surface, with two catenoidal ends, one Scherk-type middle end and genus one in the quotient. We produce two new families of examples of periodic minimal surfaces of higher genus as follows. We consider a compact part of  $M_k(\xi)$  contained in a vertical solid cylinder,  $D \times \mathbb{R}$ , of radius  $1/(2\sqrt{\epsilon})$  and centered at the origin. We glue it to a minimal graph on  $(\mathbb{R}^2 - D)/T_1$  asymptotic to  $\{x_3 = 0\}/T_1$ , to one half of  $\widetilde{M}_{\sigma,\alpha,\beta}$ , near the catenoidal limit, that is with  $\sigma$  small (one time with  $\alpha = 0$  and a second time with  $\beta = 0$ ) and with a Scherk type surface. We obtain two families of properly embedded minimal surfaces in  $\mathbb{S}^1 \times \mathbb{R}^2$  with genus  $k \ge 1$ , infinitely many parallel Scherk-type and two limit ends.

### 4.2 A Costa-Hoffman-Meeks type surface with bent catenoidal ends

In this section we recall the result shown in [11] about the existence of a family of minimal surfaces close to the Costa-Hoffman-Meeks surfaces of genus  $k \ge 1$ , one planar end and two slightly bent catenoidal ends by an angle  $\xi$ . We denote one member of the family by  $M_k(\xi)$ . Then  $M_k(0)$  is the family of the Costa-Hoffman-Meeks surface of genus k (see Costa [3], [4] and D. Hoffman and W. H. Meeks [14], [15], [16]).

The family of the Costa-Hoffman-Meeks surfaces. Each member of the family of surfaces  $M_k(0)$ , after suitable rotation and translation, enjoys the following properties.

- 1. It has one planar end  $E_m$  asymptotic to the  $x_3 = 0$  plane, one top end  $E_t$  and one bottom end  $E_b$  that are respectively asymptotic to the upper end and to the lower end of a catenoid with  $x_3$ -axis of revolution. The planar end  $E_m$  is located between the two catenoidal ends.
- 2. It is invariant under the action of the rotation of angle  $\frac{2\pi}{k+1}$  about the  $x_3$ -axis, under the action of the symmetry with respect to the  $x_2 = 0$  plane and under the action

of the composition of a rotation of angle  $\frac{\pi}{k+1}$  about the  $x_3$ -axis and the symmetry with respect to the  $x_3 = 0$  plane.

3. It intersects the  $x_3 = 0$  plane in k + 1 straight lines, which intersect themselves at the origin with angles equal to  $\frac{\pi}{k+1}$ . The intersection of  $M_k$  with the plane  $x_3 = const (\neq 0)$  is a single Jordan curve. The intersection of  $M_k$  with the upper half space  $x_3 > 0$  (resp. with the lower half space  $x_3 < 0$ ) is topologically an open annulus.

Now we give a local description of the surfaces  $M_k(0)$  near its ends and we introduce coordinates that we will use.

The planar end. The planar end  $E_m$  of the surface  $M_k$  can be parametrized by

$$X_m(x) := \left(\frac{x}{|x|^2}, u_m(x)\right) \in \mathbb{R}^3$$

where  $x \in \overline{B}_{\rho_0}(0) - \{0\} \subset \mathbb{R}^2$  and the function  $u_m$  tends to 0 like  $u_m(x) = \mathcal{O}(|x|^{k+1})$ . Here  $\rho_0 > 0$  is fixed small enough. The minimal surface equation has the following form

$$|x|^{4} \operatorname{div}\left(\frac{\nabla u}{(1+|x|^{4} |\nabla u|^{2})^{1/2}}\right) = 0.$$
(4.1)

It can be shown (see [11]) that the function  $u_m$  can be extended at the origin continuously using Weierstrass representation. We can prove that  $u_m \in C^{2,\alpha}(\bar{B}_{\rho_0})$ .

The catenoidal ends. We denote by  $X_c$  the parametrization of the standard catenoid C whose axis of revolution is the  $x_3$ -axis. Its expression is

$$X_c(s,\theta) := (\cosh s \, \cos \theta, \cosh s \, \sin \theta, s) \in \mathbb{R}^3$$

where  $(s, \theta) \in \mathbb{R} \times S^1$ . The unit normal vector field about C is given by

$$n_c(s,\theta) := \frac{1}{\cosh s} \left(\cos \theta, \sin \theta, -\sinh s\right).$$

Up to some dilation, we can assume that the two ends  $E_t$  and  $E_b$  of  $M_k$  are asymptotic to some translated copy of the catenoid parametrized by  $X_c$  in the vertical direction. Therefore,  $E_t$  and  $E_b$  can be parametrized, respectively, by

$$X_t := X_c + w_t \, n_c + \sigma_t \, e_3$$

for  $(s, \theta) \in (s_0, \infty) \times S^1$ ,

$$X_b := X_c - w_b \, n_c - \sigma_b \, e_3$$

for  $(s, \theta) \in (-\infty, -s_0, ) \times S^1$ , where  $\sigma_t, \sigma_b \in \mathbb{R}$ , functions  $w_t, w_b$  tend exponentially fast to 0 as s goes to  $\infty$  reflecting the fact that the ends are asymptotic to a catenoidal end.

We recall that the surface parametrized by by  $X := X_c + w n_c$  is minimal if and only if the function w satisfies the minimal surface equation which, for normal graphs over a catenoid has the following form

$$\frac{1}{\cosh^2 s} \left( \frac{\partial^2 w}{\partial s^2} + \frac{\partial^2 w}{\partial \theta^2} + \frac{2w}{\cosh^2 s} + Q_2 \left( \frac{w}{\cosh s} \right) + \cosh s Q_3 \left( \frac{w}{\cosh s} \right) \right) = 0, \quad (4.2)$$

where  $Q_2$ ,  $Q_3$  are linear second order differential operators which are bounded in  $\mathcal{C}^k(\mathbb{R}\times S^1)$  for all k. These functions satisfy  $Q_2(0) = Q_3(0) = 0$ ,  $\nabla Q_2(0) = \nabla Q_3(0) = 0$ ,  $\nabla^2 Q_3(0) = 0$  and then:

$$\|Q_{j}(v_{2}) - Q_{j}(v_{1})\|_{\mathcal{C}^{0,\alpha}([s,s+1]\times S^{1})} \leq c \left(\sup_{i=1,2} \|v_{i}\|_{\mathcal{C}^{2,\alpha}([s,s+1]\times S^{1})}\right)^{j-1} \|v_{2} - v_{1}\|_{\mathcal{C}^{2,\alpha}([s,s+1]\times S^{1})}$$

$$(4.3)$$

for all  $s \in \mathbb{R}$  and all  $v_1, v_2$  such that  $||v_i||_{\mathcal{C}^{2,\alpha}((s,s+1)\times S^1)} \leq 1$ . The constant c > 0 does not depend on s.

The family of Costa-Hoffman-Meeks surfaces with bent catenoidal ends. Using an elaborate version of the implicit function theorem and following [18] and [23] it is possible to prove the following

**Theorem 49** ([11]). There exists  $\xi_0 > 0$  and a smooth one parameter family of minimal hypersurfaces  $(M_k(\xi))_{\xi}$ , for  $\xi \in (-\xi_0, \xi_0)$ , with two catenoidal ends and one planar end. In particular  $M_k(0) = M_k$ , the upper (resp. lower) catenoidal end of  $M_k(\xi)$  is, up to a translation along its axis, asymptotic to the upper (resp. lower) end of the standard catenoid whose axis of revolution is directed by  $\sin \xi e_1 + \cos \xi e_3$ . Moreover  $M_k(\xi)$  has one horizontal planar end and is invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane.

The upper (lower) end of  $M_k(\xi)$  is, up to a translation, asymptotic to the upper (lower) end of the same (standard) catenoid. Then the upper end  $E_t(\xi)$  and the lower end  $E_b(\xi)$ of  $M_k(\xi)$ , if  $R_{\xi}$  denotes the rotation of angle  $\xi$  about the  $x_2$  axis, can be parametrized respectively by

$$X_{t,\xi} = R_{\xi} \left( X_c + w_{t,\xi} \, n_c \right) + \sigma_{t,\xi} \, e_3 + \varsigma_{t,\xi} \, e_1 \tag{4.4}$$

$$X_{b,\xi} = R_{\xi} \left( X_c - w_{b,\xi} \, n_c \right) - \sigma_{b,\xi} \, e_3 - \varsigma_{b,\xi} \, e_1 \tag{4.5}$$

where the functions  $w_{t,\xi}, w_{b,\xi}$ , the numbers  $\sigma_{t,\xi}, \varsigma_{t,\xi}, \sigma_{b,\xi}, \varsigma_{b,\xi} \in \mathbb{R}$  depend smoothly on  $\xi$ and satisfy  $w_{t,0} = w_t, w_{b,0} = w_b, \sigma_{b,0} = \sigma_b \sigma_{t,0} = \sigma_t, \varsigma_{t,0} = 0$  and  $\varsigma_{b,0} = 0$ . For all  $r < \rho_0$  and  $s > s_0$ , we define

$$M_k(\xi, s, r) := M_k(\xi) - \left[ X_{t,\xi}((s, \infty) \times S^1) \cup X_{b,\xi}((-\infty, -s) \times S^1) \cup X_m(B_\rho(0)) \right].$$
(4.6)

The parametrizations of the three ends of  $M_k(\xi)$  induce a decomposition of  $M_k(\xi)$  into slightly overlapping components: a compact piece  $M_k(\xi, s_0 + 1, \rho_0/2)$  and three noncompact pieces  $X_{t,\xi}((s_0, \infty) \times S^1), X_{b,\xi}((-\infty, -s_0) \times S^1)$  and  $X_m(\bar{B}_{\rho_0}(0))$ .

We define the weighted space of functions on  $M_k(\xi)$ .

**Definition 50.** Given  $\ell \in \mathbb{N}$ ,  $\alpha \in (0,1)$  and  $\delta \in \mathbb{R}$ , the space  $\mathcal{C}^{\ell,\alpha}_{\delta}(M_k(\xi))$  is defined to be the space of functions in  $\mathcal{C}^{\ell,\alpha}_{loc}(M_k)$  for which the following norm is finite

$$\|w\|_{\mathcal{C}^{\ell,\alpha}_{\delta}(M_{k})} := \|w\|_{\mathcal{C}^{\ell,\alpha}(M_{k}(\xi,s_{0}+1,\rho_{0}/2))} + \|w \circ X_{m}\|_{\mathcal{C}^{\ell,\alpha}(B_{\rho_{0}}(0))}$$
$$+ \sup_{s \geqslant s_{0}} e^{-\delta s} \left(\|w \circ X_{t,\xi}\|_{\mathcal{C}^{\ell,\alpha}((s,s+1)\times S^{1})} + \|w \circ X_{b,\xi}\|_{\mathcal{C}^{\ell,\alpha}((-s-1,-s)\times S^{1})}\right)$$

and which are invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane.

We remark that there is no weight on the middle end. In fact we compactify this end and we consider a weighted space of functions defined on a two ended surface. We will perturb the surface  $M_k(\xi)$  by the normal graph of a function  $u \in C^{2,\alpha}_{\delta}$  and the middle end  $E_m$  will be just translated in the vertical direction.

The Jacobi operator. The Jacobi operator about  $M_k(\xi)$  is

$$\mathbb{L}_{M_k(\xi)} := \Delta_{M_k(\xi)} + |A_{M_k(\xi)}|^2$$

where  $|A_{M_k(\xi)}|$  is the norm of the second fundamental form on  $M_k(\xi)$ .

In the parametrization introduced above of the ends the volume forms  $\operatorname{dvol}_{M_k(\xi)}$  can be written as  $\gamma_t \, ds \, d\theta$  and  $\gamma_b \, ds \, d\theta$  near the catenoidal type ends and as  $\gamma_m \, dx_1 \, dx_2$  near the middle end. Now we can define globally on  $M_k(\xi)$  a smooth function

$$\gamma: M_k(\xi) \longrightarrow [0,\infty)$$

that is identically equal to 1 on  $M_k(\xi, s_0 - 1, 2\rho_0)$  and equal to  $\gamma_t$  (resp.  $\gamma_b, \gamma_m$ ) on the end  $E_{t,\xi}$  (resp.  $E_{b,\xi}, E_m$ ). Observe that, on  $X_{t,\xi}((s_0, \infty) \times S^1)$  and on  $X_{b,\xi}((-\infty, s_0) \times S^1)$  we have

$$\gamma \circ X_{t,\xi}(s,\theta) \sim \cosh^2 s$$
 and  $\gamma \circ X_{b,\xi}(s,\theta) \sim \cosh^2 s$ .

Finally on  $X_m(B_{\rho_0})$ , we have

 $\gamma \circ X_m(x) \sim |x|^{-4}.$ 

Granted the above defined spaces, one can check that:

$$\mathcal{L}_{\xi,\delta}: \ \mathcal{C}^{2,\alpha}_{\delta}(M_k(\xi)) \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_k(\xi))$$
$$w \longmapsto \gamma \mathbb{L}_{M_k(\xi)}(w)$$

is a bounded linear operator. The subscript  $\delta$  is meant to keep track of the weighted space over which the Jacobi operator is acting. Observe that, the function  $\gamma$  is here to counterbalance the effect of the conformal factor  $\frac{1}{\sqrt{|g_{M_k(\xi)}|}}$  in the expression of the Laplacian in the coordinates we use to parametrize the ends of the surface  $M_k(\xi)$ . This is precisely what is needed to have the operator defined from the space  $C^{2,\alpha}_{\delta}(M_k(\xi))$  into the target space  $C^{0,\alpha}_{\delta}(M_k(\xi))$ .

To have a better grasp of what is going on, let us linearize the nonlinear equation (4.2) at w = 0 we get the expression of the Jacobi operator about the standard catenoid

$$\mathbb{L}_C := \frac{1}{\cosh^2 s} \left( \partial_s^2 + \partial_\theta^2 + \frac{2}{\cosh^2 s} \right).$$

We can observe that the operator  $\cosh^2 s \mathbb{L}_C$  maps the space  $(\cosh s)^{\delta} \mathcal{C}^{2,\alpha}((s_0,\infty) \times S^1)$ into the space  $(\cosh s)^{\delta} \mathcal{C}^{0,\alpha}((s_0,\infty) \times S^1)$ .

Similarly, if we linearize the nonlinear equation (4.1) at u = 0, we obtain the expression of the Jacobi operator about the plane

$$\mathbb{L}_{\mathbb{R}^2} := |x|^4 \,\Delta.$$

Again, the operator  $|x|^{-4} \mathbb{L}_{\mathbb{R}^2} = \Delta$  clearly maps the space  $\mathcal{C}^{2,\alpha}(\bar{B}_{\rho_0})$  into the space  $\mathcal{C}^{0,\alpha}(\bar{B}_{\rho_0})$ . Now, the function  $\gamma$  plays, for the ends of the surface  $M_k(\xi)$ , the role played by the function  $\cosh^2 s$  for the ends of the standard catenoid and the role played by the function  $|x|^{-4}$  for the plane. Since the Jacobi operator about  $M_k(\xi)$  is asymptotic to  $\mathbb{L}_{\mathbb{R}^2}$  at  $E_m$  and is asymptotic to  $\mathbb{L}_C$  at  $E_t$  and  $E_b$ , we conclude that the operator  $\mathcal{L}_{\xi,\delta}$  maps  $\mathcal{C}^{2,\alpha}_{\delta}(M_k(\xi))$  into  $\mathcal{C}^{0,\alpha}_{\delta}(M_k(\xi))$ .

We recall the notion of non degeneracy introduced in [11]:

**Definition 51.** The surface  $M_k(\xi)$  is said to be non degenerate if  $\mathcal{L}_{\xi,\delta}$  is injective for all  $\delta < -1$ .

It useful to observe that a duality argument in the weighted Lebesgue spaces, implies that

 $(\mathcal{L}_{\xi,\delta} \text{ is injective}) \Leftrightarrow (\mathcal{L}_{\xi,-\delta} \text{ is surjective})$ 

if  $\delta \notin \mathbb{Z}$ . See [27] and [18] for more details.

The non degeneracy of  $M_k(\xi)$  follows from the study of the kernel of  $\mathcal{L}_{\xi,\delta}$ .

The Jacobi fields. It is known that a smooth one parameter group of isometries containing the identity generates a Jacobi field, that is a solution of the equation  $\mathbb{L}_{M_k(\xi)} u = 0$ . These solutions are generated by the following one parameter groups of isometries: the vertical translations, the translations along the  $x_1$ -axis, the dilations. We refer [11] for details.

The group of vertical translations generated by the Killing vector field  $\Xi(p) = e_3$  gives rise to the Jacobi field

$$\Phi^{0,+}(p) := n(p) \cdot e_3.$$

The vector field  $\Xi(p) = p$  that is associated to the one parameter group of dilation generates a Jacobi fields

$$\Phi^{0,-}(p) := n(p) \cdot p$$

The Killing vector field  $\Xi(p) = e_1$  that generates the group of translations along the  $x_1$ -axis is associated to a Jacobi field

$$\Phi^{1,+}(p) := n(p) \cdot e_1$$

Finally, we denote by

$$\Phi^{1,-}(p) := n(p) \cdot (e_2 \times p)$$

the Jacobi field associated to the Killing vector field  $\Xi(p) = e_2 \times p$  that generates the group of rotations about the  $x_2$ -axis.

The Jacobi equation has other solutions which are not taken into account because in the difference with the four Jacobi fields just introduced they are not invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane.

With these notations, we define the deficiency space

$$\mathcal{D} := \text{Span}\{\chi_t \, \Phi^{j,\pm}, \chi_b \, \Phi^{j,\pm} : j = 0, 1\}$$

where  $\chi_t$  is a cutoff function that is identically equal to 1 on  $X_t((s_0+1,\infty)\times S^1)$ , identically equal to 0 on  $M_k - X_t((s_0,\infty)\times S^1)$  and that is invariant under the action of the symmetry with respect to the  $x_2 = 0$  plane. Also, we agree that

$$\chi_b(\cdot) := \chi_t(-\cdot).$$

Clearly

$$\tilde{\mathcal{L}}_{\xi,\delta}: \quad \mathcal{C}^{2,\alpha}_{\delta}(M_k(\xi)) \oplus \mathcal{D} \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_k) \\
w \longmapsto \gamma \mathbb{L}_{M_k(\xi)}(w)$$

is a bounded linear operator. The linear decomposition Lemma proved in [23] for constant mean curvature surfaces (see also [18] for minimal hypersurfaces) can be adapted to our situation and thanks to a result of S. Nayatani contained in [29],[30] and extended in chapter 1, which states that any bounded Jacobi field respecting the mirror symmetry with respect to the  $x_2 = 0$  plane is linear combination of  $\Phi^{0,\pm}$  and  $\Phi^{1,+}$ , we get the following result

**Proposition 52.** Assume that all bounded Jacobi fields on  $M_k$  that are invariant with respect to the  $x_2 = 0$  plane are linear combination of  $\Phi^{0,\pm}$  and  $\Phi^{1,+}$  and choose  $\delta \in (1,2)$ . Then (reducing  $\xi_0$  if this is necessary) the operator  $\tilde{\mathcal{L}}_{\xi,\delta}$  is surjective and has a kernel of dimension 4. Moreover, there exists  $G_{\xi,\delta}$  a right inverse for  $\tilde{\mathcal{L}}_{\xi,\delta}$  that depends smoothly on  $\xi$  and in particular whose norm is bounded uniformly as  $|\xi| < \xi_0$ .

### 4.3 An infinite dimensional family of minimal surfaces which are close to $M_k(\xi)$

In this section we consider a truncature of  $M_k(\xi)$ . First we recall a result of [11] that describes the region of the surface which can be parametrized by a graph on a  $x_3 = 0$  plane.

**Lemma 53** ([11]). There exists  $\epsilon_0 > 0$  such that, for all  $\epsilon \in (0, \epsilon_0)$  and all  $|\xi| \leq \epsilon$  an annular part of the ends  $E_t(\xi)$ ,  $E_b(\xi)$  and  $E_m$  of  $M_k(\xi)$  can be written as vertical graphs over the horizontal plane for the functions

$$U_t(r,\theta) = \sigma_{t,\xi} + \ln(2r) + \xi r \cos \theta + \mathcal{O}(\epsilon),$$
  

$$U_b(r,\theta) = -\sigma_{b,\xi} - \ln(2r) - \xi r \cos \theta + \mathcal{O}(\epsilon),$$
  

$$U_m(r,\theta) = \mathcal{O}(r^{-(k+1)}).$$

Here  $(r, \theta)$  are the polar coordinates in the  $x_3 = 0$  plane. The functions  $\mathcal{O}(\epsilon)$  are defined in the annulus  $B_{4\epsilon^{-1/2}} - B_{\epsilon^{-1/2}/4}$  and are bounded in  $\mathcal{C}_b^{\infty}$  topology by a constant (independent on  $\epsilon$ ) multiplied by  $\epsilon$ , where the partial derivatives are computed with respect to the vector fields  $r \partial_r$  and  $\partial_{\theta}$ .

Then  $M_k(\epsilon/2)$  has two ends  $E_t(\epsilon/2)$  and  $E_b(\epsilon/2)$  which are graphs over the  $x_3 = 0$ plane for functions  $U_t$  and  $U_b$  defined on the annulus  $B_{4\epsilon^{-1/2}} - B_{\epsilon^{-1/2}/4}$ . We set  $s_{\epsilon} = -\frac{1}{2} \ln \epsilon$ and we define  $M_k^T(\epsilon/2)$  to be equal to  $M_k(\epsilon/2)$  from which we have removed the image of  $(s_{\epsilon}, +\infty) \times S^1$  by  $X_{t,\epsilon/2}$ , the image of  $(-\infty, -s_{\epsilon}) \times S^1$  by  $X_{b,\epsilon/2}$  and the image of  $B_{\rho_{\epsilon}}(0)$  by  $X_m$  with  $\rho_{\epsilon} := 2\epsilon^{1/2}$ . In this section we will prove the existence of a family of surfaces close to  $M_k^T(\epsilon/2)$ . We follow the work [11].

First, we modify the parametrization of the end  $E_t(\epsilon/2)$ ,  $E_b(\epsilon/2)$  and  $E_m$ , for appropriates values of s, so that, when  $r \in [\epsilon^{-1/2}/4, 4\epsilon^{-1/2}]$  the curves corresponding to the image of

 $\theta \to (r \cos \theta, r \sin \theta, U_t(r, \theta)), \quad \theta \to (r \cos \theta, r \sin \theta, U_b(r, \theta))$ 

correspond to the curve  $s = \pm \log(2r)$ .

The curve  $\theta \to (r \cos \theta, r \sin \theta, U_m(r, \theta))$  corresponds to  $\rho = \frac{1}{r}$ .

The second step is the modification of unit normal vector field on  $M_k(\epsilon/2)$  into a transverse unit vector field  $\tilde{n}_{\epsilon/2}$  in such a way that it coincides with the normal vector field  $n_{\epsilon/2}$  on  $M_k(\epsilon/2)$ , is equal to  $e_3$  on the graph over  $B_{2\epsilon^{-1/2}} - B_{3\epsilon^{-1/2}/8}$  of the functions  $U_t$  and  $U_b$ and interpolate smoothly between the different definitions of  $\tilde{n}_{\epsilon/2}$  in different subsets of  $M_k^T(\epsilon/2)$ .

The graph of a function u, using the vector field  $\tilde{n}_{\epsilon/2}$ , will be a minimal surface if and only if u is a solution of a second order nonlinear elliptic equation of the form

$$\mathbb{L}_{M_{k}^{T}(\epsilon/2)} u = \tilde{L}_{\epsilon/2} u + Q_{\epsilon} (u)$$

where  $\mathbb{L}_{M_k^T(\epsilon/2)}$  is the Jacobi operator about  $M_k^T(\epsilon/2)$ ,  $Q_{\epsilon}$  is a nonlinear second order differential operator and  $\tilde{L}_{\epsilon/2}$  is a linear operator which takes into account the change of the parametrization and of the change of the normal vector field. It is possible to prove that this last operator has coefficients uniformly bounded by a constant times  $\epsilon^2$ .

Now, we consider three functions  $\varphi_t, \varphi_b, \varphi_m \in \mathcal{C}^{2,\alpha}(S^1)$  which are even, with respect to  $\theta$ ,  $\varphi_t, \varphi_b$  are  $L^2$  orthogonal to 1 and  $\cos \theta$  while  $\varphi_m$  is  $L^2$  orthogonal to 1. Assume that they satisfy

$$\|\varphi_t\|_{\mathcal{C}^{2,\alpha}} + \|\varphi_b\|_{\mathcal{C}^{2,\alpha}} + \|\varphi_m\|_{\mathcal{C}^{2,\alpha}} \leqslant \kappa \epsilon$$

We set  $\Phi := (\varphi_t, \varphi_b, \varphi_m)$  and we define  $w_{\Phi}$  to be the function equal to

- 1.  $\chi_+ H_{\varphi_t}(s_{\epsilon} s, \cdot)$  on the image of  $X_{t,\epsilon/2}$  where  $\chi_+$  is a cut-off function equal to 0 for  $s \leq s_0 + 1$  and identically equal to 1 for  $s \in [s_0 + 2, s_{\epsilon}]$
- 2.  $\chi_{-} H_{\varphi_{b}}(s s_{\epsilon}, \cdot)$  on the image  $X_{b,\epsilon/2}$  where  $\chi_{-}$  is a cut-off function equal to 0 for  $s \ge -s_{0} 1$  and identically equal to 1 for  $s \in [-s_{\epsilon}, -s_{0} 2]$
- 3.  $\chi_m \tilde{H}_{\rho_{\epsilon},\varphi_m}(\cdot, \cdot)$  on the image of  $X_m$ , where  $\chi_m$  is a cut-off function equal to 0 for  $r \ge \rho_0$  and identically equal to 1 for  $\rho \in [\rho_{\epsilon}, \rho_0/2]$

4. zero on the remaining part of the surface  $M_k^T(\epsilon/2)$ .

We recall that the operators  $\tilde{H}$  and H have been introduced respectively in Propositions 79 and 80.

We would like to prove that, under appropriates hypothesis, the graph about  $M_k^T(\epsilon/2)$  of the function  $u = w_{\Phi} + v$  is a minimal surface. This is equivalent to solve the equation:

$$\mathcal{L}_{M_k(\epsilon/2)}(w_{\Phi} + v) = \hat{L}_{\epsilon/2}(w_{\Phi} + v) + Q_{\epsilon}(w_{\Phi} + v)$$

on  $M_k^T(\epsilon/2)$ , so that the graph of  $u = w_{\Phi} + v$  will be a minimal surface. The resolution of the previous equation is obtained thanks to the one of the following fixed point problem:

$$v = T(\Phi, v) \tag{4.7}$$

with

$$T(\Phi, v) = G_{\epsilon/2,\delta} \circ \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon/2}(w_{\Phi} + v) - \mathbb{L}_{M_{k}^{T}(\epsilon/2)} w_{\Phi} + Q_{\epsilon}(w_{\Phi} + v) \right) \right)$$

where  $\delta \in (1, 2)$ , the operator  $G_{\epsilon/2,\delta}$  is defined in proposition 52 and  $\mathcal{E}_{\epsilon}$  is a linear extension operator such that

$$\mathcal{E}_{\epsilon}: \mathcal{C}^{0,\alpha}_{\delta}(M_k^T(\epsilon/2)) \longrightarrow \mathcal{C}^{0,\alpha}_{\delta}(M_k(\epsilon/2)),$$

where  $\mathcal{C}^{0,\alpha}_{\delta}(M_k^T(\epsilon/2))$  denotes the space of functions of  $\mathcal{C}^{0,\alpha}_{\delta}(M_k(\epsilon/2))$  restricted to  $M_k^T(\epsilon/2)$ . It is defined by  $\mathcal{E}_{\epsilon}v = v$  in  $M_k^T(\epsilon/2)$ ,  $\mathcal{E}_{\epsilon}v = 0$  in the image of  $[s_{\epsilon}+1, +\infty) \times S^1$  by  $X_{t,\epsilon/2}$ , in the image of  $(-\infty, -s_{\epsilon}-1) \times S^1$  by  $X_{b,\epsilon/2}$  and in the image of  $B_{\rho_{\epsilon}/2} \times S^1$  by  $X_m$ . Finally  $\mathcal{E}_{\epsilon}v$  is an interpolation of these values in the remaining part of  $M_k(\epsilon/2)$  such that, for example,

$$\begin{aligned} (\mathcal{E}_{\epsilon}v) \circ X_{t,\epsilon/2}(s,\theta) &= ((1+s_{\epsilon}-s)v) \circ X_{t,\epsilon/2}(s_{\epsilon},\theta), \\ (\mathcal{E}_{\epsilon}v) \circ X_{b,\epsilon/2}(s,\theta) &= ((1+s_{\epsilon}+s)v) \circ X_{b,\epsilon/2}(s_{\epsilon},\theta) \end{aligned}$$

for  $(s, \theta) \in [s_{\epsilon}, s_{\epsilon} + 1] \times S^1$  and

$$(\mathcal{E}_{\epsilon}v) \circ X_m(\rho,\theta) = ((\frac{2}{\rho_{\epsilon}}\rho - 1)v) \circ X_m(\rho_{\epsilon},\theta)$$

for  $(\rho, \theta) \in [\rho_{\epsilon}/2, \rho_{\epsilon}] \times S^1$ .

**Remark 54.** As consequence of the properties of  $\mathcal{E}_{\epsilon}$ , if  $supp v \cap B_{\rho_0} - B_{\rho_{\epsilon}} \neq \emptyset$  then

$$\|(\mathcal{E}_{\epsilon}v)\circ X_m\|_{\mathcal{C}^{0,\alpha}(B_{\rho_0})}\leqslant c\epsilon^{-\alpha}\|v\circ X_m\|_{\mathcal{C}^{0,\alpha}(B_{\rho_0})}.$$

This phenomenon of explosion of the norm does not occur near the catenoidal type ends:

$$\|(\mathcal{E}_{\epsilon}v) \circ X_{t,\epsilon/2}\|_{\mathcal{C}^{0,\alpha}([s_{\epsilon}+1,+\infty)\times S^1)} \leqslant c \|v \circ X_{t,\epsilon/2}\|_{\mathcal{C}^{0,\alpha}([s_{\epsilon}+1,+\infty)\times S^1)}$$

A similar equation holds for the bottom end.

In the following we will assume  $\alpha > 0$  and near to zero.

The existence of a solution  $v \in C^{2,\alpha}_{\delta}(M_k^T(\epsilon/2))$  for the equation (4.7) is a consequence of the following result which proves that T is a contracting mapping.

**Lemma 55.** There exist constants  $c_{\kappa} > 0$  and  $\epsilon_{\kappa} > 0$ , such that

$$\|T(\Phi,0)\|_{\mathcal{C}^{2,\alpha}_{s}} \leqslant c_{\kappa} \,\epsilon^{2} \tag{4.8}$$

and, for all  $\epsilon \in (0, \epsilon_{\kappa})$ 

$$\|T(\Phi, v_2) - T(\Phi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_k(\epsilon/2))} \leq \frac{1}{2} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_k^T(\epsilon/2))}$$

for all  $v_1, v_2 \in \mathcal{C}^{2, \alpha}_{\delta}(M_k^T(\epsilon/2))$  and satisfying  $\|v\|_{\mathcal{C}^{2, \alpha}_{\delta}} \leqslant 2 c_{\kappa} \epsilon^2$ .

**Proof.** We recall that the Jacobi operator associated to  $M_k(\epsilon/2)$ , is asymptotic to the operator of the catenoid near the catenoidal ends, and it is asymptotic to the laplacian near of the planar end. The function  $w_{\Phi}$  is identically zero far from the ends where the explicit expression of  $\mathbb{L}_{M_k(\epsilon/2)}$  is not known: this is the reason of our particular choice in the definition of  $w_{\Phi}$ . Then from the definition of  $w_{\Phi}$  and thanks to proposition 52 we obtain the estimate

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma \mathbb{L}_{M_{k}^{T}(\epsilon/2)} w_{\Phi}\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}(M_{k}(\epsilon/2))} &\leqslant \left|\left|\gamma_{t} \mathbb{L}_{C}(w_{\Phi} \circ X_{t,\epsilon/2})\right|\right|_{\mathcal{C}_{\delta}^{0,\alpha}([s_{0}+1,s_{\epsilon}]\times S^{1})} + \\ + \left|\left|\gamma_{b} \mathbb{L}_{C}(w_{\Phi} \circ X_{b,\epsilon/2})\right|\right|_{\mathcal{C}_{\delta}^{0,\alpha}([-s_{\epsilon},-s_{0}-1]\times S^{1})} &\leqslant c \left|\left|\frac{2}{\cosh^{2}s} w_{\Phi} \circ X_{t,\epsilon/2}\right|\right|_{\mathcal{C}_{\delta}^{0,\alpha}([s_{0}+1,s_{\epsilon}]\times S^{1})} \\ &\qquad c\epsilon \|w_{\Phi} \circ X_{t,\epsilon/2}\|_{\mathcal{C}_{\delta}^{2,\alpha}([s_{0}+1,s_{\epsilon}]\times S^{1})} \leqslant c_{\kappa} \epsilon^{2+\frac{\delta}{2}}. \end{aligned}$$

Using the properties of  $\tilde{L}_{\epsilon/2}$ , we obtain

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma\tilde{L}_{\epsilon/2}w_{\Phi}\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} &\leqslant c\epsilon^{2}\|w_{\Phi}\circ X_{t,\epsilon/2}\|_{\mathcal{C}^{0,\alpha}_{\delta}([s_{0}+1,s_{\epsilon}]\times S^{1})} + c\epsilon^{2}\|w_{\Phi}\circ X_{m}\|_{\mathcal{C}^{0,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})} &\leqslant \\ &\leqslant c\epsilon^{2}\|w_{\Phi}\circ X_{t,\epsilon/2}\|_{\mathcal{C}^{2,\alpha}_{\delta}([s_{0}+1,s_{\epsilon}]\times S^{1})} + c\epsilon^{2}\|w_{\Phi}\circ X_{m}\|_{\mathcal{C}^{2,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})} &\leqslant c_{\kappa}\epsilon^{2}. \end{aligned}$$

As for the last term, we recall that the operator  $Q_{\epsilon}$  has two different expressions if we consider the catenoidal type end and the planar end. It holds that

$$\left\| \mathcal{E}_{\epsilon} \left( \gamma Q_{\epsilon} \left( w_{\Phi} \right) \right) \right\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} \leqslant c_{k} \epsilon^{2}.$$

In fact

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma Q_{\epsilon}\left(w_{\Phi}\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} \leqslant c\epsilon \|w_{\Phi} \circ X_{t,\epsilon/2}\|_{\mathcal{C}^{2,\alpha}_{\delta}([s_{0}+1,s_{\epsilon}]\times S^{1})}^{2} + c\epsilon \|w_{\Phi} \circ X_{b,\epsilon/2}\|_{\mathcal{C}^{2,\alpha}_{\delta}([-s_{\epsilon},-s_{0}-1]\times S^{1})}^{2} + c\epsilon^{1-\alpha}\|w_{\Phi} \circ X_{m}\|_{\mathcal{C}^{2,\alpha}([\rho_{\epsilon},\rho_{0}/2]\times S^{1})}^{2} \leqslant c_{k}\epsilon^{2} \end{aligned}$$

As for the second estimate, we recall that

$$T(\Phi, v) := G_{\epsilon/2,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon/2}(w_{\Phi} + v) - \mathbb{L}_{M_{k}^{T}(\epsilon/2)} w_{\Phi} + Q_{\epsilon} \left( w_{\Phi} + v \right) \right) \right) \right).$$

Then

$$T(\Phi, v_2) - T(\Phi, v_1) = G_{\epsilon/2,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon/2} (w_{\Phi} + v_2) - \mathbb{L}_{M_k^T(\epsilon/2)} w_{\Phi} + Q_{\epsilon} (w_{\Phi} + v_2) \right) \right) \right) - G_{\epsilon/2,\delta} \left( \mathcal{E}_{\epsilon} \left( \gamma \left( \tilde{L}_{\epsilon/2} (w_{\Phi} + v_1) - \mathbb{L}_{M_k^T(\epsilon/2)} w_{\Phi} + Q_{\epsilon} (w_{\Phi} + v_1) \right) \right) \right) \right)$$

and

$$\begin{aligned} \|T(\Phi, v_{2}) - T(\Phi, v_{1})\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_{k}(\epsilon/2))} &\leq c \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}_{\epsilon/2}(w_{\Phi} + v_{2}) - \mathbb{L}_{M_{k}^{T}(\epsilon/2)} w_{\Phi} + Q_{\epsilon} \left(w_{\Phi} + v_{2}\right) - \tilde{L}_{\epsilon/2}(w_{\Phi} + v_{1}) + \mathbb{L}_{M_{k}^{T}(\epsilon/2)} w_{\Phi} - Q_{\epsilon} \left(w_{\Phi} + v_{1}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} = \\ \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}_{\epsilon/2}(w_{\Phi} + v_{2}) - \tilde{L}_{\epsilon/2}(w_{\Phi} + v_{1}) + Q_{\epsilon} \left(w_{\Phi} + v_{2}\right) - Q_{\epsilon} \left(w_{\Phi} + v_{1}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} = \\ &= \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}_{\epsilon/2}(v_{2} - v_{1}) + Q_{\epsilon} \left(w_{\Phi} + v_{2}\right) - Q_{\epsilon} \left(w_{\Phi} + v_{1}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} \leq \\ &\leq \|\mathcal{E}_{\epsilon} \left(\gamma \left(\tilde{L}(v_{2} - v_{1})\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))} + \|\mathcal{E}_{\epsilon} \left(\gamma \left(Q_{\epsilon} \left(w_{\Phi} + v_{1}\right) - Q_{\epsilon} \left(w_{\Phi} + v_{2}\right)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_{k}(\epsilon/2))}. \end{aligned}$$

We observe that from the considerations above it follows that

$$\|\mathcal{E}_{\epsilon}\left(\gamma\left(\tilde{L}_{\epsilon/2}(v_2-v_1)\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\delta}(M_k(\epsilon/2))} \leqslant c\epsilon^2 \|v_2-v_1\|_{\mathcal{C}^{2,\alpha}_{\delta}(M_k^T(\epsilon/2))}$$

and

$$\begin{aligned} \|\mathcal{E}_{\epsilon}\left(\gamma\left(Q_{\epsilon}\left(w_{\Phi}+v_{1}\right)-Q_{\epsilon}\left(w_{\Phi}+v_{2}\right)\right)\right)\|_{\mathcal{C}_{\delta}^{0,\alpha}\left(M_{k}\left(\epsilon/2\right)\right)} \\ &\leqslant c||v_{2}-v_{1}||_{\mathcal{C}_{\delta}^{2,\alpha}\left(M_{k}^{T}\left(\epsilon/2\right)\right)}\|w_{\Phi}\|_{\mathcal{C}_{\delta}^{0,\alpha}\left(M_{k}\left(\epsilon/2\right)\right)} \\ &\leqslant \left(c_{k}\epsilon^{1+\frac{\delta}{2}}\right)||v_{2}-v_{1}||_{\mathcal{C}_{\delta}^{2,\alpha}\left(M_{k}^{T}\left(\epsilon/2\right)\right)}. \end{aligned}$$

Then

$$||T(\Phi, v_2) - T(\Phi, v_1)||_{\mathcal{C}^{2,\alpha}_{\delta}(M_k(\epsilon/2))} \leq c\epsilon ||v_2 - v_1||_{\mathcal{C}^{2,\alpha}_{\delta}(M_k^T(\epsilon/2))}.$$

This argument provides a minimal surface  $M_k^T(\epsilon/2, \Phi)$  which is close to  $M_k^T(\epsilon/2)$  and has three boundaries. This surface is, close to its upper and lower boundary, a vertical graph over the annulus  $B_{\epsilon^{-1/2}/2} - B_{\epsilon^{-1/2}/4}$  whose parametrization is, respectively, given by

$$\bar{U}_t(r,\theta) = \sigma_{t,\epsilon/2} + \ln(2r) + \frac{\epsilon}{2}r\cos\theta + H_{\varphi_t}(s_\epsilon - \ln 2r,\theta) + V_t(r,\theta),$$

$$\bar{U}_b(r,\theta) = -\sigma_{b,\epsilon/2} - \ln(2r) + \frac{\epsilon}{2}r\cos\theta + H_{\varphi_b}(\ln 2r - s_\epsilon,\theta) + V_b(r,\theta)$$

where  $s_{\epsilon} = -\frac{1}{2} \ln \epsilon$ . The boundaries of the surface correspond to  $r_{\epsilon} = \frac{1}{2} \epsilon^{-1/2}$ . Nearby the middle boundary the surface is a vertical graph over the annulus  $B_{2\rho_{\epsilon}} - B_{\rho_{\epsilon}/2}$ , where  $\rho_{\epsilon} = 2\epsilon^{1/2}$ . Its parametrization is

$$\bar{U}_m(r,\theta) = \tilde{H}_{\rho_\epsilon,\varphi_m}(r,\theta) + V_m(r,\theta)$$

All the functions  $V_i$  for i = t, b, m depend non linearly on  $\epsilon, \varphi$ . The functions  $V_i(\epsilon, \varphi_i)$ , for i = t, b, satisfy  $\|V_i(\epsilon, \varphi_i)(r_{\epsilon})\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leq c\epsilon$  and

$$\|V_i(\epsilon,\varphi)(r_{\epsilon}\cdot) - V_i(\epsilon,\varphi')(r_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leq c\epsilon^{1-\delta/2} \|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}}$$

The function  $V_m(\epsilon, \varphi_i)$  satisfies  $\|V_m(\epsilon, \varphi_m)(\rho_{\epsilon})\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)} \leq c\epsilon$  and

$$\|V_m(\epsilon,\varphi)(\rho_{\epsilon}\cdot) - V_m(\epsilon,\varphi')(\rho_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2 - B_1)} \leq c\epsilon \|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}}$$

#### 4.4 KMR examples $M_{\sigma,\alpha,\beta}$

In 1988, H. Karcher [19, 20] defined a family of doubly periodic minimal surfaces, called toroidal halfplane layers, with genus one and four horizontal Scherk-type ends<sup>2</sup> in the quotient. In 1989, Meeks and Rosenberg [25] developed a general theory for doubly periodic minimal surfaces having finite topology in the quotient, and used an approach of minimax type to obtain the existence of a family of doubly periodic minimal surfaces, also with genus one and four horizontal Scherk-type ends in the quotient. These Karcher's and Meeks and Rosenberg's surfaces have been generalized in [35], constructing a 3-parameter family  $\mathcal{K} = \{M_{\sigma,\alpha,\beta}\}_{\sigma,\alpha,\beta}$  of surfaces, called KMR examples (sometimes, they are also referred in the literature as toroidal halfplane layers). Such examples have been classified by Pérez, Rodríguez and Traizet [33] as the only doubly periodic minimal surfaces with genus one and finitely many parallel (Scherk-type) ends in the quotient. The possible limits of KMR examples are: the catenoid, the helicoid, any singly or doubly periodic Scherk minimal surface, any Riemann minimal example or another KMR example. Our aim in this section is to study two subfamilies of KMR examples which are invariants under a reflection symmetry about a vertical plane, near the catenoidal limit.

Firstly, we briefly recall the construction of the KMR examples. For each  $\sigma \in (0, \frac{\pi}{2})$ ,  $\alpha \in [0, \frac{\pi}{2}]$  and  $\beta \in [0, \frac{\pi}{2}]$  with  $(\alpha, \beta) \neq (0, \sigma)$ , consider the rectangular torus  $\Sigma_{\sigma} = \left\{ (z, w) \in \overline{\mathbb{C}}^2 \mid w^2 = (z^2 + \lambda^2)(z^2 + \lambda^{-2}) \right\}$ , where  $\lambda = \lambda(\sigma) = \cot \frac{\sigma}{2} > 1$ . The KMR

 $<sup>^{2}</sup>$ A horizontal *Scherk-type* end is an end asymptotic to a horizontal half-plane, invariant by one of the period vectors of the surface.

#### Figure 4.1: The position of the branch values

example  $M_{\sigma,\alpha,\beta}$  is determined by its Gauss map g and the differential of its height function h, which are defined on  $\Sigma_{\sigma}$  and given by:

$$g(z,w) = \frac{az+b}{i(\overline{a}-\overline{b}z)}, \qquad dh = \mu \, \frac{dz}{w},$$

where:

•  $a = a(\alpha, \beta) = \cos \frac{\alpha + \beta}{2} + i \cos \frac{\alpha - \beta}{2};$ 

• 
$$b = b(\alpha, \beta) = \sin \frac{\alpha - \beta}{2} + i \sin \frac{\alpha + \beta}{2};$$

•  $\mu = \mu(\sigma) = \frac{\pi \csc \sigma}{\mathcal{K}(\sin^2 \sigma)}$ , where  $\mathcal{K}(m) = \int_0^{\frac{\pi}{2}} \frac{1}{\sqrt{1 - m \sin^2 u}} du$ , 0 < m < 1, is the complete elliptic integral of the first kind. Such  $\mu$  has been chosen so that the vertical part of the flux of  $M_{\sigma,\alpha,\beta}$  along any horizontal level section equals  $2\pi$ .

#### Remark 56.

- (i)  $b \to 0$  if and only if  $\alpha \to 0$  and  $\beta \to 0$ , in which case  $a \to 1+i$ .
- (ii)  $\left|\frac{b}{a}\right| = \tan \frac{\varphi}{2}$ , where  $\varphi$  is the angle between the North Pole  $(0,0,1) \in \mathbb{S}^2$  and the pole of g seen in  $\mathbb{S}^2$  via the inverse of the stereographic projection.

The KMR example  $M_{\sigma,\alpha,\beta}$  can be parametrized on  $\Sigma_{\sigma}$  by the immersion  $X = (X_1, X_2, X_3) = \Re \int \Phi$ , where  $\Phi$  is the Weierstrass form:

$$\Phi = \left(\frac{1}{2}\left(\frac{1}{g} - g\right)dh, \frac{i}{2}\left(\frac{1}{g} + g\right)dh, dh\right).$$

The ends of  $M_{\sigma,\alpha,\beta}$  corresponds to the punctures  $\{A, A', A'', A'''\} = g^{-1}(\{0, \infty\})$ , and the branch values of g are those with w = 0, i.e.

$$D = (-i\lambda, 0), \ D' = (i\lambda, 0), \ D'' = (\frac{i}{\lambda}, 0), \ D''' = (-\frac{i}{\lambda}, 0).$$
(4.9)

Seen in  $\mathbb{S}^2$ , these points form two pairs of antipodal points: D'' = -D and D''' = -D'. (Each KMR example can be given in terms of the branch values of its Gauss map.)

In [35], it is proven that the above Weierstrass data define a properly embedded minimal surface  $(M_{\sigma,\alpha,\beta})$  invariant by two independent translations: the translation by the period  $T_1$  at its ends, and the period  $T_2$  along a homology class. Moreover, the group of isometries  $\text{Iso}(M_{\sigma,\alpha,\beta})$  of  $M_{\sigma,\alpha,\beta}$  always contains a subgroup isomorphic to  $(\mathbb{Z}/2\mathbb{Z})^2$ , Figure 4.2: Left:  $M_{\sigma,0,0}$ , with  $\sigma = \frac{\pi}{4}$ . Right:  $M_{\sigma,\alpha,0}$  for  $\sigma = \alpha = \frac{\pi}{4}$ .

Figure 4.3: 
$$M_{\sigma,0,\beta}$$
, where  $\sigma = \frac{\pi}{4}$  and  $\beta = \frac{\pi}{8}$ .

with generators  $\mathcal{D}$  (corresponding to the deck transformation  $(z, w) \mapsto (z, -w)$ ), which represents a central symmetry about any of the four branch points of g, and  $\mathcal{F}$ , which consists of a translation by  $\frac{1}{2}(T_1+T_2)$ . In particular, the ends of  $M_{\sigma,\alpha,\beta}$  are equally spaced.

We are going to focus on the two most symmetric subfamilies of KMR examples:  $\{M_{\sigma,\alpha,0}\}_{\sigma,\alpha}$  and  $\{M_{\sigma,0,\beta}\}_{\sigma,\beta}$ .

- 1. When  $\alpha = \beta = 0$ ,  $M_{\sigma,0,0}$  contains four straight lines parallel to the  $x_1$ -axis, and Iso $(M_{\sigma,0,0})$  is isomorphic to  $(\mathbb{Z}/2\mathbb{Z})^4$  with generators  $S_1, S_2, S_3, R_D$ :  $S_1$  is a reflection symmetry in a vertical plane orthogonal to the  $x_1$ -axis;  $S_2$  is a reflection symmetry across a plane orthogonal to the  $x_2$ -axis;  $S_3$  is a reflection symmetries in a horizontal plane (these three planes can be chosen meeting at a point, which is not contained in the surface); and  $R_D$  is the  $\pi$ -rotation around one of the four straight lines contained in the surface, see Figure 4.2 left. In this case,  $T_1 = (0, \pi\mu, 0)$ .
- 2. When  $0 < \alpha < \frac{\pi}{2}$ ,  $\operatorname{Iso}(M_{\sigma,\alpha,0})$  is isomorphic to  $(\mathbb{Z}/2\mathbb{Z})^3$ , with generators  $\mathcal{D}$ ,  $S_2$  and  $R_2$ , where  $S_2$  represents a reflection symmetry across a plane orthogonal to the  $x_2$ -axis, and  $R_2$  is a  $\pi$ -rotation around a line parallel to the  $x_2$ -axis that cuts  $M_{\sigma,\alpha,0}$  orthogonally, see Figure 4.2 right. Now  $T_1 = (0, \pi \mu t_\alpha, 0)$ , with  $t_\alpha = \frac{\sin \sigma}{\sqrt{\sin^2 \sigma \cos^2 \alpha + \sin^2 \alpha}}$ .
- 3. Suppose that  $0 < \beta < \sigma$ . Then  $M_{\sigma,0,\beta}$  contains four straight lines parallel to the  $x_1$ -axis, and  $\operatorname{Iso}(M_{\sigma,0,\beta})$  is isomorphic to  $(\mathbb{Z}/2\mathbb{Z})^3$ , with generators  $S_1$ ,  $R_1$  and  $R_D$ :  $S_1$  represents a reflection symmetry across a plane orthogonal to the  $x_1$ -axis;  $R_1$  corresponds to a  $\pi$ -rotation around a line parallel to the  $x_1$ -axis that cuts the surface orthogonally; and  $R_D$  is the  $\pi$ -rotation around any one of the straight lines contained in the surface, see Figure 4.3. Moreover,  $T_1 = (0, \pi \mu t^\beta, 0)$ , where  $t^\beta = \frac{\sin \sigma}{\sqrt{\sin^2 \sigma \sin^2 \beta}}$ .

From now on, we will denote by  $T = T_1$  the period of the surface at its ends. Finally, it will be useful to see  $\Sigma_{\sigma}$  as a branched 2-covering of  $\overline{\mathbb{C}}$  through the map  $(z, w) \mapsto z$ . Thus  $\Sigma_{\sigma}$  can be seen as two copies  $\overline{\mathbb{C}}_1, \overline{\mathbb{C}}_2$  of  $\overline{\mathbb{C}}$  glued along two common cuts  $\gamma_1, \gamma_2$ , which we tale along the imaginary axis:  $\gamma_1$  from D to D' and  $\gamma_2$  from D'' to D'''.

#### 4.4.1 $M_{\sigma,\alpha,\beta}$ as a graph over $\{x_3 = 0\}/T$

The KMR examples  $M_{\sigma,\alpha,\beta}$  converge, as  $(\sigma,\alpha,\beta) \to (0,0,0)$ , to a vertical catenoid, since  $\Sigma_{\sigma}$  converges to two pinched spheres,  $g(z) \to z$  and  $dh \to \pm \frac{dz}{z}$  as  $\sigma, \alpha, \beta \to 0$ . In fact, we can obtain two catenoids in the limit, depending on the choice of branch for w (for each copy of  $\overline{\mathbb{C}}$  in  $\Sigma_{\sigma}$ , we obtain one catenoid in the limit). Our aim along this paper consists of gluing KMR examples  $M_{\sigma,\alpha,0}$  or  $M_{\sigma,0,\beta}$  near this catenoidal limit, to a convenient compact piece of a deformed Costa-Hoffman-Meeks surface  $M_k(\epsilon/2)$ . In this subsection we express part of  $M_{\sigma,\alpha,\beta}$  as a vertical graph over the  $\{x_3 = 0\}$ -plane when  $\sigma, \alpha, \beta$  are small.

Consider  $M_{\sigma,\alpha,\beta}$  near the catenoidal limit, i.e.  $\sigma, \alpha, \beta$  close to zero. Without lost of generality, we can assume  $dh \sim \frac{dz}{z}$  in  $\overline{\mathbb{C}}_1$ . We are studying the surface in an annulus about one of its ends, say a zero of its Gauss map.

**Lemma 57.** Let  $\sigma, \alpha, \beta$  be small. Up to translations,  $M_{\sigma,\alpha,\beta}$  can be parametrized in the annulus  $\{(z, w) \in \Sigma_{\sigma} \mid z \in \overline{\mathbb{C}}_1, |\frac{b}{a}| < z < \nu\}$  (for  $\nu > |\frac{b}{a}|$  small) as:

$$\begin{cases} X_1 + i X_2 = \frac{-1}{2} \left( z + \frac{1}{\overline{z}} \right) + \frac{i b}{\overline{a}} \ln |z| + \frac{(1+i)\overline{b}}{4\overline{z}^2} + \mathcal{O}(b^2 z^{-3} + \lambda^{-2} z^{-2}) \\ X_3 = \ln |z| + \mathcal{O}(\lambda^{-2} z^{-2}), \end{cases}$$

*Proof.* We have assumed  $dh \sim \frac{dz}{z}$  in the annulus we are working in. More precisely, we have

$$dh = \mu \frac{dz}{\sqrt{(z^2 + \lambda^2)(z^2 + \lambda^{-2})}} = \frac{\mu}{\lambda} \frac{dz}{z\sqrt{(\lambda^{-2} + z^{-2})(z^2 + \lambda^{-2})}} = \frac{\mu}{\lambda} \frac{dz}{z} \frac{1}{\sqrt{1 + \lambda^{-2} z^2 + \lambda^{-2} z^{-2} + \lambda^{-4}}}.$$

Since  $\frac{\mu(\sigma)}{\lambda(\sigma)} = \frac{\pi}{(1+\cos(\sigma))\mathcal{K}(\sin^2\sigma)} = 1 + \mathcal{O}(\sigma^4) = 1 + \mathcal{O}(\lambda^{-4})$ , and  $\frac{1}{\sqrt{1+t}} = 1 + \mathcal{O}(t)$  for t > 0 small, we get

$$dh = \frac{dz}{z} (1 + \mathcal{O}(\lambda^{-4}))(1 + \mathcal{O}(\lambda^{-2}z^2 + \lambda^{-2}z^{-2} + \lambda^{-4})).$$

Since |z| < 1, then both  $\lambda^{-2}|z|^2 < \lambda^{-2}|z|^{-2}$  and  $\lambda^{-4} < \lambda^{-2}|z|^{-2}$ . Therefore,

$$dh = \frac{dz}{z} (1 + \mathcal{O}(\lambda^{-2}z^{-2})).$$

Fix any point  $z_0 \in \overline{\mathbb{C}}_1$ . Thus  $X_1(z) + i X_2(z) = \frac{1}{2} \left( \int_{z_0}^z \frac{dh}{g} - \int_{z_0}^z g \, dh \right)$ . Straightforward computations give us

$$\int_{z_0}^{z} \frac{dh}{g} = \frac{i}{a} \int_{z_0}^{z} \frac{\overline{a} - \overline{b}\omega}{\omega(\omega + b/a)} (1 + \mathcal{O}(\sigma^2 \omega^{-2})) d\omega$$
$$= \frac{i}{a} \left( \frac{|a|^2}{b} \ln \frac{z}{z + b/a} - \overline{b} \ln(z + b/a) \right) + C_1 + \mathcal{O}(\sigma^2 z^{-2} + \sigma^2 b z^{-3}),$$

where  $C_1 = \frac{i}{a} \left( \frac{|a|^2}{b} \ln \frac{z_0 + b/a}{z_0} + \overline{b} \ln(z_0 + b/a) \right)$ ; and  $\int_{z_0}^z g \, dh = -i \int_{z_0}^z \frac{a\omega + b}{\omega(\overline{a} - \overline{b}\omega)} (1 + \mathcal{O}(\sigma^2 \omega^{-2})) d\omega$   $= \frac{-i}{\overline{a}} \left( \frac{|a|^2 + |b|^2}{-\overline{b}} \ln \frac{\overline{a} - \overline{b}z_0}{\overline{a} - \overline{b}z_0} + b \ln \frac{z}{z_0} \right) + \mathcal{O}(\sigma^2 z^{-1} + \sigma^2 b z^{-2}).$ 

For |t| < |z| we have  $\ln \frac{z}{z+t} = \frac{-t}{z} + \frac{t^2}{z^2} + \mathcal{O}(t^2 z^{-3})$  and  $\ln(z+t) = \ln z + \frac{t}{z} + \mathcal{O}(t^3 z^{-2})$ . Hence taking |b/a| < |z| < 1, we obtain:

•  $\int_{z_0}^{z} \frac{dh}{g} = -\frac{i\overline{b}}{a} \ln z - \frac{i(|a|^2 + |b|^2)}{a^2 z} + \frac{ib\overline{a}}{2a^2 z^2} + C_1 + \mathcal{O}(b^2 z^{-3} + \lambda^{-2} z^{-2}),$ •  $\int_{z_0}^{z} g \, dh = -\frac{ib}{\overline{a}} \ln z - \frac{i(|a|^2 + |b|^2)}{\overline{a}^2} z + C_2 + \mathcal{O}(\lambda^{-2} z^{-1}),$ 

where  $C_1, C_2 \in \mathbb{C}$  verify  $\overline{C_1} - C_2 = \frac{z_0(1+|z_0|^2)}{|z_0|^2} + \mathcal{O}(b)$ . Therefore,

$$\begin{aligned} X_1 + i \, X_2 &= \frac{1}{2} \left( \overline{\int_{z_0}^z \frac{dh}{g}} - \int_{z_0}^z g \, dh \right) \\ &= \frac{ib}{\overline{a}} \ln |z| + \frac{i(|a|^2 + |b|^2)}{2\overline{a}^2} \left( z + \frac{1}{\overline{z}} \right) - \frac{i\overline{b}a}{2\overline{a}^2\overline{z}^2} + \frac{z_0(1 + |z_0|^2)}{2|z_0|^2} + \mathcal{O}(b^2 \, z^{-3} + \lambda^{-2} z^{-2}). \end{aligned}$$

Taking into account that  $\frac{i(|a|^2+|b|^2)}{\overline{a}^2} = -1 + \mathcal{O}(b)$  and  $\frac{ia}{2\overline{a}^2} = -\frac{1+i}{4} + \mathcal{O}(b)$ , we have

$$X_1 + i X_2 = \frac{z_0(1 + |z_0|^2)}{2|z_0|^2} + \frac{i b}{\overline{a}} \ln|z| + \frac{-1}{2} \left(z + \frac{1}{\overline{z}}\right) + \frac{(1 + i)\overline{b}}{4\overline{z}^2} + \mathcal{O}(b^2 z^{-3} + \lambda^{-2} z^{-2}).$$

Similarly,  $\int_{z_0}^{z} dh = \ln z - \ln z_0 + \mathcal{O}(\lambda^{-2}z^{-2})$ , hence

$$X_3 = \Re \int_{z_0}^{z} dh = \ln |z| - \ln |z_0| + \mathcal{O}(\lambda^{-2} z^{-2}),$$

which finishes Lemma 57.

**Lemma 58.** Let  $(r, \theta)$  denote the polar coordinates in the  $\{x_3 = 0\}$  plane and define  $\varepsilon = b + \lambda^{-1}$ . Then a piece of  $M_{\sigma,\alpha,\beta}$  can be written as a vertical graph of

$$\widetilde{U}(r,\theta) = -\ln(2r) + r(\eta_1\cos\theta + \eta_2\sin\theta) + \mathcal{O}(\varepsilon),$$

for  $(r, \theta) \in (\frac{1}{4\sqrt{\varepsilon}}, \frac{4}{\sqrt{\varepsilon}}) \times [0, 2\pi)$ , where  $\eta_1 = \Re(b) + \Im(b)$  and  $\eta_2 = \Re(b) - \Im(b)$ .

**Remark 59.** Recall that  $b = \sin \frac{\alpha - \beta}{2} + i \sin \frac{\alpha + \beta}{2}$ . In particular:

• When  $\beta = 0$ , we have  $\eta_1 = 2 \sin \frac{\alpha}{2}$  and  $\eta_2 = 0$ , so

$$\widetilde{U}(r,\theta) = -\ln(2r) + 2\sin\frac{\alpha}{2}r\cos\theta + \mathcal{O}(\varepsilon), \qquad \mathcal{O}(\varepsilon) = \mathcal{O}(\alpha + \sigma).$$

• When  $\alpha = 0$ ,  $\eta_1 = 0$  and  $\eta_2 = 2 \sin \frac{\beta}{2}$ , so

$$\widetilde{U}(r,\theta) = -\ln(2r) + 2\sin\frac{\beta}{2}r\sin\theta + \mathcal{O}(\varepsilon), \qquad \mathcal{O}(\varepsilon) = \mathcal{O}(\beta + \sigma).$$

*Proof.* From Lemma 57, we know that  $(X_1 + iX_2)(z) = \frac{-1}{2} \left( z + \frac{1}{\overline{z}} \right) + A(z)$ , where

$$A(z) = \frac{i \, b}{\overline{a}} \ln |z| + \frac{(1+i)\overline{b}}{4\overline{z}^2} + \mathcal{O}(b^2 \, z^{-3} + \lambda^{-2} z^{-2}).$$

Denote  $z = |z|e^{i\psi}$ . Then  $z + \frac{1}{\overline{z}} = \left(|z| + \frac{1}{|z|}\right)e^{i\psi}$ , and

$$r \cos \theta = X_1 = -\frac{1}{2} \left( |z| + \frac{1}{|z|} \right) \cos \psi + A_1,$$
$$r \sin \theta = X_2 = -\frac{1}{2} \left( |z| + \frac{1}{|z|} \right) \sin \psi + A_2,$$

where  $A_1 = \Re(A)$  and  $A_2 = \Im(A)$ . Therefore,

$$r^{2} = \frac{1}{4} \left( |z| + \frac{1}{|z|} \right)^{2} - \left( |z| + \frac{1}{|z|} \right) \left( A_{1} \cos \psi + A_{2} \sin \psi \right) + A_{1}^{2} + A_{2}^{2}.$$
(4.10)

From (4.10) we deduce:

$$r^{2} = \frac{1}{4} \left( |z| + \frac{1}{|z|} \right)^{2} \left( 1 - \frac{4|z|}{|z|^{2}+1} \left( A_{1} \cos \psi + A_{2} \sin \psi \right) + \frac{4|z|^{2}}{(|z|^{2}+1)^{2}} \left( A_{1}^{2} + A_{2}^{2} \right) \right).$$

When  $|z| = \mathcal{O}(\sqrt{\varepsilon})$ , the functions  $A_i$  are bounded, and we get

$$r = \frac{1}{2} \left( |z| + \frac{1}{|z|} \right) (1 + \mathcal{O}(\sqrt{\varepsilon})), \tag{4.11}$$

and so  $r = \mathcal{O}(1/\sqrt{\varepsilon})$ . Moreover, we get  $\frac{r}{\frac{1}{2}(|z|+\frac{1}{|z|})} = 1 + \mathcal{O}(\sqrt{\varepsilon})$ , from where

$$e^{i\theta}(1 + \mathcal{O}(\sqrt{\varepsilon})) = \frac{X_1 + iX_2}{\frac{1}{2}\left(|z| + \frac{1}{|z|}\right)} = -e^{i\psi} + \frac{2|z|A}{1 + |z|^2} = -e^{i\psi} + \mathcal{O}(\sqrt{\varepsilon})).$$

Hence

$$e^{i\psi} = -e^{i\theta}(1 + \mathcal{O}(\sqrt{\epsilon})).$$

From (4.10) and (4.11) we obtain

$$\frac{(1+|z|^2)^2}{4|z|^2} = r^2 + \left(|z| + \frac{1}{|z|}\right) \left(A_1 \cos \psi + A_2 \sin \psi\right) - A_1^2 - A_2^2$$
$$= r^2 \left(1 + \frac{1}{r} (1 + \mathcal{O}(\sqrt{\varepsilon})) \left(A_1 \cos \psi + A_2 \sin \psi\right) - \frac{A_1^2 + A_2^2}{r^2}\right)$$
$$= r^2 \left(1 + \frac{1}{r} \left(A_1 \cos \psi + A_2 \sin \psi\right) + \mathcal{O}(\varepsilon)\right).$$

Therefore,

$$\frac{1}{|z|^2} = (2r)^2 \left( 1 + \frac{1}{r} \left( A_1 \cos \psi + A_2 \sin \psi \right) + \mathcal{O}(\varepsilon) \right) (1 + \mathcal{O}(\varepsilon)).$$

Since  $\ln(1+t) = t + \mathcal{O}(t^2)$ , we deduce

$$-\ln|z| = \ln(2r) + \frac{1}{2r}(A_1\cos\psi + A_2\sin\psi) + \mathcal{O}(\varepsilon).$$

Finally, it is not very difficult to prove that

$$A_1 \cos \psi + A_2 \sin \psi = \frac{1}{4|z|^2} \left( \eta_1 \cos \psi - \eta_2 \sin \psi \right)$$
$$= r^2 \left( -\eta_1 \cos \theta + \eta_2 \sin \theta \right) \left( 1 + \mathcal{O}(\sqrt{\varepsilon}) \right),$$

from where Lemma 58 follows.

If we consider small translations of  $M_{\sigma,\alpha,\beta}$  dilated by a factor  $1 + \gamma$ , for some small  $\gamma$ , we obtain

$$X_3 = -(1+\gamma)\ln\frac{2r}{1+\gamma} + r\left(-\eta_1\cos\theta + \eta_2\sin\theta\right) + \frac{1}{r}(\kappa_1\cos\theta + \kappa_2\sin\theta) + \kappa_3 + \mathcal{O}(\varepsilon),$$

for small  $\kappa_1, \kappa_2, \kappa_3 \in \mathbb{R}^+$ , where  $\eta_1 = b_1 + b_2$  and  $\eta_2 = b_1 - b_2$ .
#### 4.4.2 Parametrization of the KMR example on the cylinder

In this subsection we want to parametrize the KMR example  $M_{\sigma,\alpha,\beta}$  on a cylinder. To this aim, we introduce the sphero-conal coordinates (x, y) on the unit sphere  $\mathbb{S}^2$  (see [17]): For any  $(x, y) \in \mathbb{S}^1 \times [0, \pi) \equiv [0, 2\pi) \times [0, \pi)$ , define

$$F(x,y) = (\cos x \, \sin y, \sin x \, m(y), l(x) \, \cos y) \in \mathbb{S}^2,$$

where

 $m(y) = \sqrt{1 - \cos^2 \sigma \cos^2 y}$  and  $l(x) = \sqrt{1 - \sin^2 \sigma \sin^2 x}$ .

Recall that the conformal compactification  $\Sigma_{\sigma}$  of  $M_{\sigma,\alpha,\beta}$  only depends on  $\sigma$ . The parameter  $\sigma \in (0, \frac{\pi}{2})$  will remains fixed along this subsection, and we will omit the dependence of the functions we are introducing on  $\sigma$ .

Note that, when  $\sigma = 0$ , the conformal compactification of the limit surface  $M_{0,\alpha,\beta}$  is not a torus but a sphere, and the above sphero-conal coordinates reduce to the spherical ones.

Il faut bien ecrire ça: The coordinate surfaces  $\{x = constant\}$  and  $\{y = constant\}$  are two elliptic cones with vertex at the origin. The cross section of each of these cones with a vertical plane which is orthogonal to the axis of the cone is an ellipse.

Recall that  $\Sigma_{\sigma}$  can be seen as a branched 2-covering of  $\overline{\mathbb{C}}$ , by gluing  $\overline{\mathbb{C}}_1, \overline{\mathbb{C}}_2$  along two common cuts  $\gamma_1$  and  $\gamma_2$  along the imaginary axis joining the branch points D, D' and D'', D''' respectively (see (4.9)). If we compose F(x, y) with the stereographical projection and enlarge the domain of definition of the function, we obtain the differentiable map  $\mathbf{z}(x, y) : \mathbb{S}^1 \times \mathbb{S}^1 \equiv [0, 2\pi) \times [0, 2\pi) \to \overline{\mathbb{C}}$  given by

$$\mathbf{z}(x,y) = \frac{\cos x \, \sin y + i \, \sin x \, m(y)}{1 - l(x) \, \cos y},$$

which is a branch 2-covering of  $\overline{\mathbb{C}}$  with branch values in the four points whose spheroconal coordinates are  $(x, y) \in \{\pm \frac{\pi}{2}\} \times \{0, \pi\}$ , which also correspond to D, D', D'', D'''. Moreover,  $\mathbf{z}(x, y)$  maps  $\mathbb{S}^1 \times (0, \pi)$  on  $\overline{\mathbb{C}} - (\gamma_1 \cup \gamma_2)$ . Hence we can parametrize the KMR example by  $\mathbf{z}$ , by means of its Weierstrass data.

We denote by  $M_{\sigma,\alpha,\beta}$  the lifting of  $M_{\sigma,\alpha,\beta}$  to  $\mathbb{R} \times \mathbb{S}^1 \times \mathbb{R}$  by forgetting its non-horizontal period (i.e. its period in homology,  $T_2$ ). We can then parametrize  $\widetilde{M}_{\sigma,\alpha,\beta}$  on  $\mathbb{S}^1 \times \mathbb{R}$  by extending  $\mathbf{z}$  to  $[0, 2\pi) \times \mathbb{R}$ . But such a parametrization is not conformal, since the spheroconal coordinates  $(x, y) \mapsto F(x, y)$  of the sphere are not conformal. As the stereographic projection is a conformal map, it suffices to find new conformal coordinates (u, v) of the sphere defined on the cylinder. In particular, we look for a change of variables  $(x, y) \mapsto$ (u, v) for which  $|\widetilde{F}_u| = |\widetilde{F}_v|$  and  $\langle \widetilde{F}_u, \widetilde{F}_v \rangle = 0$ , where  $\widetilde{F}(u, v) = F(x(u, v), y(u, v))$ . We observe that

$$|F_x| = \frac{\sqrt{T(x,y)}}{m(x)}$$
 and  $|F_y| = \frac{\sqrt{T(x,y)}}{l(y)}$ 

with  $T(x,y) = \sin^2 \sigma \cos^2 x + \cos^2 \sigma \sin^2 y$ . Then it is natural to consider the change of variables  $(x,y) \in [0,2\pi) \times [0,\pi) \mapsto (u,v) \in [0,U_{\sigma}] \times ??$  defined by

$$u(x) = \int_0^x \frac{1}{l(t)} dt$$
 and  $v(y) = \int_0^y \frac{1}{m(s)} ds$ , (4.12)

where

$$U_{\sigma} = u(2\pi) = \int_0^{2\pi} \frac{dt}{\sqrt{1 - \sin^2 \sigma \sin^2 t}}.$$
 (4.13)

Note that  $U_{\sigma}$  is a function on  $\sigma$  that goes to  $2\pi$  as  $\sigma$  approaches to zero, and that the above change of variables is well defined because  $\sigma \in (0, \frac{\pi}{2})$ . From all this, we can deduce that  $\widetilde{M}_{\sigma,\alpha,\beta}$  is conformally parametrized on  $(u, v) \in I_{\sigma} \times \mathbb{R}$ , with  $I_{\sigma} = [0, U_{\sigma}]$ .

#### **RÉDIGER:**

**Remark 60.** In lemma 58 assuming  $|z| = \mathcal{O}(\sqrt{\epsilon})$  we have found the equation as a graph about an appropriate neighbourhood of the part of the surface  $\widetilde{M}_{\sigma,\alpha,\beta}$  along which we will glue it with the Costa-Hoffman-Meeks surface. It is possible to prove that if  $|z| = \mathcal{O}(\sqrt{\epsilon})$ and  $\sigma = \mathcal{O}(\sqrt{\epsilon})$  then  $y = y_{\epsilon} = \pi - \mathcal{O}(\sqrt{\epsilon})$ . Moreover using (4.12) we can obtain the corresponding value of v. It is given by  $v_{\epsilon} = -\frac{1}{2} \ln \epsilon + \mathcal{O}(1)$ .

# 4.5 The Jacobi operator about $M_{\sigma,\alpha,\beta}$

The Jacobi operator for  $M_{\sigma,\alpha,\beta}$  is given by  $\mathcal{J} = \Delta_{ds^2} + |A|^2$ , where  $|A|^2$  is the squared norm of the second fundamental form on  $M_{\sigma,\alpha,\beta}$  and  $\Delta_{ds^2}$  is the Laplace-Beltrami operator with respect to the metric  $ds^2$  induced on the surface by the immersion X defined in section 4.4. That is

$$ds^{2} = \frac{1}{4} \left( |g| + |g|^{-1} \right)^{2} |dh|^{2}$$

In subsection 4.4.2 we have presented the parametrization of  $M_{\sigma,\alpha,\beta}$  and of its lifting on a cylinder. We recall we adopted the coordinates (x, y). In this new frame the Jacobi operator,  $\mathcal{J}$ , is obtained considering as metric on the rectangular torus  $\Sigma_{\sigma}$ , the pull-back of standard metric  $ds_0^2$  of the sphere  $\mathbb{S}^2$  by the Gauss map. The following relation holds

$$\mathcal{J} = -K \left( \Delta_{ds_0^2} + 2 \right)$$

(being  $K = -\frac{1}{2}|A|^2$  the Gauss curvature) since  $\Delta_{ds^2} = -K \Delta_{ds^2_0}$  and  $dN^*(ds^2_0) = -K ds^2$ . From [17] and taking into account the parametrization of  $\widetilde{M}_{\sigma,\alpha,\beta}$  given in subsection 4.4.2, we can deduce that, in the (x, y)-variables on  $\mathbb{S}^2 \setminus \{\gamma_1 \cup \gamma_2\}$ 

$$\Delta_{ds_0^2} := \frac{l(x) m(y)}{T(x, y)} \left[ \partial_x \left( \frac{l(x)}{m(y)} \partial_x \right) + \partial_y \left( \frac{m(y)}{l(x)} \partial_y \right) \right].$$

After the conformal change of coordinates  $(x, y) \to (u, v)$  defined by (4.12), we can write  $\mathcal{J} = \frac{-K}{T(x(u), y(v))} \mathcal{L}_{\sigma}$ , where

$$\mathcal{L}_{\sigma} := \partial_{uu}^2 + \partial_{vv}^2 + 2\sin^2\sigma\,\cos^2(x(u)) + 2\cos^2\sigma\,\sin^2(y(v)) \tag{4.14}$$

is known as Lamé operator.

**Remark 61.** In proposition 66, we will take limits as  $\sigma \to 0$ . For such a limit, the Riemann surface  $\Sigma_{\sigma}$  degenerates into a Riemann surface with nodes consisting of two spheres jointed by two common points, and the corresponding Jacobi operator equals  $\mathcal{L}_0 =$  $\partial_{xx}^2 + \sin y \,\partial_y (\sin y \,\partial_y) + 2 \sin^2 y$  in the (x, y)-variables. Note that in this case the change of variables  $(x, y) \mapsto (u, v)$  is not defined.

#### 4.5.1 The mapping properties of the Jacobi operator

Our aim along this subsection is to study the mapping properties of the operator  $\mathcal{J}$ . It is clear that it is sufficient to study the simpler operator  $\mathcal{L}_{\sigma}$  defined by (4.14). So we want to study the possibility to solve in a unique way the problem

$$\begin{cases} \mathcal{L}_{\sigma}w = f, & \text{in } I_{\sigma} \times [v_0, +\infty] \\ w|_{v=v_0} = \varphi \end{cases}$$

with  $v_0 \in \mathbb{R}$ , considering convenient normed functional spaces for w, f and  $\varphi$ , so that the norm of w is bounded by the one of f.

Since  $\mathcal{L}_{\sigma}$  has separated variables, let us firstly consider the operator

$$L_{\sigma} = \partial_{uu}^2 + 2\sin^2 \sigma \cos^2(x(u)).$$

The quantity  $U_{\sigma}$  defined in (4.13) is the period of the function  $\cos(x(u))$ . It is possible to prove that  $U_{\sigma} \to 2\pi$  as  $\sigma \to 0$ . We let  $L_{\sigma}$  act on the  $U_{\sigma}$ -periodic and even functions. Moreover it is uniformly elliptic and self-adjoint. In particular,  $L_{\sigma}$  has discrete spectrum  $(\lambda_{\sigma,i})_{i\geq 0}$ , that we assume arranged so that  $\lambda_{\sigma,i} < \lambda_{\sigma,i+1}$  for every *i*. Each eigenvalue  $\lambda_{\sigma,i}$  is simple because we only consider even functions. We denote by  $e_{\sigma,i}$  the even eigenfunction associated to  $\lambda_{\sigma,i}$ , normalized so that

$$\int_0^{U_\sigma} (e_{\sigma,i}(u))^2 \, du = 1.$$

**Lemma 62.** For every  $i \ge 0$ , the eigenvalue  $\lambda_{\sigma,i}$  of the operator  $L_{\sigma}$  and its associated eigenfunctions  $e_{\sigma,i}$  satisfy

$$-2\sin^2\sigma \leqslant \lambda_{\sigma,i} - i^2 \leqslant 0, \qquad |e_{\sigma,i} - e_{0,i}|_{\mathcal{C}^2} \leqslant c_i \sin^2\sigma, \tag{4.15}$$

where  $e_{0,i}(u) := \cos(ix(u))$  for every  $u \in I_{\sigma}$ , and the constant  $c_i > 0$  depends only on *i* (it does not depend on  $\sigma$ ).

**Proof.** The bound for  $\lambda_{\sigma,i} - i^2$  comes from the variational characterization of the eigenvalues,

$$\lambda_{\sigma,i} = \sup_{\text{codim}\,E=i} \inf_{e \in E, \, ||e||_{L^2}=1} \int_0^{U_\sigma} \left( (\partial_u e)^2 - 2\sin^2 \sigma \cos^2(x(u)) \, e^2 \right) \, du$$

where E is a subset of the space of  $U_{\sigma}$ -periodic even functions in  $L^2(I_{\sigma})$ , since it always holds  $0 \leq 2 \sin^2 \sigma \cos^2(x(u)) \leq 2 \sin^2 \sigma$ .

The bound for the eigenfunctions follows from standard perturbation theory [21].

The Hilbert basis  $\{e_{\sigma,i}\}_{i\in\mathbb{N}}$  of the space of  $U_{\sigma}$ -periodic and even functions in  $L^2(I_{\sigma})$  induces the following Fourier decomposition of  $L^2$  functions g = g(u, v) which are  $U_{\sigma}$ -periodic and even in the *u*-variable,

$$g(u,v) = \sum_{i \ge 0} g_i(v) e_{\sigma,i}(u)$$

From this, we deduce that the operator  $\mathcal{L}_{\sigma}$ , can be decomposed as  $\mathcal{L}_{\sigma} = \sum_{i \ge 0} L_{\sigma,i}$ , being

$$L_{\sigma,i} = \partial_{vv}^2 + 2\cos^2\sigma\sin^2(y(v)) - \lambda_{\sigma,i}, \quad \text{for every } i \ge 0.$$

Since  $0 \leq 2\cos^2 \sigma \sin^2(y(v)) \leq 2\cos^2 \sigma = 2 - 2\sin^2 \sigma$ , the lemma 62 give us

$$P_{\sigma,i} := 2\cos^2 \sigma \,\sin^2(y(v)) - \lambda_{\sigma,i} \leqslant 2 - i^2. \tag{4.16}$$

 $\square$ 

This fact allows us to prove the following lemma, which assures that  $\mathcal{L}_{\sigma}$  is injective when restricted to the set of functions that are  $L^2$ -orthogonal to  $e_{\sigma,0}$  and  $e_{\sigma,1}$  in the *u*-variable.

**Lemma 63.** Given  $v_0 < v_1$ , let w be a solution of  $\mathcal{L}_{\sigma}w = 0$  on  $I_{\sigma} \times [v_0, v_1]$  such that

(i) 
$$w(\cdot, v_0) = w(\cdot, v_1) = 0.$$
  
(ii)  $\int_0^{T_{\sigma}} w(u, v) e_{\sigma,i}(u) du = 0$ , for every  $v \in [v_0, v_1]$  and  $i = 0, 1.$   
Then  $w = 0.$ 

*Proof.* By (*ii*),  $w = \sum_{i \ge 2} w_i(v) e_{\sigma,i}(u)$ . Since the potential  $P_{\sigma,i}$  of the operator  $L_{\sigma,i}$  is negative for every  $i \ge 2$  (see (4.16)) and the operator  $L_{\sigma,i}$  is elliptic, the maximum principle holds. We can then conclude the lemma 63 from (*i*).

Now we can state the following

**Lemma 64.** For all  $i \ge 2$  and  $\forall \sigma$  there exists a unique positive solution of

$$L_{\sigma,i}w_{\sigma,i} = 0$$
 with  $w_{\sigma,i}(0) = 1$ 

defined on  $I_{\sigma} = [0, U_{\sigma}]$  such that

$$\frac{1}{c_{\sigma,i}} e^{\pm \gamma_{\sigma,i} v} \leqslant w_{\sigma,i}(v) \leqslant c_{\sigma,i} e^{\pm \gamma_{\sigma,i} v}$$
(4.17)

for some constants  $\gamma_{\sigma,i} > 0$  and  $c_{\sigma,i} > 1$ .

*Proof.* For  $i \ge 2$  and for  $0 < \sigma < \pi/2$ , the potential  $P_{\sigma,i}$  of  $L_{\sigma,i}$  is negative, hence this operator satisfies the maximum principle. We can choose two constants  $\gamma < 0$  close enough to 0 and  $\gamma' < 0$  with  $|\gamma'|$  large enough so that

$$\gamma^2 + P_{\sigma,i} < 0$$
 and  $\gamma'^2 + P_{\sigma,i} > 0.$ 

This choice allows us to use the Perron method to prove the existence of the solution. In fact the functions  $v \to e^{\gamma v}$  and  $v \to e^{\gamma' v}$  respectively satisfy

$$L_{\sigma,i}e^{\gamma v} = (\gamma^2 + P_{\sigma,i})e^{\gamma v} < 0$$

and

$$L_{\sigma,i}e^{\gamma' v} = (\gamma'^2 + P_{\sigma,i})e^{\gamma' v} > 0$$

that is they are respectively a subsolution and a supersolution and so they can be used like barrier functions.

The existence of the constants  $\gamma_{\sigma,i}$  follows from the fact that the potential of  $L_{\sigma,i}$  consists of the function  $\sin^2(y(v))$  which is periodic in the variable v. We denote the period by  $R_{\sigma}$ . It is possible to show that  $R_{\sigma} \to +\infty$  if  $\sigma \to 0$ . Indeed, we can define the linear operator  $M_{\sigma,i}$  by

 $M_{\sigma,i}(s_1, s_2) = (s(R_{\sigma}), \partial_v s(R_{\sigma}))$ 

where s is the unique solution of

$$L_{\sigma,i} s = 0$$

which is defined on  $[0, R_{\sigma}]$  and which satisfies  $s(0) = s_1$  and  $\partial_v s(0) = s_2$ .

We claim that  $M_{\sigma,i}$  has two positive eigenvalues which satisfy  $m_{\sigma,i} < 1 < n_{\sigma,i}$  and  $m_{\sigma,i} n_{\sigma,i} = 1$ . The solution s which has been described above is exponentially decaying and  $(s(0), \partial_v s(0))$  corresponds to an eigenvector of  $M_{\sigma,i}$  with eigenvalue  $m_{\sigma,i} < 1$ . Therefore, we have

$$M_{\sigma,i}(s(0),\partial_v s(0)) = m_{\sigma,i}\left(s(0),\partial_v s(0)\right) = \left(s(R_{\sigma}),\partial_v s(R_{\sigma})\right)$$

This implies that

$$(s(2R_{\sigma}), \partial_v s(2R_{\sigma})) = M_{\sigma,i}(s(R_{\sigma}), \partial_v s(R_{\sigma})) = m_{\sigma,i}^2(s(0), \partial_v s(0))$$

and so for all  $k \in \mathbb{N}$  we have

$$(s(k R_{\sigma}), \partial_v s(k R_{\sigma})) = m_{\sigma,i}^k(s(0), \partial_v s(0)).$$

Hence we have the relation

$$e^{-\gamma_{\sigma,i}R_{\sigma}} = m_{\sigma,i}$$

which defines  $\gamma_{\sigma,i}$ .

Now we prove the claim. Assume that  $g_1$  and  $g_2$  are two solutions of  $L_{\sigma,i}s = 0$  with  $g_1(0) = \partial_v g_1 = 0$  and  $g_2(0) = \partial_v g_1 = 0$ . The Wronskian  $W(g_1, g_2)$  associated to  $g_1$  and  $g_2$  does not depend on the variable v because the Wronskian of a differential equation of the form

$$\partial_{vv}f + a(v)\partial_v f + b(v)f = 0$$

must satisfy the differential equation  $\partial_v W + a(v)W = 0$ . In this case a(v) = 0 and so we have  $\partial_v W = 0$ . Then if we compute W at v = 0 and  $v = R_{\sigma}$ , we can write

$$g_1(0)g_2'(0) - g_1'(0)g_2(0) = g_1(R_{\sigma})g_2'(R_{\sigma}) - g_1'(R_{\sigma})g_2(R_{\sigma}).$$

Now we observe that from the previous assumption it follows that

$$g_1(0)g_2'(0) = det(N_{\sigma,i})g_1(0)g_2'(0),$$

where  $N_{\sigma,i}$  is the matrix associated to the operator  $M_{\sigma,i}$ . But the determinant is equal to the product of its eigenvalues so we have proven that  $m_{\sigma,i}n_{\sigma,i} = 1$ . Finally the entries of  $N_{\sigma,i}$  are real and so it is for its trace. This implies that the eigenvalues are real. We must prove that the eigenvalues cannot be equal to 1. It is sufficient to observe that it is not possible, otherwise it should exist a nontrivial bounded solution of the homogeneous problem  $L_{\sigma,i} s = 0$ . Namely at a point where this solution has a positive maximum (that is its second derivative is negative) we have  $0 = L_{\sigma,i} s \leq P_{\sigma,i} s < 0$  and in correspondence of a negative minimum (that is its second derivative is positive)  $0 = L_{\sigma,i} s \geq P_{\sigma,i} s > 0$ ).  $\Box$ 

The parameters  $\gamma_{\sigma,i}$  are called the indicial roots of the operator  $L_{\sigma,i}$ .

When i = 0 and i = 1, the argument used above does not hold since the potential of  $L_{\sigma,i}$  is not negative. In this case, the explicit solutions of the equation  $L_{\sigma,i} s = 0$  are obtained thanks to the existence of the Jacobi fields. The idea is that our surface has many properties of symmetry and this induces 4 independent Jacobi fields that we are going to describe. We recall that in the following we will consider the two subfamilies of  $M_{\sigma,\alpha,\beta}$  for which  $\alpha = 0$  and  $\beta = 0$ . We described them in section 4.4. These surfaces are invariant by the symmetry about the plane  $\{x_1 = 0\}$  ( $\alpha = 0$ ) and the plane  $\{x_2 = 0\}$  ( $\beta = 0$ ). So the set of the Jacobi fields to consider are different in the two cases.

- Two Jacobi fields can be obtained by considering the one parameter families of minimal surfaces which is induced by the translations in the  $x_3$ -direction and by
  - the translations in the  $x_2$ -direction in the case where  $\alpha = 0$ ,
  - the translations in the  $x_1$ -direction in the case where  $\beta = 0$ .

These Jacobi fields are clearly periodic and hence bounded.

- A third Jacobi field can be obtained by considering the one parameter family of minimal surfaces which is induced by dilatation from the origin. The so-obtained Jacobi field is not bounded and in fact it grows linearly.
- The last Jacobi field can be obtained by considering the one parameter family of minimal surfaces which is induced by changing the parameter  $\sigma$ . Again, this Jacobi field is not periodic and grows linearly.

The Jacobi operator  $\mathcal{L}_{\sigma}$  becomes a Fredholm operator when restricted to the following functional space.

**Definition 65.** Given  $\sigma \in (0, \pi/2)$ ,  $\ell \in \mathbb{N}$ ,  $\alpha \in (0, 1)$ ,  $\mu \in \mathbb{R}$  and an interval I, we define  $\mathcal{C}^{\ell,\alpha}_{\mu}(I_{\sigma} \times I)$  to be the space of functions  $w \equiv w(u, v)$  in  $\mathcal{C}^{\ell,\alpha}_{loc}(I_{\sigma} \times I)$  which are even in the variable u and for which the following norm is finite:

$$\|w\|_{\mathcal{C}^{\ell,\alpha}_{\mu}} := \sup_{v \in I} e^{-\mu v} \|w\|_{\mathcal{C}^{\ell,\alpha}(I_{\sigma} \times [v,v+1])}.$$

**Proposition 66.** Given  $\mu \in (-2, -1)$ , there exists a  $\sigma_0 \in (0, \pi/2)$  such that, for every  $\sigma \in (0, \sigma_0)$  and  $v_0 \in \mathbb{R}$ , there exists an operator

$$\begin{array}{ccc} G_{\sigma,v_0}: & \mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_0, +\infty)) & \longrightarrow & \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_0, +\infty)) \\ & f & \longmapsto & w := G_{\sigma,v_0}(f) \end{array}$$

satisfying the following statements:

(i)  $\mathcal{L}_{\sigma} w = f \text{ on } I_{\sigma} \times [v_0, +\infty);$ 

(ii) 
$$w \in Span\{e_{\sigma,0}, e_{\sigma,1}\}$$
 on  $I_{\sigma} \times \{v_0\}$ ;

(iii)  $||w||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c ||f||_{\mathcal{C}^{0,\alpha}_{\mu}}$ , for some constant c > 0 which does not depend on  $\sigma, v_0$ .

*Proof.* Every  $f \in \mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_0, +\infty))$  can be decomposed as

$$f = f_0 \, e_{\sigma,0} + f_1 \, e_{\sigma,1} + \bar{f},$$

where  $f(\cdot, v)$  is  $L^2$ -orthogonal to  $e_{\sigma,0}$  and  $e_{\sigma,1}$  for each v.

**Step 1.** Firstly, let's prove the proposition 66 for the functions  $f \in \mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_0, +\infty))$  that are  $L^2$ -orthogonal to  $\{e_{\sigma,0}, e_{\sigma,1}\}$ . As a consequence of the lemma 63,  $\mathcal{L}_{\sigma}$  is injective when it acts on this set of functions. Hence, the Fredholm alternative assures that there exists, for each  $v_1 > v_0 + 1$ , an unique  $w \in \mathcal{C}^{2,\alpha}_{\mu}$ , with  $w(\cdot, v) L^2$ -orthogonal to  $e_{\sigma,0}, e_{\sigma,1}$  satisfying:

$$\begin{cases} \mathcal{L}_{\sigma} w = f \quad \text{on } I_{\sigma} \times [v_0, v_1], \\ w(\cdot, v_0) = w(\cdot, v_1) = 0. \end{cases}$$

$$(4.18)$$

Assertion 67. There exists a constant c and  $\sigma_0 \in (0, \pi/2)$  such that for every  $\sigma \in (0, \sigma_0)$ ,  $v_0 \in \mathbb{R}, v_1 > v_0 + 1, f \in \mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_0, v_1])$  and  $w \in \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_0, v_1])$  satisfying the equation (4.18) and

$$\|w\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c \, \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}.\tag{4.19}$$

Suppose by contradiction that the assertion 67 is false. Then, for every  $n \in \mathbb{N}$ , there exists  $\sigma_n \in (0, 1/n)$ ,  $v_{1,n} > v_{0,n} + 1$  and  $f_n, w_n$  satisfying (4.18) (for  $\sigma_n, v_{0,n}, v_{1,n}$  instead of  $\sigma, v_0, v_1$ ) such that

$$||f_n||_{\mathcal{C}^{0,\alpha}_{\mu}} = 1$$
 and  $||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}} \to +\infty$ , when  $n \to \infty$ .

Since  $I_{\sigma_n} \times [v_{0,n}, v_{1,n}]$  is a compact set,  $A_n := \sup_{I_{\sigma_n} \times [v_{0,n}, v_{1,n}]} e^{-\mu v} |w_n|$  is achieved at a point  $(u_n, v_n) \in I_{\sigma_n} \times [v_{0,n}, v_{1,n}]$ . We define

$$\widetilde{w}_n(u,v) := \frac{e^{-\mu v_n}}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}}} w_n(u,v+v_n),$$

for all  $(u, v) \in I_{\sigma_n} \times I_n$ , with  $I_n = [v_{0,n} - v_n, v_{1,n} - v_n]$ . Clearly,  $A_n \leq ||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}}$ , and

$$|\widetilde{w}_n(u,v)| \leqslant e^{\mu v} \, \frac{e^{-\mu(v+v_n)}|w_n(u,v+v_n)|}{A_n} \leqslant e^{\mu v}$$

On the other hand,  $e^{-\mu v} |\nabla \widetilde{w}_n| \leq \|\widetilde{w}_n\|_{\mathcal{C}^{2,\alpha}_{\mu}} = \frac{\|w_n\|_{\mathcal{C}^{2,\alpha}_{\mu}}}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}}}$ . Thanks to Schauder estimate, we obtain  $||w_n||_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c' \left( ||f_n||_{\mathcal{C}^{0,\alpha}_{\mu}} + ||e^{-\mu v}w_n||_{\mathcal{C}^0} \right) = c' (1 + A_n)$ . Hence<sup>3</sup>,

$$|\nabla \widetilde{w}_n| \leqslant c' e^{\mu v} \frac{1 + ||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}}}{||w_n||_{\mathcal{C}^{0,\alpha}_{\mu}}} \leqslant c e^{\mu v}.$$

<sup>&</sup>lt;sup>3</sup>From now on, c will denote any arbitrary positive constant.

The intervals  $I_n$  converge to a nonempty (and possibly unbounded) interval  $I_{\infty}$ . Since the sequences  $(\tilde{w}_n)_n$  and  $(\nabla \tilde{w}_n)_n$  are uniformly bounded, Ascoli-Arzelà theorem assures that a subsequence of  $(\tilde{w}_n)_n$  converges for  $n \to \infty$  (and  $\sigma_n \to 0$ ) on compact sets of  $I_{\sigma_n} \times I_{\infty}$  to a function  $w_{\infty}$ , which is  $L^2$ -orthogonal to  $\{e_{0,0}, e_{0,1}\}$  for each  $v \in I_{\infty}$ , and vanishes on  $I_0 \times \partial I_{\infty}$ , when  $\partial I_{\infty} \neq \emptyset$ . Note that

$$\sup_{v \in I_n} e^{-\mu v} \| \widetilde{w}_n \|_{\mathcal{C}^{0,\alpha}(I_{\sigma_n} \times [v,v+1])} = \frac{A_n}{\| w_n \|_{\mathcal{C}^{0,\alpha}_{\mu}}},$$
(4.20)

does not converge to zero. In fact  $A_n \to \infty$ ,  $\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leq \|w_n\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c'(1+A_n)$  then

$$\frac{A_n}{\|w_n\|_{\mathcal{C}^{0,\alpha}_{\mu}}} \geqslant \frac{A_n}{c'(1+A_n)} \to \frac{1}{c'} > 0.$$

In particular,

$$\sup_{I_{\sigma_n} \times I_n} e^{-\mu v} \left| \tilde{w}_n \right| = 1.$$
(4.21)

Since if  $n \to \infty$  we have  $\sigma_n \to 0$ , from (4.12) we can conclude

$$u \to x$$
 and  $v \to \frac{1}{2} \ln |\tan \frac{y}{2}|$ .

From the last expression we get

$$y(v) = 2 \arctan(e^{2v})$$
 and  $e^{2v} = |\tan \frac{y}{2}|$ .

Using well known trigonometric formulae also we find

$$\cos y(v) = \frac{1 - e^{4v}}{1 + e^{4v}}$$
 and  $\sin y(v) = \frac{2e^{2v}}{1 + e^{4v}}.$  (4.22)

It is possible to find the expression of the function  $w_{\infty}$  working with the x, y coordinates and after that to come back to the u, v coordinates. In fact we can observe that, up to subsequence, the function  $w_{\infty}$  satisfies  $\mathcal{L}_0 w_{\infty} = 0$  with

$$\mathcal{L}_0 = \partial_{xx}^2 + \sin y \, \partial_y \left( \sin y \, \partial_y \right) + 2 \sin^2 y.$$

Now we consider the eigenfunctions decomposition of  $w_{\infty}$ ,

$$w_{\infty}(x,y) = \sum_{j \ge 2} a_j(y) \, \cos(j \, x)$$

Each coefficient  $a_j$  must satisfy the associate Legendre differential equation

$$\left(\sin^2 y \,\partial_{yy}^2 + \cos y \,\sin y \,\partial_y - j^2 + 2\sin^2 y\right) \,a_j = 0$$

Since  $j \ge 2$  the solutions are the associated Legendre functions of second kind (the functions of second kind being zero)  $a_j(y) = Q_1^j(\cos y)$ , where

$$Q_1^j(t) = (-1)^j \sqrt{(1-t^2)^j} \, \frac{d^j Q_1^0(t)}{dt^j}, \quad \text{with} \ \ Q_1^0(t) = \frac{t}{2} \ln\left(\frac{1+t}{1-t}\right) - 1.$$

Now we come back to the variables (u, v). We observe that the function

$$w_{\infty}(u,v) = \sum_{j \ge 2} Q_1^j(\cos y(v)) \cos(j u)$$

does not satisfy the inequality (4.21) with  $n \to +\infty$  and  $\mu \in (-2, -1)$ , a contradiction.

This proves the assertion 67, that is, for every  $v_1 > v_0 + 1$ , there exists a function  $\bar{w}$  satisfying (4.19). Let's take the limit as  $v_1 \to \infty$ . Clearly,

$$e^{-\mu v} |\bar{w}| \leq \|\bar{w}\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leq c \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}.$$

And the Schauder estimates assures

$$e^{-\mu v} |\nabla \bar{w}| \leq \|\bar{w}\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \left( \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}} + \|\bar{w}\|_{\mathcal{C}^{0}_{\mu}} \right) \leq c \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}.$$

Hence Ascoli-Arzelà theorem says to us that a subsequence of  $\{w_{v_1}\}_{v_1>v_0+1}$  converges to a function  $w \in C^{2,\alpha}_{\mu}$  defined on  $I_{\sigma} \times [v_0, \infty)$ , which clearly satisfies the statement *(iii)* of proposition 66.

**Step 2** Let's now consider  $f \in \mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_0, +\infty))$  in  $Span\{e_{\sigma,0}, e_{\sigma,1}\}$ , i.e.

$$f(u, v) = f_0(v) e_{\sigma,0}(u) + f_1(v) e_{\sigma,1}(u).$$

We extend the functions  $f_0(v)$ ,  $f_1(v)$  for  $v \leq v_0$  to be equal, respectively, to  $f_0(v_0)$ ,  $f_1(v_0)$ . Given  $v_1 > v_0 + 1$ , consider

$$\begin{cases}
L_{\sigma,j}w_j = f_j, \quad v \in (-\infty, v_1] \\
w_j(v_1) = \partial_v w_j(v_1) = 0
\end{cases}$$
(4.23)

Peano theorem assures the existence and the uniqueness of the solution  $w_j$ . Our aim consists in proving the following

Assertion 68.  $||w_j||_{\mathcal{C}^{0,\alpha}_{\mu}} \leq c ||f_j||_{\mathcal{C}^{0,\alpha}_{\mu}}$  for some constant c which does not depend on  $v_1$ .

Suppose by contradiction that, for every  $n \in \mathbb{N}$ , there exists  $\sigma_n \in (0, 1/n)$ ,  $v_{1,n} > v_{0,n} + 1$ and  $f_{j,n}, w_{j,n}$  satisfying (4.23) such that

$$\|f_{j,n}\|_{\mathcal{C}^{0,\alpha}_{\mu}} = 1$$
 and  $\|w_{j,n}\|_{\mathcal{C}^{0,\alpha}_{\mu}} \to +\infty$ , when  $n \to \infty$ .

The solution  $w_{j,n}$  of the previous equation is a linear combination of the two solutions of the homogeneous problem  $L_{\sigma_{n,j}} w = 0$ . They are the Jacobi fields associated to the isometries and it is known that they are at most linearly growing functions. Hence the supremum

$$A_n := \sup_{(-\infty,v_{1,n}]} e^{-\mu v} |w_{j,n}|$$

is achieved in a point which we call  $v_n \in (-\infty, v_{1,n}]$ . We define on  $I_n := (-\infty, v_{1,n} - v_n]$ the function  $\tilde{w}_{j,n}$  by

$$\tilde{w}_{j,n}(v) := \frac{1}{\|w_{j,n}\|_{\mathcal{C}^{0,\alpha}_{\mu}}} e^{-\mu v_n} w_{j,n}(v_n + v).$$

As above, one shows that the sequence  $(v_{1,n} - v_n)_n$  remains bounded away from 0, that is  $v_{1,n} > v_n$  for each n. Without loss of generality, we can assume that the sequence  $(v_{1,n} - v_n)_n$  converges to  $\bar{v}_1 \in (0, +\infty]$ . We set  $I_{\infty} = (-\infty, \bar{v}_1]$ .

As in Step 1, we can also assume that the sequence of functions  $(\tilde{w}_{j,n})_n$  converges on compact subsets of  $I_{\infty}$  to a nontrivial function  $\tilde{w}_j$ . We observe that  $\tilde{w}_j(\bar{v}_1) = 0$  if  $\bar{v}_1 < +\infty$  and that

$$\sup_{v \in I_{\infty}} e^{-\mu v} \left| \tilde{w}_j \right| = 1. \tag{4.24}$$

Secondly  $\tilde{w}_i$ , in the coordinates x, y is a solution of

$$\left(\sin^2 y \,\partial_{yy}^2 + \cos y \,\sin y \,\partial_y - j^2 + 2\sin^2 y\right) \,a_j = 0. \tag{4.25}$$

This is again the associated Legendre differential equation. The solutions of the equation (4.25) are the associated Legendre functions of first  $P_1^j(\cos y)$  and second kind  $Q_1^j(\cos y)$  with j = 0, 1.

We have reached a contradiction because we can observe that these solutions, after the change of coordinates to come back to the u, v coordinates, do not satisfy the equation (4.24) with  $\mu \in (-2, -1)$ .

So we have proved that

$$\sup_{(-\infty,v_1]} e^{-\mu v} |w_j| \leqslant c \sup_{(-\infty,v_1]} e^{-\mu v} |f_j|.$$

Now we pass to the limit as  $v_1$  tends to  $+\infty$  in a sequence of solutions which are defined on  $I_{\infty}$ . This proves the existence of a solution of

$$L_{\sigma,j} w_j = f_j$$

which is defined in  $[v_0, +\infty)$ . In addition, we know that

$$\sup_{[v_0,+\infty)} e^{-\mu v} |w_j| \leqslant c \sup_{[v_0,+\infty)} e^{-\mu v} |f_j|.$$

Multiplying for the eigenfunctions, taking the supremum on  $I_{\sigma} \times [v_0, +\infty)$  and using a last time elliptic estimates, we get the wanted estimate. So the proof of the result is complete.

# 4.6 A family of minimal surfaces close to $M_{\sigma,0,\beta}$ and $\widetilde{M}_{\sigma,0,\beta}$

The aim of this section is to find a family of minimal surfaces near to a translated and dilated copy of  $\widetilde{M}_{\sigma,0,\beta}$  and  $\widetilde{M}_{\sigma,0,\beta}$  with given Dirichlet data on the boundary. We start recalling that in subsection 4.4.1 we got that a translated and dilated copy of  $\widetilde{M}_{\sigma,\alpha,\beta}$  can be expressed as the graph over the  $x_3 = 0$  plane of the function

$$(1+\gamma)\ln\frac{2r}{1+\gamma} + r\left[\eta_1\cos\theta + \eta_2\sin\theta\right] + \frac{1}{r}(\kappa_1\cos\theta + \kappa_2\sin\theta) + \kappa_3 + g_t.$$
(4.26)

where  $g_t = \mathcal{O}(\epsilon)$ ,  $\kappa_1, \kappa_2, \kappa_3 \in \mathbb{R}^+$  and small,  $\eta_1 = b_1 + b_2$ ,  $\eta_2 = b_1 - b_2$ ,  $b_1 = \sin \frac{\alpha - \beta}{2}$ ,  $b_2 = \sin \frac{\alpha + \beta}{2}$ , and r belongs to a neighbourhood of  $r_{\epsilon} = \frac{1}{2\sqrt{\epsilon}}$ .

We denote by Z the immersion of the surface  $\widetilde{M}_{\sigma,\alpha,\beta}$ . The following proposition, whose proof is contained in section 4.11, states that the linearized of the mean curvature operator is the Lamé operator introduced in section 4.4.2.

**Proposition 69.** The surface parameterized by  $Z_f := Z + f N$  is minimal if and only if the function f is a solution of

$$\mathcal{L}_{\sigma}f = Q_{\sigma}(f, f_u, f_v, f_{uu}, f_{uv}, f_{vv})$$

where  $\mathcal{L}$  is the Lamé operator and  $Q_{\sigma}$  is a nonlinear operator which satisfies

$$\|Q_{\sigma}(f_{2}) - Q_{\sigma}(f_{1})\|_{\mathcal{C}^{0,\alpha}(I_{\sigma} \times [v,v+1])} \leq c \sup_{i=1,2} \|f_{i}\|_{\mathcal{C}^{2,\alpha}(I_{\sigma} \times [v,v+1])} \|f_{2} - f_{1}\|_{\mathcal{C}^{2,\alpha}(I_{\sigma} \times [v,v+1])}$$

for all  $f_1, f_2$  such that  $||f_i||_{\mathcal{C}^{2,\alpha}(I_{\sigma}\times[v,v+1])} \leq 1$ . Here the constant c > 0 does not depend on  $v \in \mathbb{R}$ , nor on  $\sigma \in (0,1)$ .

As a consequence of the dilation of factor  $1 + \gamma$  of the surface the minimal surface equation becomes

$$\mathcal{L}_{\sigma} w = \frac{1}{1+\gamma} Q_{\sigma} \left( (1+\gamma) \cdot \right), \qquad (4.27)$$

where hereafter we use a simplified notation for the operator  $Q_{\sigma}$ .

We now truncate the surfaces  $M_{\sigma,0,\beta}$  and  $M_{\sigma,\alpha,0}$  at the graph of the curve  $r = \frac{1}{2\sqrt{\epsilon}}$  of the function (4.26) with, respectively,  $\alpha = 0$  and  $\beta = 0$ , and we consider only the upper half

of these surfaces which we call  $M_1$  and  $M_2$ . We are interested in minimal normal graphs over these surfaces which are asymptotic to them. The normal graph of the function wover  $M_1, M_2$  is minimal, if and only if w is a solution of (4.27).

We make the following change of coordinates

$$(r,\theta) = \left(\frac{1}{2}e^{v},\theta\right).$$

It is important to remark that though the surfaces  $M_1, M_2$  can be parameterized by (58), their boundary does not correspond to the curve  $v = v_{\epsilon} = -1/2 \ln \epsilon$ . We therefore modify the above parametrization so that over the annulus  $B_{4r_{\epsilon}} - B_{r_{\epsilon}}$  the image of the function (4.26) corresponds to the horizontal curve  $v = v_{\epsilon}$ . Finally, we interpolate smoothly the two parametrizations over the annulus  $B_{6r_{\epsilon}} - B_{2r_{\epsilon}}$ . We would like that the normal vector field relative to  $M_1, M_2$  is vertical near the boundary of this surface. This can be achieved by modifying the normal vector field into a transverse vector field  $\tilde{N}$  which agrees with the normal vector field N for all  $t \ge v_{\epsilon} + \ln 4$  and with the vector  $e_3$  for all  $v \in [v_{\epsilon}, v_{\epsilon} + \ln 2]$ .

Now, we consider a graph over this surface for some function u, using the modified vector field  $\tilde{N}$ . This graph will be minimal if and only if the function u is a solution of a nonlinear elliptic equation related to (4.27). To get the new equation, we take into account the effects of the change of parameterization and the change in the vector field N into  $\tilde{N}$ . The new minimal surface equation is

$$\mathcal{L}_{\sigma} w = \tilde{L}_{\epsilon} w + \tilde{Q}_{\sigma} \left( \cdot \right). \tag{4.28}$$

Here  $\tilde{Q}_{\sigma}$  enjoys the same properties of  $Q_{\sigma}$ , since it is obtained by a slight perturbation from it. The operator  $\tilde{L}_{\epsilon}$  is a linear second order operator whose coefficients are supported in  $[v_{\epsilon}, v_{\epsilon} + \ln 4] \times S^1$  and are bounded by a constant multiplied for  $\epsilon^{1/2}$ , in  $\mathcal{C}^{\infty}$  topology, where partial derivatives are computed with respect to the vector fields  $\partial_u$  and  $\partial_v$ .

As a fact, if we take into account the effect of the change of the normal vector field, we would obtain, applying the result of Appendix B of [11], a similar formula where the coefficients of the corresponding operator  $\tilde{L}_{\epsilon}$  are bounded by a constant multiplied for  $\epsilon$ since

$$\tilde{N}_{\epsilon} \cdot N_{\epsilon} = 1 + \mathcal{O}(\epsilon)$$

for  $t \in [t_{\epsilon}, t_{\epsilon} + \ln 2]$ . Instead, if we take into account the effect of the change in the parameterization, we would obtain a similar formula where the coefficients of the corresponding operator  $\tilde{L}_{\epsilon}$  are bounded by a constant multiplied for  $\epsilon^{1/2}$ . The estimate of the coefficients of  $\tilde{L}_{\epsilon}$  follows from these considerations.

Now, assume that we are given a function  $\varphi \in C^{2,\alpha}(I_{\sigma})$  which is even with respect to u,  $L^2$ -orthogonal to  $e_{\sigma,0}, e_{\sigma,1}$  and such that  $\|\varphi\|_{C^{2,\alpha}} \leq k\epsilon$ . We define

$$w_{\varphi}(\cdot, \cdot) := \bar{\mathcal{H}}_{v_{\epsilon}, \varphi}(\cdot, \cdot),$$

where  $v_{\epsilon} = -1/2 \ln \epsilon + \mathcal{O}(1)$  and  $\overline{\mathcal{H}}$  is introduced in proposition 81.

In order to solve the equation (4.28), we choose  $\mu \in (-2, -1)$  and look for u of the form  $w = w_{\varphi} + g$  where  $g \in \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty))$  and  $w = \varphi$  on  $I_{\sigma} \times \{v_{\epsilon}\}$ . Using Proposition 66, we can rephrase this problem as a fixed point problem

$$g = S(\varphi, g) \tag{4.29}$$

where the nonlinear mapping S which depends on  $\epsilon$  and  $\varphi$  is defined by

$$S(\varphi,g) := G_{\epsilon,v_{\epsilon}} \left( \tilde{L}_{\epsilon}(w_{\varphi} + g) - \mathcal{L}_{\sigma} w_{\varphi} + \tilde{Q}_{\epsilon} (w_{\varphi} + g) \right).$$

where the operator  $G_{\epsilon,v_{\epsilon}}$  is defined in Proposition 66. To prove the existence of a fixed point for (4.29) we need the following

**Lemma 70.** There exist some constants  $c_k > 0$  and  $\epsilon_k > 0$ , such that

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant c_k \epsilon^{1+\mu/2}$$

$$\tag{4.30}$$

and, for all  $\epsilon \in (0, \epsilon_k)$ 

$$\|S(\varphi,g_2) - S(\varphi,g_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant c_k \epsilon^{\frac{1}{2}} \|g_2 - g_1\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))}$$

for all  $g_1, g_2 \in \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty))$  such that  $\|g_i\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty))} \leq c_k \epsilon^{1+\mu/2}$ .

**Proof:** We know from Proposition 66 that  $\|G_{\epsilon,v_{\epsilon}}(f)\|_{\mathcal{C}^{2,\alpha}_{\mu}} \leq c \|f\|_{\mathcal{C}^{0,\alpha}_{\mu}}$ , then

$$\|S(\varphi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant c \|\tilde{L}_{\epsilon}(w_{\varphi}) - \mathcal{L}_{\sigma}w_{\varphi} + \tilde{Q}_{\sigma}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant$$

$$\leq c \left( \|\tilde{L}_{\epsilon}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} + \|\mathcal{L}_{\sigma}w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} + \|\tilde{Q}_{\sigma}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \right).$$

So we need to find the estimates for the three above norms. We recall that  $|\varphi|_{2,\alpha} \leq k\epsilon$ . For all  $\mu \in (-2, -1)$ , thanks to Proposition 81 we know that

$$|w|_{2,\alpha;[v,v+1]} \leqslant e^{\mu(v-v_{\epsilon})}|\varphi|_{2,\alpha} \tag{4.31}$$

Using the relation  $e^{-\mu v_{\epsilon}} = \epsilon^{\mu/2}$  we know that

$$||w_{\varphi}||_{\mathcal{C}^{2,\alpha}_{\mu}} = \sup_{v \in [v_{\epsilon},\infty]} e^{-\mu v} |w|_{2,\alpha;[v,v+1]} \leqslant \sup_{v \in [v_{\epsilon},\infty]} e^{-\mu v} e^{\mu(v-v_{\epsilon})} |\varphi|_{2,\alpha} \leqslant \epsilon^{\mu/2} |\varphi|_{2,\alpha} \leqslant c_k \epsilon^{1+\mu/2}.$$

From this inequality and from the estimates of the coefficients of  $\tilde{L}_{\epsilon}$ , it follows that

$$\|\tilde{L}_{\epsilon}(w_{\varphi})\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c\epsilon^{1/2} \|w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant c_k \epsilon^{(3+\mu)/2}.$$

As for  $\mathcal{L}_{\sigma}$  we consider the following relation

$$\mathcal{L}_{\sigma} w_{\varphi} = 2T w_{\varphi},$$

where  $T \leq 1$ . It comes from the definition of  $w_{\varphi}$  and Proposition 81 which gives us the following relation:

$$\partial_{uu}^2 w_\varphi + \partial_{vv}^2 w_\varphi = 0.$$

Therefore, we conclude that

$$\|\mathcal{L}_{\sigma} w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant 2\|w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant c_{k}\epsilon^{1+\mu/2}.$$

The last term is estimated by

$$\|\tilde{Q}_{\sigma}\left(w_{\varphi}\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}\left(I_{\sigma}\times\left[v_{\epsilon},\infty\right)\right)} \leqslant c_{k}\epsilon^{2+\mu/4}$$

In fact

$$\begin{split} \|\tilde{Q}_{\sigma}\left(w_{\varphi}\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}\left(I_{\sigma}\times\left[v_{\epsilon},\infty\right)\right)} &\leqslant c \sup_{v\in\left[v_{\epsilon},\infty\right)} e^{-\mu v} |w|^{2}_{0,\alpha;I_{\sigma}\times\left[v,v+1\right]} \leqslant c \|w\|^{2}_{2,\alpha,\mu/2} \leqslant c |\varphi|^{2}_{2,\alpha} \sup_{\left[v_{\epsilon},\infty\right)} e^{-\mu v/2} e^{\mu(v-v_{\epsilon})} \\ &\leqslant c |\varphi|^{2}_{2,\alpha} \sup_{\left[v_{\epsilon},\infty\right)} e^{+\mu v/2} e^{-\mu v_{\epsilon}} \leqslant c \epsilon^{+\mu/4} |\varphi|^{2}_{2,\alpha} \leqslant c_{k} \epsilon^{2+\mu/4}. \end{split}$$

Putting together these estimates we get the first result. As for the second estimate, we recall that

$$S(\varphi,g) := G_{\epsilon,v_{\epsilon}} \left( \tilde{L}_{\epsilon}(w_{\varphi}+g) - \mathcal{L}_{\sigma} w_{\varphi} + \tilde{Q}_{\sigma} (w_{\varphi}+g) \right).$$

Then

$$S(\varphi, g_2) - S(\varphi, g_1) = G_{\epsilon, v_{\epsilon}} \left( \tilde{L}_{\epsilon}(w_{\varphi} + g_2) - \mathcal{L}_{\sigma} w_{\varphi} + \tilde{Q}_{\sigma} (w_{\varphi} + g_2) \right) - G_{\epsilon, v_{\epsilon}} \left( \tilde{L}_{\epsilon}(w_{\varphi} + g_1) - \mathcal{L}_{\sigma} w_{\varphi} + \tilde{Q}_{\sigma} (w_{\varphi} + g_1) \right)$$

and

$$\begin{split} \|S(\varphi,g_2) - S(\varphi,g_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant c \|\tilde{L}_{\epsilon}(w_{\varphi}+g_2) - \mathcal{L}_{\sigma}w_{\varphi} + \tilde{Q}_{\sigma}(w_{\varphi}+g_2) - \\ & -\tilde{L}_{\epsilon}(w_{\varphi}+g_1) + \mathcal{L}_{\sigma}w_{\varphi} - \tilde{Q}_{\sigma}(w_{\varphi}+g_1) \|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} = \\ \|\tilde{L}_{\epsilon}(w_{\varphi}+g_2) - \tilde{L}_{\epsilon}(w_{\varphi}+g_1) + \tilde{Q}_{\sigma}(w_{\varphi}+g_2) - \tilde{Q}_{\sigma}(w_{\varphi}+g_1) \|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} = \\ & = \|\tilde{L}_{\epsilon}(g_2-g_1) + \tilde{Q}_{\sigma}(w_{\varphi}+g_2) - \tilde{Q}_{\sigma}(w_{\varphi}+g_1) \|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma}\times[v_{\epsilon},\infty))} \leqslant \end{split}$$

$$\leq \|\tilde{L}_{\epsilon}(g_2 - g_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} + \|\tilde{Q}_{\sigma}(w_{\varphi} + g_1) - \tilde{Q}_{\sigma}(w_{\varphi} + g_2)\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))}.$$

We observe that from the considerations above it follows that

$$\|\tilde{L}_{\epsilon}(g_2 - g_1)\|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \leqslant c_k \epsilon^{1/2} ||g_2 - g_1||_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))}$$

and that there is  $j + j' \ge 1$  such that:

$$\begin{split} \|\tilde{Q}_{\sigma} (w_{\varphi} + g_{1}) - \tilde{Q}_{\sigma} (w_{\varphi} + g_{2}) \|_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \leqslant \\ \leqslant c ||g_{2} - g_{1}||_{\mathcal{C}^{0,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \left( |w_{\varphi} + g_{2}|^{j} |w_{\varphi} + g_{1}|^{j'} \right) \leqslant \\ \leqslant c_{k} \left( \epsilon^{2+\mu/2} \right)^{j+j'} ||g_{2} - g_{1}||_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \leqslant c_{k} \epsilon^{4+2\mu} ||g_{2} - g_{1}||_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \cdot \\ \|S(\varphi, g_{2}) - S(\varphi, g_{1})\|_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \leqslant c_{k} \epsilon^{4+2\mu} ||g_{2} - g_{1}||_{\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon},\infty))} \cdot \end{split}$$

Then

**heorem 71.** Let be 
$$B := \{g \in \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty)) \mid ||g||\mathcal{C}^{2,\alpha}_{\mu} \leqslant c_k \epsilon^{1+\mu/2} \}$$
. Then the non-

**Theorem 71.** Let be  $B := \{g \in \mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty)) \mid ||g||\mathcal{C}^{2,\alpha}_{\mu} \leq c_k \epsilon^{1+\mu/2} \}$ . Then the nonlinear mapping S defined above has a unique fixed point g in B.

**Proof.** The previous lemma shows that, if  $\epsilon$  is chosen small enough, the nonlinear mapping S is a contraction mapping<sup>4</sup> from the ball B of radius  $c_k \epsilon^{1+\mu/2}$  in  $\mathcal{C}^{2,\alpha}_{\mu}(I_{\sigma} \times [v_{\epsilon}, \infty))$  into itself. This value comes from the estimate of the norm of  $S(\varphi, 0)$ . Consequently thanks to the Schauder theorem, S has a unique fixed point v in this ball.  $\Box$ 

This argument provides a minimal surface  $M_i(\phi)$  which is close to  $M_i$  and has one boundary. This surface is, close to its boundary, a vertical graph over the annulus  $B_{\epsilon^{-1/2}/2} - B_{\epsilon^{-1/2}/4}$  whose parametrization is given, for  $\alpha = 0$  by

$$U_{t,1}(r,\theta) = (1+\gamma)\ln\frac{2r}{1+\gamma} + r\eta_2\sin\theta + \frac{1}{r}(\kappa_1\cos\theta + \kappa_2\sin\theta) + \kappa_3 + \bar{\mathcal{H}}_{v_{\epsilon},\varphi}(v_{\epsilon} - \ln 2r,\theta) + V(r,\theta).$$

and, for  $\beta = 0$ , by

$$U_{t,2}(r,\theta) = (1+\gamma)\ln\frac{2r}{1+\gamma} + r\eta_1\cos\theta + \frac{1}{r}(\kappa_1\cos\theta + \kappa_2\sin\theta) + \kappa_3 + \bar{\mathcal{H}}_{v_{\epsilon},\varphi}(v_{\epsilon} - \ln 2r,\theta) + V(r,\theta) +$$

where  $v_{\epsilon} = -\frac{1}{2} \ln \epsilon$ . The boundary of the surface corresponds to  $r_{\epsilon} = \frac{1}{2} \epsilon^{-1/2}$ . The function V depends non linearly on  $\epsilon, \phi$ . It satisfies  $\|V(\epsilon, \phi_i)(r_{\epsilon})\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1-B_{1/2})} \leq c\epsilon$  and

$$\|V_i(\epsilon,\phi)(r_{\epsilon}\cdot) - V_i(\epsilon,\phi')(r_{\epsilon}\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_1 - B_{1/2})} \leq c\epsilon^{1-\mu/2} \|\phi - \phi'\|_{\mathcal{C}^{2,\alpha}(I_{\sigma})}.$$

<sup>&</sup>lt;sup>4</sup>after the correct choise of the constant k that appears in the estimate of the norm of  $\varphi$ .

#### 4.7 Periodic flat cylinder

In this section we are interested in finding an infinite family of minimal surface close to a horizontal strip from which we have removed a circle. We shall use the following model. We denote by  $\Sigma = \{x + iy \in \mathbb{C}; y \in [-\frac{\pi}{\eta}, \frac{\pi}{\eta}]\}$ , with  $\eta > 0$  and enough small, the horizontal flat cylinder whose embedding in  $\mathbb{R}^3/T$  is X(z) = (z, 0), where  $T = \frac{2\pi}{\eta}e_2$  is the period. Topologically  $\Sigma$  is equivalent to  $\mathbb{R} \times S^1$ . We denote  $B_s$  the ball of radius *s* centered in the origin. The equation to consider is

$$\operatorname{div}\frac{\nabla u}{\sqrt{1+|\nabla u|^2}} = 0 \tag{4.32}$$

for u defined on  $\Sigma - B_s$  and with boundary data on  $\partial B_s$ .

We define the subdomains  $\Omega_{x_1} = \{z \in \Sigma; |x| \leq x_1\}$  and denote the two cylindrical ends with  $E_1 = \{z \in \Sigma; x \geq x_1\}, E_2 = \{z \in \Sigma; x \leq -x_1\}.$ 

**Definition 72.** Given  $k \in \mathbb{N}$ ,  $\alpha \in (0,1)$  and  $\mu \in \mathbb{R}$ , we define the weighted Hölder space  $C^{k,\alpha}_{\mu}(\Sigma)$  to be the space of functions of  $C^{k,\alpha}_{loc}(\Sigma)$  for which the following norm is finite

$$||u||_{C^{k,\alpha}_{\mu}} := [u]_{k,\alpha;\Omega_{x_1}} + \sup_{x \geqslant x_1} e^{-\mu x} ([u]_{k,\alpha,([x,x+1] \times S^1)} + [u]_{k,\alpha,([-x-1,-x] \times S^1)})$$

where  $[u]_{k,\alpha,\Omega}$  denotes the usual  $C^{k,\alpha}$  Hölder norm on the set  $\Omega$ .

We denote by  $C^{k,\alpha}_{\mu}(\Sigma - B_s)$  the subspace of the functions of  $C^{k,\alpha}_{\mu}(\Sigma)$  restricted to  $\Sigma - B_s$ and by  $[C^{k,\alpha}_{\mu}(\Sigma - B_s)]_0$  the subspace of the functions vanishing on the boundary.

There exists an extension operator  $\mathcal{E}_s : C^{0,\alpha}_{\mu}(\Sigma - B_s) \to C^{0,\alpha}_{\mu}(\Sigma)$  which satisfies  $||\mathcal{E}_s(u)|| \leq C||u||$ . The operator is defined by  $\mathcal{E}_s(u) = u$  on  $\Sigma - B_s$  and  $\mathcal{E}_s(u) = (2|z|/s - 1)u(z)$  on  $B_s - B_{s/2}$  and  $\mathcal{E}_s(u) = 0$  on  $B_{s/2}$ .

Now we consider the bounded operator  $\Delta_{\mu}$ :

$$\Delta: C^{2,\alpha}_{\mu}(\Sigma) \longrightarrow C^{0,\alpha}_{\mu}(\Sigma)$$

where  $\Delta = \partial_y^2 + \partial_x^2$ . It follows from the general theory of elliptic partial differential operators that  $\Delta_{\mu}$  is a Fredholm operator for all  $\mu \notin \mathbb{Z}$ . It is well known that, if  $\mu \in \mathbb{R} - \mathbb{Z}$ , then  $\Delta_{\mu}$  is injective if and only if  $\Delta_{-\mu}$  is surjective and the dimension of the kernel of  $\Delta_{-\mu}$  is equal to the cokernel of  $\Delta_{\mu}$ . Furthermore, it is possible to show that the operator  $\Delta_{\mu}$  is injective for  $-1 < \mu < 0$  and so it is surjective for  $0 < \mu < 1$ . When the weight parameter is negative, we can still make the operator surjective by considering a finite dimensional extension of the Hölder space defined above. To be more precise we need additional notation. Let us set  $x_1 = 2s$  such that  $B_s \subset \{-x_1 < x < x_1\}$ . We introduce the cut-off functions  $\xi_1$  and  $\xi_2$  satisfying the following conditions:

- $\xi_1 = 0$  for  $x < x_1$  and  $\xi_1 = 1$  for  $x > 2x_1$ ,
- $\xi_2 = 0$  for  $x > -x_1$  and to  $\xi_2 = 1$  for  $x < -2x_1$ .

Now we are ready to define

$$\mathcal{D}_0 := \operatorname{Span}\{\xi_1, \xi_2, \xi_1 x, \xi_2 x\}.$$

This space is identified with  $\mathbb{R}^4$  and is endowed with the Euclidean norm. For  $-1 < \mu < 0$ we have Ker  $\Delta_{-\mu} \subset C^{2,\alpha}_{\mu}(\Sigma) \oplus \mathcal{D}_0$ . In fact if we expand a solution of  $\Delta u = 0$  on  $\{x > x_1\}$  by

$$u = \sum_{n \in \mathbb{Z}} u_n(x) e^{i\frac{2\pi}{T}ny},$$

where T is the period with respect to the variable y, then  $u = u_0 + \bar{u}$ , with  $\bar{u} \in C^{2,\alpha}_{\mu}$  with  $-1 < \mu < 0$ , and  $u_0$  affine function, i.e. a linear combination of the constant function and x near the cylindrical ends  $E_1$  and  $E_2$ . The index theory assures that the kernel is two dimensional. We define  $K_0 = \text{Span}\{\xi_1, \xi_2\}$ . The linear decomposition Lemma proved in [23] for constant mean curvature surfaces (see also [18] for minimal hypersurfaces) can be adapted to our situation. So  $D_0 = K_0 \oplus N_0$ , where  $N_0$  denotes a complementary space. Then

$$\ker \Delta_{-\mu} \subset C^{2,\alpha}_{\mu}(\Sigma) \oplus N_0$$

and the application

$$\Delta: C^{2,\alpha}_{\mu}(\Sigma) \oplus K_0 \longrightarrow C^{0,\alpha}_{\mu}(\Sigma)$$

is an isomorphism.

**Proposition 73.** If  $0 < \mu < \epsilon^2$  and  $T = \frac{2\pi}{\eta}$  then there exists an operator

$$F_{\mu}: \mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \to \mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0$$

such that for all  $f \in C^{0,\alpha}_{\mu}(\Sigma)$ , the function  $v := F_{\mu}(f) = w + a_1\xi_1 + a_2\xi_2$  solves  $\Delta v = f$  in  $\Sigma$ . Moreover,

$$||F_{\mu}(f)||_{\mathcal{C}^{2,\alpha}_{\mu}\oplus K_{0}} = |a_{1}| + |a_{2}| + ||w||_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c \, ||f||_{\mathcal{C}^{0,\alpha}_{\mu}},$$

for some constant c > 0.

*Proof.* We solve the equation  $\Delta v = f$  at the ends  $E_1$  and  $E_2$  of  $\Sigma$ . For  $x > x_1$  we consider the Fourier series of f and v (a similar argument holds for  $E_2$ ).

$$v = \sum_{n \in \mathbb{Z}} v_n e^{i\frac{2\pi}{T}ny}$$
 and  $f = \sum_{n \in \mathbb{Z}} f_n e^{i\frac{2\pi}{T}ny}$ 

For  $n \neq 0$ , applying the barrier functions method we find that the solution  $v_n$  satisfies

$$|v_n(x)| \leq \frac{||f||_{\mathcal{C}^{0,\alpha}_{\mu}}}{\left(\frac{2\pi}{T}\right)^2 n^2 - \mu^2} e^{\mu x},$$

and for  $n = 0, v_n(x)$  is given by

$$v_0(x) = \int_x^\infty \int_t^\infty f_0(s) ds dt.$$

We can conclude that  $e^{-\mu x}|v| \leq c||f||_{\mathcal{C}^{0,\alpha}_{\mu}}$  and applying Schauder estimates we obtain

$$||\xi_1 v||_{\mathcal{C}^{2,\alpha}_{\mu}} \leqslant c||f||_{\mathcal{C}^{0,\alpha}_{\mu}}.$$

Obviously we can apply the same argument for  $E_2$ . We denote by  $v_1, v_2$  the solutions corresponding to  $E_1, E_2$ , we define  $g = \xi_1 v_1 + \xi_2 v_2$  and we set  $\overline{f} = f - \Delta g$ . We parametrize  $\Sigma$  conformally on  $\mathbb{C}^*$  and we solve  $\Delta w = \overline{f}$  on  $\mathbb{C} \cup \{\infty\}$ . Then the function v = w + gsatisfies  $\Delta v = f$ .

It is possible to show the following result.

**Proposition 74.** Let  $\phi \in C^{2,\alpha}(\partial B_s)$  a function  $L^2$ -orthogonal to  $z \to 1$ , then there exists an operator  $\mathcal{H}$  such that  $w_{\phi} = \mathcal{H}_{\phi} \in C^{2,\alpha}_{-2}(\Sigma - B_s) \oplus \operatorname{Span}\{\xi_1, \xi_2\}$  which solves the following problem

$$\begin{cases} \Delta w_{\phi} = 0 \quad on \quad \Sigma - B_s \\ w_{\phi} = \phi \quad on \quad \partial B_s. \end{cases}$$

and satisfies  $\|\mathcal{H}_{\phi}\|_{C^{2,\alpha}_{-2}(\Sigma-B_s)\oplus K_0} \leq c \|\phi\|_{C^{2,\alpha}}.$ 

In the following we consider  $s = \bar{s} = \frac{1}{2\sqrt{\epsilon}}$ . We recall that we have set  $x_1 = 2s$ .

Let  $\phi(\theta)$  be a  $C^{2,\alpha}$  function defined on  $\partial B_{\bar{s}}$ , even and  $L^2$  orthogonal to the costant function and to  $\theta \to \cos \theta$  and such that

$$||\phi||_{\mathcal{C}^{2,\alpha}(S^1)} \leqslant \kappa \epsilon \tag{4.33}$$

and  $w_{\phi} = \mathcal{H}_{\phi}$  the harmonic extension in  $C^{2,\alpha}_{\mu}(\Sigma - B_{\bar{s}}) \oplus \text{Span}\{\xi_1, \xi_2\}$  obtained applying Proposition 74. Our aim is to find a minimal surface close to  $\Sigma - B_{\bar{s}}$  and which is the graph of a function whose form is  $w_{\phi} + v$ . The equation (4.32) can be written in the following form

$$\Delta v = Q(v + w_{\phi}) \text{ on } \Sigma - B_{\bar{s}},$$

where

$$Q(v + w_{\phi}) = \frac{\nabla(v + w_{\phi})\nabla|\nabla(v + w_{\phi})|^2}{2(1 + |\nabla(v + w_{\phi})|^2)}.$$

When  $v \in \mathcal{C}^{2,\alpha}_{\mu}(\Sigma - B_{\bar{s}}) \oplus K_0$  then  $\nabla v \in C^{1,\alpha}_{\mu}(\Sigma - B_{\bar{s}}) \oplus K_0$  and  $Q(v) \in C^{0,\alpha}_{\mu}(\Sigma - B_{\bar{s}}) \oplus K_0$ . We rephrase the problem as a fixed point problem, that is

$$v = T(\phi, v) \tag{4.34}$$

where

$$T(\phi, v) = F_{\mu} \circ \mathcal{E}_{\bar{s}}(Q(v + w_{\phi})).$$

To prove the existence of a solution of (4.34) we need the following result which states that T is a contracting mapping.

**Lemma 75.** There exist constants  $c_{\kappa} > 0$  and  $\epsilon_{\kappa} > 0$ , such that

$$\|T(\phi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}\oplus K_0} \leqslant c_{\kappa}\,\epsilon^{\frac{3}{2}} \tag{4.35}$$

and, for all  $\epsilon \in (0, \epsilon_{\kappa})$ 

$$\|T(\phi, v_2) - T(\phi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0} \leqslant \frac{1}{2} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0}$$

and

$$\|T(\phi_2, v_2) - T(\phi_1, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0} \leq c_{\kappa} \,\epsilon \,\|\phi_2 - \phi_1\|_{\mathcal{C}^{2,\alpha}(S^1)}$$

for all  $v, v_1, v_2 \in \mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0$  whose norm is bounded by  $2 c_{\kappa} \epsilon^{\frac{3}{2}}$  and for all boundary data  $\phi, \phi_1, \phi_2$  that are orthogonal to the constant function and to  $\theta \to \cos \theta$  and satisfy (4.33).

**Proof.** We use the result of Proposition 74 to obtain the estimate

$$\|\mathcal{E}_{\bar{s}}(Q(w_{\phi}))\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leqslant \|w_{\phi} \circ E_{1}\|^{2}_{\mathcal{C}^{0,\alpha}_{\mu/2}(\Sigma)} + c\|w_{\phi}\|^{2}_{K_{0}} \leqslant c\epsilon^{\frac{3}{2}}.$$

As for the second estimate, we recall that

$$T(\varphi, v) := F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( w_{\varphi} + v \right) \right).$$

Then

$$T(\varphi, v_2) - T(\varphi, v_1) = F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( w_{\varphi} + v_2 \right) \right) - F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( w_{\varphi} + v_1 \right) \right)$$

and

$$\|T(\varphi, v_2) - T(\varphi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_1\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_2\right) - Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leq \|\mathcal{E}_{\bar{s}}\left(Q\left(w_{\varphi} + v_2\right)\right)\|_{\mathcal{C}^{0,$$

$$\leqslant c \|w_{\varphi}\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma)\oplus K_0} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma)\oplus K_0} \leqslant c_k \epsilon^{\frac{1}{2}} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma)\oplus K_0}.$$

 $\square$ 

The previous lemma shows that, provided  $\epsilon$  is chosen small enough, the nonlinear mapping  $T(\phi, \cdot)$  is a contraction mapping from the ball of radius  $2c_{\kappa} \epsilon^{\frac{3}{2}}$  in  $C^{2,\alpha}_{\mu}(\Sigma - B_{\bar{s}})$  into itself. Consequently the equation (4.34) admits a solution v in this ball. The graph over  $\Sigma - B_{\bar{s}}$  for the function  $w_{\phi} + v$  is a minimal surface close to  $\Sigma - B_{\bar{s}}$ , it has two horizontal ends and one boundary. This surface is by construction a vertical graph over the annulus  $\bar{B}_{2\bar{s}} - B_{\bar{s}}$  for some function  $\bar{U}_B$  that can be expanded as

$$\bar{U}_B(r,\theta) = \mathcal{H}_\phi(r,\theta) + V_B(r,\theta)$$

where  $\tilde{V}_B(\epsilon, \varphi)$  depends nonlinearly on  $\epsilon$  and  $\phi$ . The boundary of the surface corresponds to  $r = r_{\epsilon} = \frac{1}{2\sqrt{\epsilon}}$ . Furthermore the following estimates hold

$$\|V_B(\epsilon,\varphi)(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)}\leqslant c\,\epsilon$$

and

$$\tilde{V}_B(\epsilon,\phi)(r_\epsilon\,\cdot) - \tilde{V}_B(\epsilon,\phi')(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)} \leqslant c\,\epsilon^{\frac{1}{2}} \,\|\phi - \phi'\|_{\mathcal{C}^{2,\alpha}(S^1)} \tag{4.36}$$

where the constant c > 0 does not depend on  $\epsilon$ .

# 4.8 The existence of minimal graph close to a Scherk type surface.

In this section we are interested in proving the existence of a family of minimal surfaces close to a Scherk type surface K defined on  $\Sigma - B_{r_{\epsilon}}$  where  $r_{\epsilon} = 1/2\sqrt{\epsilon}$ . We solve the following equivalent problem. Let  $\bar{\Sigma} = \{x + iy \in \mathbb{C}; y \in [-\pi, \pi]\}$  the horizontal flat cylinder whose embedding in  $\mathbb{R}^3/T$  is X(z) = (z, 0), where  $T = 2\pi e_2$  is the period. Topologically  $\bar{\Sigma}$  is equivalent to  $\mathbb{R} \times S^1$ . We denote by  $B_s$  the ball of radius *s* centered in the origin. We denote by *X* the solution of

$$\Delta u = 0 \quad \text{in} \quad \bar{\Sigma} - B_s, \tag{4.37}$$

where  $\delta_0$  is the Dirac distribution, such that, up to an additive constant, in a neighbourhood of the point z = 0 is asymptotic to  $\epsilon \ln r$  and in  $\{x > x_k\}$ , with k enough big, is asymptotic to a function which is linear in the variable x. We want to show the existence of a minimal surface which is the graph of the function given by X + w about  $\overline{\Sigma} - B_s$ . The equation to consider is

$$\operatorname{div}\frac{\nabla u}{\sqrt{1+|\nabla u|^2}} = 0 \tag{4.38}$$

for u defined on  $\overline{\Sigma} - B_s$  and with boundary data on  $\partial B_s$ .

Let  $\phi(\theta)$  be a  $C^{2,\alpha}$  function defined on  $\partial B_{\bar{s}}$ , even and  $L^2$  orthogonal to the costant function and to  $\theta \to \cos \theta$  and such that

$$\|\phi\|_{\mathcal{C}^{2,\alpha}(S^1)} \leqslant \kappa \epsilon^2 \tag{4.39}$$

and  $w_{\phi} = \mathcal{H}_{\phi}$  the harmonic extension in  $C^{2,\alpha}_{\mu}(\bar{\Sigma} - B_{\bar{s}}) \oplus \text{Span}\{\xi_1, \xi_2\}$  obtained applying Proposition 74. Our aim is to find a minimal surface close to  $\bar{\Sigma} - B_{\bar{s}}$  and which is the graph of a function whose form is  $X + w_{\phi} + v$ . The equation (4.38) can be written in the following form

$$\Delta v = Q(X + v + w_{\phi}) \text{ on } \bar{\Sigma} - B_{\bar{s}},$$

where

$$Q(X + w_{\phi} + v) = \frac{\nabla (X + w_{\phi} + v)\nabla |\nabla (X + w_{\phi} + v)|^2}{2(1 + |\nabla (X + w_{\phi} + v)|^2)}$$

When  $t \in \mathcal{C}^{2,\alpha}_{\mu}(\bar{\Sigma} - B_{\bar{s}}) \oplus K_0$  then  $\nabla t \in C^{1,\alpha}_{\mu}(\bar{\Sigma} - B_{\bar{s}}) \oplus K_0$  and  $Q(t) \in C^{0,\alpha}_{\mu}(\bar{\Sigma} - B_{\bar{s}}) \oplus K_0$ . We rephrase the problem as a fixed point problem, that is

$$v = T(\phi, v) \tag{4.40}$$

where

$$T(\phi, v) = F_{\mu} \circ \mathcal{E}_{\bar{s}}(Q(X + w_{\phi} + v))$$

To prove the existence of a solution of (4.40) we need the following result which states that T is a contracting mapping.

**Lemma 76.** There exist constants  $c_{\kappa} > 0$  and  $\epsilon_{\kappa} > 0$ , such that

$$\|T(\phi,0)\|_{\mathcal{C}^{2,\alpha}_{\mu}\oplus K_0} \leqslant c_{\kappa} \epsilon^{\frac{3}{2}}$$

$$\tag{4.41}$$

and, for all  $\epsilon \in (0, \epsilon_{\kappa})$ 

$$\|T(\phi, v_2) - T(\phi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0} \leq \frac{1}{2} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0}$$

and

$$\|T(\phi_2, v_2) - T(\phi_1, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu} \oplus K_0} \leq c_{\kappa} \,\epsilon \, \|\phi_2 - \phi_1\|_{\mathcal{C}^{2,\alpha}(S^1)}$$

for all  $v, v_1, v_2 \in \mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0$  whose norm is bounded by  $2 c_{\kappa} \epsilon^{\frac{3}{2}}$  and for all boundary data  $\phi, \phi_1, \phi_2$  that are orthogonal to the constant function and to  $\theta \to \cos \theta$  and satisfy (4.39).

**Proof.** We use the result of Proposition 74 to obtain the estimate

$$\|\mathcal{E}_{\bar{s}}(Q(X+w_{\phi}))\|_{\mathcal{C}^{0,\alpha}_{\mu}} \leq \|(X+w_{\phi}) \circ E_{1}\|^{2}_{\mathcal{C}^{0,\alpha}_{\mu/2}(\Sigma)} + c\|(X+w_{\phi})\|^{2}_{K_{0}} \leq c\epsilon^{\frac{3}{2}}.$$

As for the second estimate, we recall that

$$T(\varphi, v) := F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( X + w_{\phi} + v \right) \right).$$

Then

$$T(\phi, v_2) - T(\phi, v_1) = F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( X + w_{\phi} + v_2 \right) \right) - F_{\mu} \circ \mathcal{E}_{\bar{s}} \left( Q \left( X + w_{\phi} + v_1 \right) \right)$$

and

$$\|T(\phi, v_2) - T(\varphi, v_1)\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0} \leqslant \|\mathcal{E}_{\bar{s}} \left(Q \left(X + w_{\phi} + v_1\right) - Q \left(X + w_{\phi} + v_2\right)\right)\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \leqslant c \|X + w_{\phi}\|_{\mathcal{C}^{0,\alpha}_{\mu}(\Sigma) \oplus K_0} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0} \leqslant c_k \epsilon^{\frac{1}{2}} \|v_2 - v_1\|_{\mathcal{C}^{2,\alpha}_{\mu}(\Sigma) \oplus K_0}.$$

The previous lemma shows that, provided  $\epsilon$  is chosen small enough, the nonlinear mapping  $T(\phi, \cdot)$  is a contraction mapping from the ball of radius  $2c_{\kappa} \epsilon^{\frac{3}{2}}$  in  $\mathcal{C}^{2,\alpha}_{\mu}(\bar{\Sigma} - B_{\bar{s}})$  into itself. Consequently the equation (4.40) admits a solution v in this ball. The graph over  $\bar{\Sigma} - B_{\bar{s}}$  for the function  $X + w_{\phi} + v$  is a minimal surface close to the graph of the function X about  $\bar{\Sigma} - B_{\bar{s}}$ , it has a Scherk type end and one boundary. This surface is by construction a vertical graph over the annulus  $\bar{B}_{2\bar{s}} - B_{\bar{s}}$  for some function  $\bar{U}_K$  that can be expanded as

$$\overline{U}_K(r,\theta) = \epsilon \ln r + \mathcal{H}_{\phi}(r,\theta) + V_K(r,\theta)$$

where  $\tilde{V}_K(\epsilon, \phi)$  depends nonlinearly on  $\epsilon$  and  $\phi$ . The boundary of the surface corresponds to  $r = \sqrt{\epsilon/2}$ . Furthermore the following estimates hold

$$\|V_K(\epsilon,\phi)(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)}\leqslant c\,\epsilon$$

and

$$\|\tilde{V}_K(\epsilon,\phi)(r_\epsilon\,\cdot) - \tilde{V}_K(\epsilon,\phi')(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2 - B_1)} \leqslant c\,\epsilon^{\frac{1}{2}} \,\|\phi - \phi'\|_{\mathcal{C}^{2,\alpha}(S^1)} \tag{4.42}$$

where the constant c > 0 does not depend on  $\epsilon$ .

By the equations written above it is easy to get the solution of our initial problem: that is finding a Scherk type minimal graph  $S_K$  about  $\Sigma - B_{r_{\epsilon}}$  with Dirichlet condition on  $\partial B_{r_{\epsilon}}$ . This surface is by construction a vertical graph over the annulus  $\bar{B}_{2r_{\epsilon}} - B_{r_{\epsilon}}$  for some function  $U_K$  that can be expanded as

$$U_K(r,\theta) = \ln r + \mathcal{H}_{\varphi}(r,\theta) + V_K(r,\theta)$$

where  $V_K(\epsilon, \varphi)$  depends nonlinearly on  $\epsilon$  and  $\varphi$ . It is important to remark that the norm of the Dirichlet data satisfies  $\|\varphi\|_{\mathcal{C}^{2,\alpha}(S^1)} \leq \kappa \epsilon$ . The boundary of the surface corresponds to  $r = r_{\epsilon} = \frac{1}{2\sqrt{\epsilon}}$ . Furthermore the following estimates hold

$$\|V_K(\epsilon,\varphi)(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)} \leqslant c\,\epsilon$$

and

$$\|V_K(\epsilon,\varphi)(r_\epsilon\,\cdot) - V_K(\epsilon,\varphi')(r_\epsilon\,\cdot)\|_{\mathcal{C}^{2,\alpha}(\bar{B}_2-B_1)} \leqslant c\,\epsilon^{\frac{1}{2}} \,\|\varphi - \varphi'\|_{\mathcal{C}^{2,\alpha}(S^1)}$$
(4.43)

where the constant c > 0 does not depend on  $\epsilon$ .

#### 4.9 The matching of Cauchy data

In the following we will need a new notation. Given an even function  $f \in \mathcal{C}^{2,\alpha}(S^1)$  with the following Fourier expansion

$$f(\theta) = \sum_{n \in \mathbb{N}} a_n \cos(n\theta),$$

then we denote with  $\pi''(f)$  the function

$$\sum_{n \geqslant 2} a_n \cos(n\theta)$$

and with  $\pi'(f)$  the function

$$a_0 + a_1 \cos(\theta).$$

Along this section  $r_{\epsilon} = \frac{1}{2\sqrt{\epsilon}}$  and  $\varphi$  is a function in  $\pi'' C^{2,\alpha}(S^1)$  such that  $\|\varphi\|_{C^{2,\alpha}} \leq c\epsilon$ .

In Section 4.7 we have shown the existence of a surface which is a graph on  $\Sigma - B_{r_{\epsilon}}$  and is asymptotic to it. In other words we are able to solve the minimal surface equation, that we will write for short as follows

$$\begin{cases} L_p u^0 = Q_{\Sigma}(u^0) \text{ on } \Sigma - B_{r_{\epsilon}} \\ \pi'' u^0 = \varphi \text{ on } \partial B_{r_{\epsilon}} \end{cases}$$

where  $L_p$  denotes the linearized mean curvature operator about the plane.

So chosen a function  $\varphi$ , we can obtain a surface  $S^0$ , whose parametrization  $X_{0,\varphi}$ , in a neighbourhood of its boundary (a curve defined on  $\partial B_{r_{\epsilon}}$ ), satisfies

$$X_{0,\varphi} = \varphi + \pi'(u^0)$$
$$\partial_r X_{0,\varphi} = \partial_r u^0$$

In Section 4.6 we showed how to obtain a surface which is a graph on  $\widetilde{M}^+_{\sigma,\alpha,0}$  (or  $\widetilde{M}^+_{\sigma,0,\beta}$ ) and is asymptotic to it. We have solved a problem of the form

$$\begin{cases} \mathcal{L}_{\sigma}u^{+} = Q_{\sigma}(u^{+}) & \text{on} \quad \widetilde{M}^{+}_{\sigma,\alpha,0} \quad (\widetilde{M}^{+}_{\sigma,0,\beta}) \\ \pi''u^{+} = \varphi - \pi''g_{t} & \text{on} \quad \partial\widetilde{M}^{+}_{\sigma,\alpha,0} \quad (\partial\widetilde{M}^{+}_{\sigma,0,\beta}) \end{cases}$$

where  $\mathcal{L}_{\sigma}$  denotes the Lamé operator and  $g_t = \mathcal{O}(\epsilon)$ .

So we can obtain two surfaces  $S^+_{\alpha,0}$  and  $S^+_{0,\beta}$ , whose parametrizations  $X^+_{\alpha,\varphi}$ ,  $X^+_{\beta,\varphi}$  satisfy in a neighbourhood of its boundary

$$\begin{split} X^+_{\alpha,\varphi} &= -(1+\kappa)\ln(2r) + 2\sin\frac{\alpha}{2}r\cos\theta + \varphi + \pi'(u^+ + g_t),\\ \partial_r X^+_{\alpha,\varphi} &= -(1+\kappa)\frac{1}{r} + 2\sin\frac{\alpha}{2}\cos\theta + g_{t,d} + \partial_r u^+,\\ X^+_{\beta,\varphi} &= -(1+\kappa)\ln(2r) + 2\sin\frac{\beta}{2}r\sin\theta + \varphi + \pi'(u^+ + g_t),\\ \partial_r X^+_{\beta,\varphi} &= -(1+\kappa)\frac{1}{r} + 2\sin\frac{\beta}{2}\sin\theta + g_{t,d} + \partial_r u^+, \end{split}$$

where  $g_{t,d} = \mathcal{O}(\epsilon)$ . In the following to simplify the notation we will write for short  $S^+$  and  $X_{+,\varphi}$ .

Analogously, in Section 4.8 we showed the existence of a surface close to a Scherk type surface, whose boundary is a curve on  $\partial B_{r_{\epsilon}}$ . In other words we are able to solve a problem of the form

$$\begin{cases} \mathcal{L}_M u^- &= Q(u^-) \quad \text{on} \quad \Sigma - B_{r_{\epsilon}} \\ \pi'' u^- &= \varphi - \pi'' g_b \quad \text{on} \quad \partial B_{r_{\epsilon}} \end{cases}$$

where  $\mathcal{L}_M$  denotes the linearized mean curvature operator and  $g_b = \mathcal{O}(\epsilon)$ .

So we can obtain a surface  $S^-$  that has a parametrization  $X_{-,\varphi}$ , which in a neighbourhood of its boundary satisfies

$$X_{-,\varphi} = -\ln(2r) + \varphi + \pi'(u^- + g_b),$$
  
$$\partial_r X_{-,\varphi} = -\frac{1}{r} + g_{b,d} + \partial_r u^-$$

where  $g_{b,d} = \mathcal{O}(\epsilon)$ .

Finally, in Section 4.3 we have obtained the surface  $M_k(\epsilon/2, \Phi)$ , whose boundaries are curves about  $\partial B_{r_{\epsilon}}$ . In particular we are able to solve, for  $\Phi = (\varphi_t, \varphi_b, \varphi_m) \in (\pi'' C^{2,\alpha}(S^1))^3$  and  $\|\varphi_i\|_{C^{2,\alpha}} \leq \epsilon$  the problem

$$\begin{cases} L_{M_k(\epsilon/2)}u_C = Q(u_C) & \text{on} \quad M_k^T(\epsilon/2) \\ \pi''u_C = (\varphi_t - \pi''f_t, \varphi_b - \pi''f_b, \varphi_m - \pi''f_m) & \text{on} \quad \partial M_k^T(\epsilon/2). \end{cases}$$

The functions  $f_t$ ,  $f_b$ ,  $f_m$  denote the higher order terms appearing in lemma 53. This result gives the parametrizations of part of the ends of  $M_k(\epsilon/2)$  seen as graphs of appropriate functions over the  $x_3 = 0$  plane.  $f_{t,d}$ ,  $f_{b,d}$ ,  $f_{m,d}$  denote their derivatives. As for the catenoidal type ends these functions, that are defined on a neighbourhood of  $\partial B_{r_{\epsilon}}$ , have the following expressions

$$U_t(r,\theta) = \sigma_t + \ln(2r) + \frac{\epsilon}{2}r\cos\theta + f_t(r,\theta), \qquad (4.44)$$

$$\partial_r U_t(r,\theta) = \frac{1}{r} + \frac{\epsilon}{2}\cos\theta + f_{t,d}(r,\theta),$$

$$U_b(r,\theta) = -\sigma_b - \ln(2r) + \frac{\epsilon}{2}r\cos\theta + f_b(r,\theta),$$

$$\partial_r U_b(r,\theta) = -\frac{1}{r} + \frac{\epsilon}{2}\cos\theta + f_{b,d}(r,\theta),$$
(4.45)

where  $f_i = \mathcal{O}(\epsilon)$ ,  $f_{i,d} = \mathcal{O}(\epsilon^{3/2})$ , i = t, b. As for the parametrization of the planar end, it satisfies

$$U_m(r,\theta) = f_m(r,\theta) = \mathcal{O}(r^{-k}), \qquad (4.46)$$
$$\partial_r U_m(r,\theta) = f_{m,d}(r,\theta) = \mathcal{O}(r^{-k+1}),$$

in a neighbourhood of  $\partial B_{r_{\epsilon}}$ .

Then we can obtain in particular a minimal graph over the planar end whose parametrization,  $X_{m,\Phi}$  in a neighbourhood of its boundary, satisfies

$$X_{m,\Phi} = \varphi_m + \pi'(u_C + f_m)$$
$$\partial_r X_{m,\Phi} = f_{m,d} + \partial_r u_C.$$

Then we can obtain in particular a minimal graph over the top end whose parametrization,  $X_{t,\Phi}$  in a neighbourhood of its boundary, satisfies

$$X_{t,\Phi} = -\ln(2r) + \frac{\epsilon}{2}r\cos\theta + \varphi_t + \pi'(u_C + f_t)$$
$$\partial_r X_{t,\Phi} = -\frac{1}{r} + \frac{\epsilon}{2}\cos\theta + f_{t,d} + \partial_r u_C.$$

Now we can define

$$E_{\epsilon} : (\pi''C^{2,\alpha}(S^1))^3 \longrightarrow (C^{2,\alpha}(S^1))^3 \times (C^{1,\alpha}(S^1))^3$$
  
$$\Phi = (\phi_t, \phi_b, \phi_m) \longrightarrow [(X_{+,\phi_t}, X_{-,\phi_b}, X_{0,\phi_m}), (\partial_r X_{+,\phi_t}, \partial_r X_{-,\phi_b}, \partial_r X_{0,\phi_m})]_{|\partial B_{r\epsilon}}.$$

and

$$\begin{array}{cccc} F_{\epsilon}: \pi''C^{2,\alpha}(S^1) & \longrightarrow & C^{2,\alpha}(S^1) \times C^{1,\alpha}(S^1) \\ \Phi = (\phi_t, \phi_b, \phi_m) & \longrightarrow & [(X_{t,\phi_t}, X_{b,\phi_b}, X_{m,\phi_m}), (\partial_r X_{t,\phi_t}, \partial_r X_{b,\phi_b}, \partial_r X_{m,\phi_m})]_{|\partial B_{r_{\epsilon}}}. \end{array}$$

We set  $C_{\epsilon} := E_{\epsilon} - F_{\epsilon}$ .

We want to prove that the surfaces  $S^+$ ,  $S^-$ ,  $S^0$  and  $M_k^T(\epsilon/2, \Phi)$  can be glued along their boundaries to obtain a  $C^{\infty}$ -surface. Firstly we will show that these surface correspond in a  $C^1$  way along the boundaries curves. This is true if it exists  $\Psi = (\psi_1, \psi_2, \psi_3)$  such that  $C_{\epsilon}(\Psi) = 0$ . The existence of the appropriates boundary functions is proven in the following theorem. Finally, to show that the surface is  $C^{\infty}$ , it is sufficient to apply the regularity theory. **Theorem 77.** For some  $\epsilon_0$  and every  $0 < \epsilon < \epsilon_0$ , there exists  $\Psi = (\psi_1, \psi_2, \psi_3) \in [\pi'' C^{2,\alpha}(S^1)]^3$  which solves  $C_{\epsilon}(\Psi) = 0$ .

**Proof.** We consider the harmonic extensions of  $\psi_i$ , i = 1, 2, 3, on the ends of  $M_k(\xi, \epsilon)$ , that is

- 1.  $\bar{w}_t = \chi_+ H_{\psi_1}(s_{\epsilon} s, \cdot)$ , on the upper end;
- 2.  $\bar{w}_b = \chi_- H_{\psi_2}(s s_{\epsilon}, \cdot)$ , on the lower end;
- 3.  $\bar{w}_m = \chi_p \tilde{H}_{\rho_{\epsilon},\psi_3}(1/r,\cdot)$  on the middle end

and its harmonic extensions

- 1.  $w_t = \bar{H}_{v_{\epsilon},\psi_1}$  on  $S^+$ ;
- 2.  $w_b = -\bar{H}_{-v_{\epsilon},\psi_2}$  on  $S^-$ ;
- 3.  $w_m = \mathcal{H}_{\psi_3}$  on  $S^0$

(see Section 4.3 for the definitions of the cut-off functions). We recall that the operators  $\tilde{H}$ , H,  $\bar{\mathcal{H}}$  and  $\mathcal{H}$  have been introduced respectively in Propositions 79, 80, 81 and 74. We consider the following maps

$$\begin{aligned} E_0 &: [\pi'' C^{2,\alpha}(S^1)]^3 &\longrightarrow C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3 \\ \Psi &\longrightarrow [(w_t, w_b, w_m), (\partial_r w_t, \partial_r w_b, \partial_r w_m)]_{|r_e} \end{aligned}$$

and

$$\begin{aligned} F_0 &: [\pi'' C^{2,\alpha}(S^1)]^3 &\longrightarrow C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3 \\ \Psi &\longrightarrow [(\bar{w}_t, \bar{w}_b, \bar{w}_m), (\partial_r \bar{w}_t, \partial_r \bar{w}_b, \partial_r \bar{w}_m)]_{|r_e} \end{aligned}$$

Now using Fourier expansion of the function, we can see that  $C_0 = E_0 - F_0$  has an inverse which is bounded independently of  $\epsilon$ . In particular, the equation  $C_0(\Psi) = 0$  has the unique solution  $\Psi = (0, 0, 0)$ . Now we consider  $(C_{\epsilon} - C_0)(\Psi)$ , whose expression is

$$\left( r \cos \theta (2b_1 - \epsilon/2) + \pi'(u^+ - u_C) + \pi'(g_t - f_t), \\ r \cos \theta (-\xi) + \pi'(u^- - u_C) + \pi'(g_b - f_b), \\ \pi'(u^0 - u_C) + \pi'(-f_m) \right), \\ \left( \partial_r (u^+ - w_t) - \partial_r (u_C - \bar{w}_t) + g_{t,d} - f_{t,d} \\ \cos \theta (-\epsilon/2) + \partial_r (u^- - w_b) - \partial_r (u_C - \bar{w}_b) + g_{b,d} - f_{b,d} \\ \partial_r (u^0 - w_m) - \partial_r (u_C - \bar{w}_m) - f_{m,d} \right).$$

It is easy to prove that

$$||(C_{\epsilon} - C_0)(\Psi)||_{C^{2,\alpha}(S^1)^3 \times C^{1,\alpha}(S^1)^3} \leqslant c\epsilon$$

In order to solve  $C_{\epsilon}(\Psi) = 0$ , we find a fixed point for the mapping

$$D_{\epsilon}(\Psi) := C_0^{-1} ((C_{\epsilon} - C_0)(\Psi)).$$

### 4.10 Appendix A

**Definition 78.** Given  $\ell \in \mathbb{N}$ ,  $\alpha \in (0,1)$  and  $\nu \in \mathbb{R}$ , the space  $\mathcal{C}_{\nu}^{\ell,\alpha}(B_{\rho_0}(0))$  is defined to be the space of functions in  $\mathcal{C}_{loc}^{\ell,\alpha}(B_{\rho_0}(0))$  for which the following norm is finite

$$\|\rho^{-\nu} w\|_{\mathcal{C}^{\ell,\alpha}(B_{\rho_0}(0))}.$$

Now we can state the following result.

Proposition 79. There exists an operator

$$\tilde{H}: C^{2,\alpha}(S^1) \longrightarrow C^{2,\alpha}_0(S^1 \times [\bar{\rho}, +\infty)),$$

such that for each even function  $\varphi(\theta) \in C^{2,\alpha}(S^1)$ , which is  $L^2$ -orthogonal to the constant function, then  $w_{\varphi} = \tilde{H}_{\bar{\rho},\varphi}$  solves

$$\begin{cases} \Delta w_{\varphi} = 0 \quad on \quad S^{1} \times [\bar{\rho}, +\infty) \\ w_{\varphi} = \varphi \quad on \quad S^{1} \times \{\bar{\rho}\}. \end{cases}$$

Moreover,

 $||\tilde{H}_{\bar{\rho},\varphi}||_{C^{2,\alpha}_{-1}(S^1 \times [\bar{\rho}, +\infty))} \leqslant c \, ||\varphi||_{C^{2,\alpha}(S^1)},\tag{4.47}$ 

for some constant c > 0.

**Proof.** We consider the decomposition of the function  $\varphi$  with respect to the basis  $\{\cos(i\theta)\}$ , that is

$$\varphi = \sum_{i=1}^{\infty} \varphi_i \cos(i\theta).$$

Then the solution  $w_{\varphi}$  is given by

$$w_{\varphi}(\rho,\theta) = \sum_{i=1}^{\infty} \left(\frac{\bar{\rho}}{\rho}\right)^{i} \varphi_{i} \cos(i\theta).$$

Since  $\frac{\bar{\rho}}{\rho} \leqslant 1$ , then  $\left(\frac{\bar{\rho}}{\rho}\right)^i \leqslant \left(\frac{\bar{\rho}}{\rho}\right)$ , we can conclude that  $|w(r,\theta)| \leqslant c\rho^{-1}|\varphi(\theta)|$  and then  $||w_{\varphi}||_{C^{2,\alpha}_{-1}} \leqslant c||\varphi||_{C^{2,\alpha}}$ .

Calcolo1

$$\sup_{[s_0,t_{\epsilon}]\times S^1} e^{-\delta s} |w_{\Phi}|_{0,\alpha;[s,s+1]} \leqslant \sup_{[s_0,t_{\epsilon}]\times S^1} e^{-\delta s} e^{-\delta(t_{\epsilon}-s)} |\phi_t|_{2,\alpha} \leqslant$$
$$\sup_{[s_0,t_{\epsilon}]\times S^1} e^{-\delta t_{\epsilon}} |\phi_t|_{2,\alpha} \leqslant c_k \epsilon^{1+\delta/2}.$$

Calcolo 2

$$\sup_{s \in [s_0, t_{\epsilon}] \times S^1} e^{-\delta s} |w|^2_{0,\alpha;[s,s+1]} \leqslant ||w||^2_{2,\alpha,\mu/2} \leqslant c |\varphi|^2_{2,\alpha} \sup_{[s_0, t_{\epsilon}] \times S^1} e^{-\delta s/2} e^{-\delta(t_{\epsilon}-s)}$$
$$\leqslant c |\varphi|^2_{2,\alpha} \sup_{[s_0, t_{\epsilon}] \times S^1} e^{-\delta(t_{\epsilon}-s/2)} \leqslant c e^{-\delta t_{\epsilon}} e^{\delta s_0/2} |\varphi|^2_{2,\alpha} \leqslant c \epsilon^{-\delta/2} e^{\delta s_0/2} |\varphi|^2_{2,\alpha} \leqslant c_k \epsilon^{2-\delta/2}.$$

Now we give the statement of an useful result whose proof is contained in [7].

**Proposition 80.** There exists an operator

$$H: \mathcal{C}^{2,\alpha}(S^1) \longrightarrow \mathcal{C}^{2,\alpha}_{-2}([0,+\infty) \times S^1),$$

such that for all  $\varphi \in C^{2,\alpha}(S^1)$ , even function and orthogonal to 1 and  $\cos \theta$ , in the  $L^2$ -sense, the function  $w = H_{\varphi}$  solves

$$\begin{cases} \left(\partial_s^2 + \partial_\theta^2\right) w = 0 & in \quad [0, +\infty) \times S^1 \\ \\ w = \varphi & on \quad \{0\} \times S^1 \end{cases}$$

Moreover

 $\|H_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{-2}} \leqslant c \, \|\varphi\|_{\mathcal{C}^{2,\alpha}},$ 

for some constant c > 0.

Proposition 81. There exists an operator

$$\bar{\mathcal{H}}_{v_0}: C^{2,\alpha}(S^1) \longrightarrow C^{2,\alpha}_{\mu}(S^1 \times [v_0, +\infty)),$$

 $\mu \in (-2, -1)$ , such that for every function  $\varphi(v) \in C^{2,\alpha}(S^1)$ , which is  $L^2$ -orthogonal to  $e_{0,i}(u)$  with i = 0, 1 and even, the function  $w_{\varphi} = \overline{\mathcal{H}}_{v_0}(\varphi)$  solves

$$\begin{cases} \partial^2_{uu} w_{\varphi} + \partial^2_{vv} w_{\varphi} = 0 & on \quad S^1 \times [v_0, +\infty) \\ w_{\varphi} = \varphi & on \quad S^1 \times \{v_0\}. \end{cases}$$

Moreover,

$$|\bar{\mathcal{H}}_{v_0}(\varphi)||_{C^{2,\alpha}_{\mu}(S^1 \times [v_0, +\infty])} \leqslant c \, ||\varphi||_{C^{2,\alpha}(S^1)},\tag{4.48}$$

for some constant c > 0.

*Proof.* We consider the decomposition of the function  $\varphi$  with respect to the basis  $\{e_{0,i}(u)\}$ , that is

$$\varphi = \sum_{i=2}^{\infty} \varphi_i e_{0,i}(u).$$

Then the solution  $w_{\varphi}$  is given by

$$w_{\varphi}(u,v) = \sum_{i=2}^{\infty} e^{-i(v-v_0)} \varphi_i e_{0,i}(u).$$

We recall that  $\mu \in (-2, -1)$  so we have  $-i \leq \mu$  from which it follows  $|w_{\varphi}|_{2,\alpha;[v,v+1]} \leq e^{\mu(v-v_0)}|\varphi|_{2,\alpha}$  and

$$\|w_{\varphi}\|_{\mathcal{C}^{2,\alpha}_{\mu}} = \sup_{v \in [v_0,\infty]} e^{-\mu v} |w|_{2,\alpha;[v,v+1]} \leq \sup_{v \in [v_0,\infty]} e^{-\mu v} e^{\mu(v-v_0)} |\varphi|_{2,\alpha} \leq e^{-\mu v_0} |\varphi|_{2,\alpha}.$$

# 4.11 Appendix B

In section 4.5 we introduced the Jacobi operator about the surface  $\widetilde{M}_{\sigma,\alpha,\beta}$ . Its expression is  $\mathcal{J} = \frac{-K}{T(x(u),y(v))} \mathcal{L}_{\sigma}$ . Here we want to verify that the factor  $\frac{-K}{T(x(u),y(v))}$  is bounded. It is well known that the Gauss curvature has the following expression in terms of the Weierstrass data g, dh:

$$K = -16\left(|g| + \frac{1}{|g|}\right)^{-4} |\frac{dg}{g}|^2 |dh|^{-2}$$

We recall that  $dh = \frac{\mu dz}{\sqrt{(z^2 + \lambda^2)(z^2 + \lambda^{-2})}}$ . Since  $|z^2 + \lambda^2||z^2 + \lambda^{-2}|$  and  $T(x, y) = q^2 \cos^2 x(u) + p^2 \sin^2 y(v)$  have the same zeroes, that is the points D, D', D'', D''', then -K/T is bounded.

We can give an estimate of the derivatives of K and  $\sqrt{-K}$ . We can write  $\sqrt{-K} = \sqrt{T}\sqrt{\frac{-K}{T}}$ . So it is sufficient to study the derivatives of T.

We recall that

$$l(x) = \sqrt{1 - \sin^2 \sigma \sin^2 x} \quad m(y) = \sqrt{1 - \cos^2 \sigma \cos^2 y}$$

From the expression of T, using (4.12) it is easy to get:

$$\frac{\partial}{\partial u}\sqrt{T} = -\frac{\sin^2 \sigma \sin 2x(u)}{2\sqrt{T}}l(x(u)),$$
$$\frac{\partial}{\partial v}\sqrt{T} = \frac{\cos^2 \sigma \sin 2y(v)}{2\sqrt{T}}m(y(v)).$$

Then

$$\left|\frac{\partial}{\partial u}\sqrt{T}\right| = \frac{\sin^2\sigma|\sin 2x(u)|l(u)}{2\sqrt{\sin^2\sigma\cos^2x(u) + \cos^2\sigma\sin^2y(v)}} \leqslant \frac{\sin^2\sigma|\sin 2x(u)|}{2\sin\sigma|\cos x(u)|} \leqslant \sin\sigma,$$

$$\left|\frac{\partial}{\partial v}\sqrt{T}\right| = \frac{\cos^2\sigma|\sin 2y(v)|m(v)}{2\sqrt{\sin^2\sigma\cos^2x(u) + \cos^2\sigma\sin^2y(v)}} \leqslant \frac{\cos^2\sigma|\sin 2y(v)|}{2\cos\sigma|\sin y(v)|} \leqslant \cos\sigma.$$

We can conclude that the derivatives of  $\sqrt{T}$  (and trivially also the derivatives of T) are bounded.

### 4.12 Appendix C

**Proof of proposition 69.** In section 4.4.2 we parametrized the surface  $\widetilde{M}_{\sigma,\alpha,\beta}$  on the cylinder  $\mathbb{S}^1 \times \mathbb{R}$ . We introduced the map  $z(x, y) : \mathbb{S}^1 \times [0, \pi[\to \overline{\mathbb{C}} \text{ where } x, y \text{ denote the sphero-conal coordinates. Let } p, q \text{ be the real and the imaginary part of } z \in \overline{\mathbb{C}}$ . If Z denotes the immersion of the surface in  $\mathbb{R}^3$ , N its normal vector, then it holds that

$$|Z_p|^2 = |Z_q|^2 = \Lambda, \quad |N_p|^2 = |N_q|^2 = -K\Lambda,$$
  
$$\langle N_p, N \rangle = \langle N_q, N \rangle = 0, \quad \langle Z_p, Z_q \rangle = 0, \quad \langle N_p, N_q \rangle = 0,$$
  
$$\langle N_q, Z_q \rangle = -\langle N_p, Z_p \rangle, \quad \langle N_q, Z_p \rangle = \langle N_p, Z_q \rangle,$$

 $\mathbf{SO}$ 

$$\langle N_p, Z_p \rangle = |N_p| |Z_p| \cos \gamma_1 = \sqrt{-K} \Lambda \cos \gamma_1,$$
  
 
$$\langle N_p, Z_q \rangle = |N_p| |Z_q| \cos \gamma_2 = \sqrt{-K} \Lambda \cos \gamma_2.$$

Here  $\gamma_1$  is the angle between the vectors  $N_p$  and  $Z_p$ ,  $\gamma_2$  is the angle between the vectors  $N_p$  and  $Z_q$ .

The proof of proposition 69 is articulated in some lemmas. We recall that  $Z_f = Z + fN$ . We denote by  $E_f, F_f, G_f$  the coefficients of the second fundamental form for  $Z_f$ . The following lemma gives the expression of the area energy functional.

#### Lemma 82.

$$A(f) := \int \sqrt{E_f G_f - F_f^2} \, dp \, dq,$$

with

$$E_f G_f - F_f^2 = \Lambda^2 + \Lambda (f_p^2 + f_q^2) + 2K\Lambda^2 f^2 + 2f (f_q^2 - f_p^2) \sqrt{-K}\Lambda \cos\gamma_1 -4f f_p f_q \sqrt{-K}\Lambda \cos\gamma_2 - K\Lambda f^2 (f_p^2 + f_q^2) + f^4 K^2 \Lambda^2.$$

**Proof.** The coefficients of the second fundamental form are:

$$\begin{split} E_f &= |\partial_p Z_f|^2 = |Z_p|^2 + f_p^2 + f^2 |N_p|^2 + 2f \langle N_p, Z_p \rangle, \\ G_f &= |\partial_q Z_f|^2 = |Z_q|^2 + f_q^2 + f^2 |N_q|^2 + 2f \langle N_q, Z_q \rangle, \\ F_f &= |\partial_p Z_f \cdot \partial_q Z_f| = f_p f_q + f(\langle Z_p, N_q \rangle + \langle Z_q, N_p \rangle). \end{split}$$

Then

$$E_{f}G_{f} = |Z_{p}|^{2}|Z_{q}|^{2} + f_{p}^{2}|Z_{q}|^{2} + f_{q}^{2}|Z_{p}|^{2} + f^{2}(|N_{q}|^{2}|Z_{p}|^{2} + |N_{p}|^{2}|Z_{q}|^{2}) + f^{2}(f_{p}^{2}|N_{q}|^{2} + f_{q}^{2}|N_{p}|^{2}) + f^{4}|N_{p}|^{2}|N_{q}|^{2} + 4f^{2}(\langle N_{p}Z_{p}\rangle)(\langle N_{q}Z_{q}\rangle) + 2f(f_{p}^{2}\langle N_{q}, Z_{q}\rangle + f_{q}^{2}\langle N_{p}, Z_{p}\rangle) + f_{p}^{2}f_{q}^{2} + 2f(\langle N_{q}, Z_{q}\rangle|Z_{p}|^{2} + \langle N_{p}, Z_{p}\rangle|Z_{q}|^{2}) + 2f^{3}(\langle N_{q}, Z_{q}\rangle|Z_{p}|^{2} + \langle N_{p}, Z_{p}\rangle|Z_{q}|^{2})$$

Since  $\langle N_q, Z_q \rangle + \langle N_p, Z_p \rangle = 0$  and  $|Z_p|^2 = |Z_q|^2$  we can conclude that the last two terms of the previous expression are zero. Since  $\langle N_q, Z_p \rangle = \langle N_p, Z_q \rangle$  we have

$$F_f = f_p f_q + 2f \langle N_p, Z_q \rangle$$

Then

$$F_f^2 = f_p^2 f_q^2 + 4f^2 (\langle N_p, Z_q \rangle)^2 + 4f f_p f_q \langle N_p, Z_q \rangle.$$

So the expression of  $E_f G_f - F_f^2$  is:

$$\begin{split} |Z_p|^2 |Z_q|^2 + f_p^2 |Z_q|^2 + f_q^2 |Z_p|^2 + f^2 (|N_q|^2 |Z_p|^2 + |N_p|^2 |Z_q|^2) + \\ f^2 (f_p^2 |N_q|^2 + f_q^2 |N_p|^2) + f^4 |N_p|^2 |N_q|^2 + 4f^2 (\langle N_p, Z_p) (\langle N_q, Z_q \rangle) + 2f (f_p^2 \langle N_q, Z_q \rangle + f_q^2 \langle N_p, Z_p \rangle) \\ - 4f^2 (\langle N_p, Z_q \rangle)^2 - 4f f_p f_q \langle N_p, Z_q \rangle. \end{split}$$

Ordering the terms we get:

$$\begin{split} |Z_p|^2 |Z_q|^2 + f_p^2 |Z_q|^2 + f_q^2 |Z_p|^2 + f^2 (|N_q|^2 |Z_p|^2 + |N_p|^2 |Z_q|^2) &- 4f^2 (\langle N_p, Z_q \rangle)^2 \\ + 4f^2 (\langle N_p, Z_p \rangle) (\langle N_q, Z_q \rangle) + 2f (f_p^2 \langle N_q, Z_q \rangle + f_q^2 \langle N_p, Z_p \rangle) - 4f f_p f_q \langle N_p, Z_q \rangle + \\ &+ f^2 (f_p^2 |N_q|^2 + f_q^2 |N_p|^2) + f^4 |N_p|^2 |N_q|^2. \end{split}$$

The expression of  $E_f G_f - F_f^2$  becomes:

$$\Lambda^{2} + \Lambda (f_{p}^{2} + f_{q}^{2}) - 2K\Lambda^{2}f^{2} + 4f^{2}K\Lambda^{2} \left(\cos^{2}\gamma_{1} + \cos^{2}\gamma_{2}\right) +$$

$$+2f(f_q^2 - f_p^2)\sqrt{-K}\Lambda\cos\gamma_1 - 4ff_pf_q\sqrt{-K}\Lambda\cos\gamma_2 - K\Lambda f^2(f_p^2 + f_q^2) + f^4K^2\Lambda^2.$$

Using the relations  $\langle N_q, Z_p \rangle = \langle N_p, Z_q \rangle$  and  $\langle N_q, Z_q \rangle = -\langle N_p, Z_p \rangle$ , it is possible to understand that the relative positions of these vectors are such that  $\gamma_2 = \frac{\pi}{2} \pm \gamma_1$ . So  $\cos^2 \gamma_2 = \cos^2(\frac{\pi}{2} \pm \gamma_1) = \sin^2 \gamma_1$  and  $\cos^2 \gamma_1 + \cos^2 \gamma_2 = 1$ . Then we can write:

$$\Lambda^{2} + \Lambda (f_{p}^{2} + f_{q}^{2}) + 2K\Lambda^{2}f^{2} + 2f(f_{q}^{2} - f_{p}^{2})\sqrt{-K}\Lambda\cos\gamma_{1}$$
$$-4ff_{p}f_{q}\sqrt{-K}\Lambda\cos\gamma_{2} - K\Lambda f^{2}(f_{p}^{2} + f_{q}^{2}) + f^{4}K^{2}\Lambda^{2}.$$

The next lemma completes the proof of the proposition 69.

**Lemma 83.** The surface whose immersion is given by Z + fN, is minimal if and only if f satisfies

$$\mathcal{L}_{\sigma}f + Q_{\sigma}(f, f_u, f_v, f_{uu}, f_{uv}, f_{vv}) = 0.$$

**Proof.** The surface parameterized by  $Z_f = Z + fN$  is minimal if and only the first variation of A(f) is 0. That is

$$2 DA_{|f}(g) = \int \frac{1}{\sqrt{(E_f G_f - F_f^2)}} D_f(E_f G_f - F_f^2)(g) \, dp \, dq.$$

Thanks to the previous lemma it holds that

$$\frac{1}{\sqrt{(E_f G_f - F_f^2)}} D_f (E_f G_f - F_f^2)(g) = \frac{1}{\Lambda} \left( 2\Lambda (f_p g_p + f_q g_q) + 4K\Lambda^2 fg + 2\sqrt{-K}\Lambda \cos\gamma_1 \left[ 2ff_q g_q + gf_q^2 - 2ff_p g_p - gf_p^2 \right] + -4\sqrt{-K}\Lambda \cos\gamma_2 \left[ ff_q g_p + fg_q f_p + gf_p f_q \right] + -2K\Lambda \left[ fgf_p^2 + f_p g_p f^2 + fgf_q^2 + f_q g_q f^2 \right] + 4K^2\Lambda^2 f^3 g \right).$$

Reordering the summands, we have:

$$\frac{1}{\sqrt{(E_f G_f - F_f^2)}} D_f (E_f G_f - F_f^2)(g) = 2 \left( f_p g_p + f_q g_q + 2K\Lambda f g + \sqrt{-K} \cos \gamma_1 \left[ 2f (f_q g_q - f_p g_p) + g (f_q^2 - f_p^2) \right] + 2\sqrt{-K} \cos \gamma_2 \left[ f (f_q g_p + g_q f_p) + g f_p f_q \right] + K \left[ f g (f_p^2 + f_q^2) + f^2 (f_p g_p + f_q g_q) \right] + 2K^2 \Lambda f^3 g \right).$$

In the next computation we can skip the factor 2 in front of the last expression.

$$f_p g_p + f_q g_q + 2K\Lambda f g + Q_1(f, f_p, f_q)g - Q_2(f, f_p, f_q)g_p - Q_3(f, f_p, f_q)g_q = 0,$$

where

$$\begin{aligned} Q_1(f, f_p, f_q) &= -(f_p^2 - f_q^2)\sqrt{-K}\cos\gamma_1 - 2f_p f_q \sqrt{-K}\cos\gamma_2 - Kf(f_p^2 + f_q^2) + 2K^2\Lambda f^3, \\ Q_2(f, f_p, f_q) &= 2f f_p \sqrt{-K}\cos\gamma_1 + 2f f_q \sqrt{-K}\cos\gamma_2 + Kf^2 f_p, \\ Q_3(f, f_p, f_q) &= -2f f_q \sqrt{-K}\cos\gamma_1 + 2f f_p \sqrt{-K}\cos\gamma_2 + Kf^2 f_q. \end{aligned}$$

An integration by parts and a change of sign give us the equation:

$$(f_{pp} + f_{qq} - 2K\Lambda f - Q_1(f, f_p, f_q) +$$

$$+P_2(f, f_p, f_q, f_{pp}, f_{pq}, f_{qq}) + P_3(f, f_p, f_q, f_{pp}, f_{pq}, f_{qq})) g = 0,$$

where

$$P_2(f, f_p, f_q, f_{pp}, f_{pq}, f_{qq}) = \partial_p Q_2(f, f_p, f_q)$$

and

$$P_3(f, f_p, f_q, f_{pp}, f_{pq}, f_{qq}) = \partial_q Q_3(f, f_p, f_q)$$

That is

$$P_{2}(f, f_{p}, f_{q}, f_{pp}, f_{pq}, f_{qq}) = 2(f_{p}^{2} + ff_{pp})\sqrt{-K}\cos\gamma_{1} + 2(f_{p}f_{q} + ff_{pq})\sqrt{-K}\cos\gamma_{2} + K(2ff_{p}^{2} + f^{2}f_{pp}) + 2f(f_{p}(\sqrt{-K}\cos\gamma_{1})_{p} + f_{q}(\sqrt{-K}\cos\gamma_{2})_{p}) + f^{2}f_{p}K_{p}$$

and

$$P_{3}(f, f_{p}, f_{q}, f_{pp}, f_{pq}, f_{qq}) = -2(f_{q}^{2} + ff_{qq})\sqrt{-K}\cos\gamma_{1} + 2(f_{p}f_{q} + ff_{pq})\sqrt{-K}\cos\gamma_{2} + K(2ff_{q}^{2} + f^{2}f_{qq}) + 2f(-f_{q}(\sqrt{-K}\cos\gamma_{1})_{q} + f_{p}(\sqrt{-K}\cos\gamma_{2})_{q}) + f^{2}f_{q}K_{q}.$$

Now we change the variables passing from the variables (p,q) to the (u,v) variables. Then we want to understand how the minimal surfaces equation changes. We recall that p and qare the real and imaginary part of the variable z, the same that appears in the Weierstrass representation of the surface. It is known that the metric  $\bar{g}$  induced on a surface whose immersion Z is given by the Weierstrass representation on a domain of the complex zplane, can be expressed in terms of the metric  $d\bar{s}^2 = dp^2 + dq^2$ , by  $\bar{g} = \Lambda(dp^2 + dq^2)$ , where  $\Lambda = |Z_p|^2 = |Z_q|^2$ . It is well known that in this case then there exists a simple relation between the Laplace-Beltrami operators written with respect to the metrics  $d\bar{s}^2$  and  $\bar{g}$ . As a fact they differ for a conformal factor:

$$\Delta_{d\bar{s}^2} = \frac{1}{\Lambda} \Delta_{\bar{g}}.$$

In section 69 we observed that the conformal factor related to the change of coordinates  $(x, y) \rightarrow (u, v)$  is -K/T. So the conformal factor due to the change of coordinates  $(p, q) \rightarrow (u, v)$  is obtained by multiplication of the conformal factors described above. Summarizing it holds that

$$f_{pp} + f_{qq} = \frac{-K\Lambda}{T}(f_{uu} + f_{vv})$$

So we can write

$$\frac{-K\Lambda}{T} (f_{uu} + f_{vv}) + 2 (-K\Lambda) f + R_1 + R_2 + R_3 = 0,$$

where

$$R_{1}(f, f_{u}, f_{v}) = -\frac{-K\Lambda}{T} \left[ -(f_{u}^{2} - f_{v}^{2})\sqrt{-K}\cos\gamma_{1} - 2f_{u}f_{v}\sqrt{-K}\cos\gamma_{2} - Kf(f_{u}^{2} + f_{v}^{2}) \right] - 2K^{2}\Lambda f^{3}$$

$$= \frac{-K\Lambda}{T} \left[ (f_{u}^{2} - f_{v}^{2})\sqrt{-K}\cos\gamma_{1} + 2f_{u}f_{v}\sqrt{-K}\cos\gamma_{2} + Kf(f_{u}^{2} + f_{v}^{2}) - 2KTf^{3} \right] = \frac{-K\Lambda}{T} \bar{P}_{1}(f, f_{u}, f_{v}),$$

$$R_{2}(f, f_{u}, f_{v}, f_{uu}, f_{uv}, f_{vv}) = \frac{-K\Lambda}{T} P_{2}(f, f_{u}, f_{v}, f_{uu}, f_{uv}, f_{vv})$$

and

$$R_3(f, f_u, f_v, f_{uu}, f_{uv}, f_{vv}) = \frac{-K\Lambda}{T} P_3(f, f_u, f_v, f_{uu}, f_{uv}, f_{vv}).$$

We can write

$$\frac{-K\Lambda}{T} \left[ f_{uu} + f_{vv} + 2T(u,v)f + \bar{P}_1(f) + P_2(f) + P_3(f) \right] = 0.$$

We can recognize the Lamé operator,

$$\mathcal{L}_{\sigma}f = f_{uu} + f_{vv} + 2(\sin^2 \sigma \cos^2 x(u) + \cos^2 \sigma \sin^2 y(v))f,$$

then, if we set  $Q_{\sigma} = \overline{P}_1(f) + P_2(f) + P_3(f)$ , the equation can be written

$$\mathcal{L}_{\sigma}f + Q_{\sigma}(f) = 0.$$

The estimate about  $Q_{\sigma}$  is an easy ensequence of the fact that all its coefficients are bounded. That completes the proof.

# Bibliography

- M. Abramowitz, I. Stegun, Handbook of Mathematical Functions, National Bureau of Standards Series, 55.
- [2] V. Ramos Batista, Singly periodic Costa surfaces, J. London Math. Soc. 2, 72(2), 478
   496, 2005.
- [3] C. J. Costa, Imersões minimas en ℝ<sup>3</sup> de gênero un e curvatura total finita, PhD thesis, IMPA, Rio de Janeiro, Brasil, 1982.
- [4] C. J. Costa, Example of a complete minimal immersion in ℝ<sup>3</sup> of genus one and three embedded ends, Bol. Soc. Brasil. Mat. 15 (1984), no. 1-2, 47-54.
- [5] V.I. Danilov, V.V. Shokurov, Algebraic curves, algebraic manifolds and schemes, Springer Verlag, (1998).
- [6] N. Ejiri, M. Kotani, Index and flat ends of minimal surfaces, Tokyo J. Math., 16 (1993), no. 1, 37-48.
- [7] S. Fakhi and F. Pacard, Existence result for minimal hypersurfaces with a prescribed finite number of planar ends, Manuscripta Math., 103 (2000), no. 4, 465-512.
- [8] D.Fischer-Colbrie, On complete minimal surfaces with Morse index in three manifold, Inventiones Mathematicae, 82 (1985), 121-132.
- [9] D. Gilbarg, N. Trudinger, *Elliptic partial differential equations of second order*, Springer-Verlag 1998.
- [10] R. Gulliver, H.B. Lawson, Jr., The structure of stable minimal hypersurfaces near a singularity, Proc. Symp. Pure Mathematics, 44 (1986), 213-237
- [11] L. Hauswirth, F. Pacard, *Higher genus Riemann minimal surfaces*, to appear in Inventiones Mathematicae.
- [12] L. Hauswirth, R. Sa Earp, E. Toubiana, Associate and conjugate minimal immersions in  $M \times \mathbb{R}$ , Preprint.
- [13] D. Hoffman and H. Karcher, Complete embedded minimal surfaces of finite total curvature, Geometry, V, 5-93, 267-272, Encyclopaedia Math. Sci., 90, Springer, Berlin, 1997.
- [14] D. Hoffman and W.H. Meeks III, A Complete Embedded Minimal Surface in  $\mathbb{R}^3$  with Genus One and Three Ends, Journal of Differential Geometry, 21 (1985), 109-127.
- [15] D. Hoffman and W.H. Meeks III, The asymptotic behavior of properly embedded minimal surfaces of finite topology, Journal of the AMS, (2) 4 (1989), 667-681.
- [16] D. Hoffman and W.H. Meeks III, Embedded minimal surfaces of finite topology, Annals of Mathematics, 131 (1990), 1-34.
- [17] J. K. M. Jansen, Simple-periodic and non-periodic Lamé functions, Mathematisch Centrum, Amsterdam 1977.
- [18] M. Jleli, Constant mean curvature hypersurfaces, PhD Thesis, University Paris 12 (2004).
- [19] H. Karcher, Embedded minimal surfaces derived from Scherk's examples, Manuscripta Math., 62, 83-114, 1988.
- [20] H. Karcher, Construction of minimal surfaces, Surveys in Geometry, 1 96, University of Tokyo, 1989 and Lecture Notes n. 12, SFB256, Bonn, 1989.
- [21] T. Kato, *Perturbation theory for linear operators*, Springer-Verlag, Berlin 1980.
- [22] N. Korevaar, R. Kusner and B. Solomon, The Structure of Complete Embedded n. Surfaces with Constant Mean Curvature, Jour. of Differential Geometry, 30 (1989), 465-503.
- [23] R. Kusner, R. Mazzeo and D. Pollack, The moduli space of complete embedded constant mean curvature surfaces, Geom. Funct. Anal., 6 (1996) 120 - 137.
- [24] R. Mazzeo, F. Pacard, D. Pollack, Connected sums of constant mean curvature surfaces in Euclidean 3 space, J. Reine Angew. Math., 536, (2001), 115-165.
- [25] W. H. Meeks III, H. Rosenberg, The theory of minimal surfaces in  $M \times \mathbb{R}$ , Comment. Math. Helv., 80, (2005), no. 4, 811-858.
- [26] W. H. Meeks III, H. Rosenberg, Stable minimal surfaces in in  $M \times \mathbb{R}$ , J. Differential Geom., 68, (2004), no. 3, 515-534.
- [27] R. Melrose, The Atiyah-Patodi-Singer index theorem, Research Notes in Mathematics, (1993).

- [28] S. Montiel, A. Ros, Schrödinger operators associated to a holomorphic map, Global differential geometry and global analysis, Lecture Notes in Mathematics n. 1481, Springer, Berlin, (1991), 147-174.
- [29] S. Nayatani, Morse index of complete minimal surfaces, The problem of Plateau, ed. Th. M. Rassias, (1992), 181-189.
- [30] S. Nayatani, Morse Index and Gauss maps of complete minimal surfaces in Euclidean 3-space, Comment. Math. Helv., 68, (1993), Number 4, 511-537.
- [31] S. Nayatani, Personal communication.
- [32] B. Nelli, H. Rosenberg, Minimal surfaces in  $\mathbb{H}^2 \times \mathbb{R}$ , Bull. Braz. Math. Soc., 33, (2002), 263-292.
- [33] J. Pérez, M. M. Rodríguez, M. Traizet, The classification of doubly periodic minimal tori with parallel ends, Journal of Differential Geometry, 69, no. 3, 523-577, 2005.
- [34] J. Pérez and A. Ros, The space of properly embedded minimal surfaces with finite total curvature, Indiana Univ. Math. Journal, 45, (1996), no. 1, 177-204.
- [35] M. M. Rodríguez The space of doubly periodic minimal tori with parallel ends: the standard examples, Michigan Math. Journal, 55, (2007), 103-122.
- [36] H. Rosenberg, Minimal surfaces in  $M \times \mathbb{R}$ , Illinois J. Math., 46, (2002), no. 4, 1177-1195.
- [37] H.F. Scherk, Bemerkungen ber die kleinste Flche innerhalb gegebener Grenzen, J. R. Angew. Math., 13, 185-208, 1935.